# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

# **FORM 10-Q**

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SEC 1934	CURITIES EXCHANGE ACT OF
For the quarterly period en	nded March 31, 2017
OR	
$\hfill\Box$ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECU1934	URITIES EXCHANGE ACT OF
For the transition period from	to
Commission File Nun	nber 001-31539
SM <b>≜EN</b>	ERGY
SM ENERGY O	COMPANY
(Exact name of registrant as	
<b>Delaware</b> (State or other jurisdiction of incorporation or organization)	41-0518430 (I.R.S. Employer Identification No.)
1775 Sherman Street, Suite 1200, Denver, Colorado	80203
(Address of principal executive offices)  (303) 861-8  (Registrant's telephone number)	
Indicate by check mark whether the registrant (1) has filed all reports required to be filed by months (or for such shorter period that the registrant was required to file such reports), and	
Indicate by check mark whether the registrant has submitted electronically and posted on its posted pursuant to Rule 405 of Regulation S-T ( $\S232.405$ of this chapter) during the preceding and post such files). Yes $\square$ No $\square$	
Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer company. See the definitions of "large accelerated filer," "accelerated filer," "smaller report	
Large accelerated filer   ✓	Accelerated filer □
Non-accelerated filer □ (Do not check if a smaller reporting company)	Smaller reporting company □
	Emerging growth company $\square$
If an emerging growth company, indicate by check mark if the registrant has elected not to financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. $\Box$	use the extended transition period for complying with any new or revised
Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2	of the Exchange Act). Yes □ No ☑
Indicate the number of shares outstanding of each of the issuer's classes of common stock, a	as of the latest practicable date.
As of April 26, 2017, the registrant had 111,258,225 shares of common stock, $\$0.01$ par values of the property of the stock of the property	ue, outstanding.
1	

# $\frac{\text{SM ENERGY COMPANY}}{\text{TABLE OF CONTENTS}}$

Part I.	<u>FINANCIA</u>	AL INFORMATION	<u>PAGE</u>
	Item 1.	Financial Statements (Unaudited)	
		Condensed Consolidated Balance Sheets March 31, 2017, and December 31, 2016	<u>3</u>
		Condensed Consolidated Statements of Operations Three Months Ended March 31, 2017, and 2016	4
		Condensed Consolidated Statements of Comprehensive Income (Loss) Three Months Ended March 31, 2017, and 2016	<u>5</u>
		Condensed Consolidated Statement of Stockholders' Equity Three Months Ended March 31, 2017	<u>6</u>
		Condensed Consolidated Statements of Cash Flows Three Months Ended March 31, 2017, and 2016	2
		Notes to Condensed Consolidated Financial Statements	9
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>25</u>
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk (included within the content of Item 2)	44
	Item 4.	Controls and Procedures	<u>44</u>
Part II.	OTHER IN	IFORMATION	
	Item 1.	Legal Proceedings	<u>45</u>
	Item 1A.	Risk Factors	<u>45</u>
	Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>46</u>
	<u>Item 6.</u>	<u>Exhibits</u>	<u>47</u>
		2	

# PART I. FINANCIAL INFORMATION

## ITEM 1. FINANCIAL STATEMENTS

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED) (in thousands, except share amounts)

		March 31, 2017		December 31, 2016
ASSETS				
Current assets:				
Cash and cash equivalents	\$	659,147	\$	9,372
Accounts receivable		108,368		151,950
Derivative asset		73,978		54,521
Prepaid expenses and other		8,053		8,799
Total current assets		849,546		224,642
Property and equipment (successful efforts method):				
Proved oil and gas properties		4,803,068		5,700,418
Less - accumulated depletion, depreciation, and amortization		(2,589,204)		(2,836,532)
Unproved oil and gas properties		2,483,601		2,471,947
Wells in progress		186,707		235,147
Oil and gas properties held for sale, net		455,943		372,621
Other property and equipment, net of accumulated depreciation of \$44,662 and \$42,882, respectively		110,005		137,753
Total property and equipment, net		5,450,120		6,081,354
Noncurrent assets:				
Derivative asset		84,195		67,575
Other noncurrent assets		15,847		19,940
Total other noncurrent assets		100,042		87,515
Total Assets	\$	6,399,708	\$	6,393,511
1 Otal Assets		0,333,708	9	0,373,311
LIADII ITIES AND STOCKHOLDEDS! FOLIITY				
LIABILITIES AND STOCKHOLDERS' EQUITY  Current liabilities:				
	\$	299,676	\$	299,708
Accounts payable and accrued expenses  Derivative liability	Ф	53,809	Þ	
Total current liabilities		-		115,464
Total Current Habilities		353,485		415,172
Noncurrent liabilities:				
Revolving credit facility		_		_
Senior Notes, net of unamortized deferred financing costs		2,765,714		2,766,719
Senior Convertible Notes, net of unamortized discount and deferred financing costs		132,889		130,856
Asset retirement obligation		83,160		96,134
Asset retirement obligation associated with oil and gas properties held for sale		16,056		26,241
Deferred income taxes		304,331		315,672
Derivative liability		81,306		98,340
Other noncurrent liabilities		47,252		47,244
Total noncurrent liabilities		3,430,708		3,481,206
Committee and a participant in Control				
Commitments and contingencies (note 6)				
Stockholders' equity:				
Common stock, \$0.01 par value - authorized: 200,000,000 shares; issued and outstanding: 111,258,225 and 111,257,500 shares, respectively		1,113		1,113
Additional paid-in capital		1,723,010		1,716,556
Retained earnings		906,515		794,020
Accumulated other comprehensive loss		(15,123)		(14,556)
Total stockholders' equity		2,615,515		2,497,133
Total Liabilities and Stockholders' Equity	\$	6,399,708	\$	6,393,511

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED) (in thousands, except per share amounts)

	For the Three Months Ended March			led March 31,
		2017		2016
Operating revenues and other income:				
Oil, gas, and NGL production revenue	\$	333,198	\$	211,823
Net gain (loss) on divestiture activity		37,463		(69,021)
Other operating revenues		2,077		274
Total operating revenues and other income		372,738		143,076
Operating expenses:				
Oil, gas, and NGL production expense		138,046		144,543
Depletion, depreciation, amortization, and asset retirement obligation liability accretion		137,812		214,207
Exploration		11,978		15,273
Impairment of proved properties		_		269,785
Abandonment and impairment of unproved properties		_		2,311
General and administrative		29,224		32,238
Net derivative gain		(114,774)		(14,228)
Other operating expenses		4,859		5,672
Total operating expenses		207,145		669,801
Income (loss) from operations		165,593		(526,725)
Non-operating income (expense):				
Interest expense		(46,953)		(31,088)
Gain (loss) on extinguishment of debt		(35)		15,722
Other, net		335		6
Income (loss) before income taxes		118,940		(542,085)
Income tax (expense) benefit		(44,506)		194,875
Net income (loss)	\$	74,434	\$	(347,210)
Basic weighted-average common shares outstanding		111,258		68,077
Diluted weighted-average common shares outstanding		111,329		68,077
Basic net income (loss) per common share	\$	0.67	\$	(5.10)
Diluted net income (loss) per common share	\$	0.67	\$	(5.10)
Dividends per common share	\$	0.05	\$	0.05

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (UNAUDITED) (in thousands)

	]	For the Three Mare	
		2017	2016
Net income (loss)	\$	74,434	\$ (347,210)
Other comprehensive loss, net of tax:			
Pension liability adjustment		(567)	(236)
Total other comprehensive loss, net of tax		(567)	(236)
Total comprehensive income (loss)	\$	73,867	\$ (347,446)

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (UNAUDITED) (in thousands, except share amounts)

	Common	Stock Amoun		Additional Paid- in Capital	Retained Earnings	 imulated Other prehensive Loss	\$ Total Stockholders' Equity
Balances, December 31, 2016	111,257,500	\$ 1,11	3 \$	\$ 1,716,556	\$ 794,020	\$ (14,556)	\$ 2,497,133
Net income	_	-	_	_	74,434	_	74,434
Other comprehensive loss	_	-	_	_	_	(567)	(567)
Dividends declared, \$ 0.05 per share	_	_	_	_	(5,563)	_	(5,563)
Issuance of common stock upon vesting of restricted stock units, net of shares used for tax withholdings	725	-	_	(11)	_	_	(11)
Stock-based compensation expense	_	-	_	5,455	_	_	5,455
Cumulative effect of accounting change (1)	_	-	_	1,108	43,624	_	44,732
Other	_	_	_	(98)	_	_	(98)
Balances, March 31, 2017	111,258,225	\$ 1,11	3 \$	\$ 1,723,010	\$ 906,515	\$ (15,123)	\$ 2,615,515

<sup>(1)</sup> Refer to Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards.

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

	For	For the Three Months Ended March		
		2017	2016	
Cash flows from operating activities:				
Net income (loss)	\$	74,434 \$	(347,210)	
Adjustments to reconcile net income (loss) to net cash provided by operating activities:				
Net (gain) loss on divestiture activity		(37,463)	69,021	
Depletion, depreciation, amortization, and asset retirement obligation liability accretion		137,812	214,207	
Impairment of proved properties		_	269,785	
Abandonment and impairment of unproved properties		_	2,311	
Stock-based compensation expense		5,455	6,868	
Net derivative gain		(114,774)	(14,228)	
Derivative settlement gain		7	147,028	
Amortization of discount and deferred financing costs		4,946	(920)	
Non-cash (gain) loss on extinguishment of debt, net		22	(15,722)	
Deferred income taxes		33,225	(195,039)	
Plugging and abandonment		(1,191)	(604)	
Other, net		4,567	(1,151)	
Changes in current assets and liabilities:				
Accounts receivable		30,407	26,922	
Prepaid expenses and other		178	4,984	
Accounts payable and accrued expenses		(5,497)	(52,294)	
Accrued derivative settlements		2,838	4,318	
Net cash provided by operating activities		134,966	118,276	
Cash flows from investing activities:				
Net proceeds from the sale of oil and gas properties		744,333	1,206	
Capital expenditures		(154,401)	(176,370)	
Acquisition of proved and unproved oil and gas properties		(75,105)	(15,044)	
Other, net		2,486	885	
Net cash provided by (used in) investing activities		517,313	(189,323)	
Cash flows from financing activities:				
Proceeds from credit facility		397,500	317,000	
Repayment of credit facility		(397,500)	(226,000)	
Cash paid to repurchase Senior Notes		(2,344)	(19,917)	
Other, net		(160)	(3)	
Net cash provided by (used in) financing activities		(2,504)	71,080	
Net change in cash and cash equivalents		649,775	33	
Cash and cash equivalents at beginning of period		9,372	18	
Cash and cash equivalents at end of period	\$	659,147 \$	51	

 $The \ accompanying \ notes \ are \ an \ integral \ part \ of \ these \ condensed \ consolidated \ financial \ statements.$ 

# SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (Continued) (in thousands)

Supplemental schedule of additional cash flow information and non-cash activities:

	For the Three Months Ended March 31,			
		2017		2016
Supplemental Cash Flow Information:		_		
Operating Activities:				
Cash paid for interest, net of capitalized interest	\$	(42,872)	\$	(24,453)
Net cash refunded for income taxes	\$	15	\$	4,689
Investing Activities:				
Changes in capital expenditure accruals and other	\$	27,214	\$	20,643
Supplemental Non-Cash Investing Activities:				
Fair value of properties exchanged	\$	24,544	\$	733
Supplemental Non-Cash Financing Activities:				
Dividends declared, but not paid	\$	5,563	\$	3,404
The accompanying notes are an integral part of these condensed consolidated financial statements.				
8				

#### SM ENERGY COMPANY AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### Note 1 - The Company and Business

SM Energy Company, together with its consolidated subsidiaries ("SM Energy" or the "Company"), is an independent energy company engaged in the acquisition, exploration, development, and production of crude oil and condensate, natural gas, and natural gas liquids (also respectively referred to as "oil," "gas," and "NGLs" throughout this report) in onshore North America.

#### Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards

#### Basis of Presentation

The accompanying unaudited condensed consolidated financial statements include the accounts of SM Energy and its wholly-owned subsidiaries and have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial information and the instructions to Quarterly Report on Form 10-Q and Regulation S-X. These financial statements do not include all information and notes required by GAAP for annual financial statements. However, except as disclosed herein, there has been no material change in the information disclosed in the notes to consolidated financial statements included in SM Energy's Annual Report on Form 10-K for the year ended December 31, 2016 (the "2016 Form 10-K"). In the opinion of management, all adjustments, consisting of normal recurring adjustments considered necessary for a fair presentation of interim financial information, have been included. Operating results for the periods presented are not necessarily indicative of expected results for the full year. In connection with the preparation of the Company's unaudited condensed consolidated financial statements, the Company evaluated events subsequent to the balance sheet date of March 31, 2017, and through the filing of this report. Certain prior period amounts have been reclassified to conform to the current presentation on the accompanying condensed consolidated financial statements.

#### Significant Accounting Policies

The significant accounting policies followed by the Company are set forth in Note 1 to the Company's consolidated financial statements in its2016 Form 10-K, and are supplemented by the notes to the unaudited condensed consolidated financial statements in this report. These unaudited condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes included in the 2016 Form 10-K.

#### Recently Issued Accounting Standards

Effective January 1, 2017, the Company adopted, using various transition methods, Financial Accounting Standards Board ("FASB") Accounting Standards Update ("ASU") No. 2016-09, Compensation-Stock Compensation (Topic 718): Improvements to Employee Share-Based Payment Accounting ("ASU 2016-09"). ASU 2016-09 is meant to simplify certain aspects of accounting for share-based payment arrangements, including income tax effects, accounting for forfeitures, and net share settlements. The Company adopted the various applicable amendments as summarized below:

• ASU 2016-09 requires all future excess tax benefits and deficiencies related to share-based payment arrangements to be recognized as income tax benefit or expense as discrete events in the period in which they occur. The Company adopted this amendment under a modified retrospective transition method, which resulted in a \$44.3 million cumulative-effect adjustment to retained earnings as of January 1, 2017, with a corresponding deferred tax asset recorded for previously unrecognized excess tax benefits. Consequentially, the Company's diluted share count calculation changed prospectively beginning in the period ended March 31, 2017. Any future excess tax benefits and deficiencies are now being excluded from the assumed proceeds calculated under the treasury stock method for share-based payment arrangements. Lastly, this amendment requires companies to present excess tax benefits as an operating activity on the statement of cash flows rather than as a financing activity, which the Company applied prospectively beginning in the period ended March 31, 2017. There are no periods presented that would require reclassification of cash flows had the Company elected to apply this amendment retrospectively.

- ASU 2016-09 allows entities to elect whether to account for forfeitures of share-based payment awards by recognizing forfeitures of awards as they occur or by estimating the number of awards expected to be forfeited. This amendment is to be applied using a modified retrospective transition method. Upon adopting ASU 2016-09, the Company elected to change its policy to account for forfeitures of share-based payment awards as they occur effective January 1, 2017. This change in accounting policy resulted in a net \$0.7 million cumulative-effect adjustment to retained earnings as of January 1, 2017, to adjust for actual forfeitures versus the previously estimated forfeiture rate. The corresponding impacts were an increase in additional paid-in capital and a decrease in deferred tax assets.
- ASU 2016-09 allows entities to withhold an amount up to the employees' maximum individual tax rate in the relevant jurisdiction, without triggering liability classification of the award; however, the Company does not plan on changing its current withholding process as outlined in its share-based award agreements. Related to this amendment, ASU 2016-09 requires entities to present cash payments made to tax authorities on the employees' behalf for withheld tax shares as a financing activity on the statement of cash flows retrospectively. However, this presentation is already consistent with the Company's historical and current cash flow presentation of net share settlement from issuance of stock awards, and therefore, there was no impact from this amendment.

In May 2014, the FASB issued ASU No. 2014-09, Revenue from Contracts with Customers (Topic 606) for the recognition of revenue from contracts with customers. Several additional related ASUs have been issued. The Company has established a cross-functional implementation team that is currently evaluating the provisions of each of these standards, analyzing their impact on the Company's contract portfolio, reviewing current accounting policies and practices to identify potential differences that would result from applying the requirements of these standards to the Company's revenue contracts, and assessing their potential impact on the Company's financial statements and disclosures. The Company currently plans to apply the modified retrospective method upon adoption and plans to adopt the guidance on the effective date of January 1, 2018.

In February 2016, the FASB issued ASU No. 2016-02, Leases (Topic 842), which changes the accounting for leases. The Company is currently establishing a cross-functional implementation team to analyze the impact of the standard on the Company's contract portfolio by reviewing current accounting policies and practices to identify potential differences that would result from applying the requirements of the new standard. In addition, the Company is working to identify appropriate changes to the Company's business processes, systems, and controls to support recognition and disclosure under the new standard. The Company currently plans to adopt the guidance on the effective date of January 1, 2019.

In March 2017, the FASB issued ASU No. 2017-07, Compensation - Retirement Benefits (Topic 715): Improving the Presentation of Net Periodic Pension Cost and Net Periodic Postretirement Benefit Cost ("ASU 2017-07"). This ASU is intended to improve the presentation of net periodic pension cost and net periodic postretirement benefit cost by providing additional guidance on the presentation of net benefit cost on the income statement and on the components eligible for capitalization in assets. As outlined in ASU 2017-07, certain amendments are to be applied using a retrospective method while others are required to be applied using a prospective method. The guidance is effective for annual periods, and interim periods within those annual periods, beginning after December 15, 2017. Early adoption is permitted as outlined in ASU 2017-07. The Company is currently evaluating the provisions of this guidance and assessing the potential impact on the Company's financial statements and disclosures and does not intend to early adopt this guidance.

Other than as disclosed above or in the 2016 Form 10-K, there are no other accounting standards applicable to the Company that would have a material effect on the Company's financial statements and related disclosures that have been issued but not yet adopted by the Company as of March 31, 2017, and through the filing of this report.

#### Note 3 - Divestitures, Assets Held for Sale, and Acquisitions

#### Divestitures

On March 10, 2017, the Company divested its outside-operated Eagle Ford shale assets, including ownership interest in related midstream assets, for total cash received at closing, net of commissions (referred to throughout this report as "net divestiture proceeds"), of \$747.4 million, subject to post-closing adjustments, and recorded a net estimated gain of \$398.1 million for the three months ended March 31, 2017. These assets were classified as held for sale as of December 31, 2016.

The following table presents income (loss) before income taxes of the outside-operated Eagle Ford shale assets sold for the three months endedMarch 31, 2017, and 2016. This divestiture is considered a disposal of a significant asset group.

	F	or the Three Mon	ths En	ded March 31,
		2017		2016
		(in tho	usand	s)
Income (loss) before income taxes (1)	\$	24,324	\$	(286,399)

(I) Income (loss) before income taxes reflects oil, gas, and NGL production revenue, less oil, gas, and NGL production expense and depletion, depreciation, amortization, and asset retirement obligation liability accretion. Additionally, income (loss) before income taxes included impairment of proved properties expense of approximately \$269.6 million for the three months ended March 31, 2016.

#### Assets Held for Sale

Assets are classified as held for sale when the Company commits to a plan to sell the assets and there is reasonable certainty the sale will take place within year. Upon classification as held for sale, long-lived assets are no longer depreciated or depleted, and a measurement for impairment is performed to identify and expense any excess of carrying value over fair value less estimated costs to sell. Any subsequent changes to the fair value less estimated costs to sell impact the measurement of assets held for sale, with any gain or loss reflected in the net gain (loss) on divestiture activity line item in the accompanying condensed consolidated statements of operations ("accompanying statements of operations").

As of March 31, 2017, the accompanying condensed consolidated balance sheets ("accompanying balance sheets") present \$455.9 million of assets held for sale, net of accumulated depletion, depreciation, and amortization expense, which consisted primarily of the Company's remaining Williston Basin assets in Divide County, North Dakota (referred to as "Divide County" throughout this report). A corresponding asset retirement obligation liability of \$16.1 million was separately presented. The Company expects to close the divestiture of these assets during the second quarter of 2017. These assets were written down by \$359.6 million to reflect fair value less estimated costs to sell upon classification as held for sale during the three months ended March 31, 2017.

The following table presents loss before income taxes of the Company's Divide County assets held for sale for the three months endedMarch 31, 2017, and 2016:

	 For the Three Mon	ths En	ded March 31,
	2017		2016
	 (in tho	usand	s)
Loss before income taxes (1)	\$ (335,561)	\$	(16,813)

(1) Loss before income taxes reflects oil, gas, and NGL production revenue less oil, gas, and NGL production expense and depletion, depreciation, amortization, and asset retirement obligation liability accretion. Additionally, loss before income taxes for the three months ended March 31, 2017, included the \$359.6 million write-down on these assets held for sale as discussed above, and loss before income taxes for the three months ended March 31, 2016, included \$1.6 million of unproved property impairments.

#### Acquisitions

During the first quarter of 2017, the Company acquired approximately 2,900 net acres of proved and unproved properties in the Midland Basin for\$59.6 million, subject to post-closing adjustments. Under authoritative accounting guidance, this transaction was considered an asset acquisition, and therefore, the properties were recorded based on the fair value of the total consideration transferred on the acquisition date and transaction costs were capitalized as a component of the cost of the assets acquired.

The Company finalized the 2016 acquisition of Midland Basin properties from Rock Oil Holdings, LLC (referred to as the "Rock Oil Acquisition") during the first quarter of 2017 by paying an additional \$7.4 million of cash consideration, resulting in total consideration of\$998.4 million paid after final closing adjustments. There were no material changes to the recorded basis of the proved and unproved properties acquired as a result of the final settlement.

Also, during the first quarter of 2017, and subsequent to March 31, 2017, the Company completed trades of properties, primarily unproved, in Howard and Martin Counties, Texas resulting in the Company acquiring approximately 3,020 net acres in exchange for 2,310 net acres. There was no cash consideration for the trade that was completed during the three months ended March 31, 2017, and this trade was recorded at the fair value of the assets surrendered with no gain or loss recognized.

#### Note 4 - Income Taxes

The income tax (expense) benefit recorded for the three months ended March 31, 2017, and 2016, differs from the amounts that would be provided by applying the statutory United States federal income tax rate to income or loss before income taxes primarily due to the effect of state income taxes, changes in valuation allowances, excess tax benefits and deficiencies from share-based payment awards, and accumulated impacts of other smaller permanent differences. The quarterly rate can also be affected by the proportional impacts of forecasted net income or loss as of each period end presented.

The provision for income taxes for the three months ended March 31, 2017, and 2016, consisted of the following:

	For the Three Months Ended March 31,			ed March 31,
		2017		2016
		(in tho	usands)	
Current portion of income tax (expense) benefit:				
Federal	\$	(7,439)	\$	_
State		(3,842)		(164)
Deferred portion of income tax (expense) benefit		(33,225)		195,039
Income tax (expense) benefit	\$	(44,506)	\$	194,875
Effective tax rate		37.4 %		35.9 %

On a year-to-date basis, a change in the Company's effective tax rate between reported periods will generally reflect differences in its estimated highest marginal state tax rate due to changes in the composition of income or loss from Company activities among multiple state tax jurisdictions. Cumulative effects of state tax rate changes are reflected in the period that legislation is enacted. As a result of adopting ASU 2016-09, excess tax benefits and deficiencies from share-based payment awards are expected to impact the Company's effective tax rate between periods. Please refer to Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards.

The change in the current portion of income tax expense relates to the effect of anticipated utilization of carryover net operating losses, deferred expenses, and carryover tax credits. The Company is generally no longer subject to United States federal or state income tax examinations by tax authorities for years before 2013, with the exception of its 2003 tax year.

#### Note 5 - Long-Term Debt

Credit Facility

The Company's Fifth Amended and Restated Credit Agreement, as amended (the "Credit Agreement"), provides for a maximum loan amount of 2.5 billion and has a maturity date of December 10, 2019.

On March 31, 2017, the Company entered into a Ninth Amendment to the Credit Agreement (the "Ninth Amendment") with its lenders. Pursuant to the Ninth Amendment, and as part of the regular, semi-annual borrowing base redetermination process, the borrowing base and aggregate lender commitments were reduced to \$925 million. This expected reduction was primarily due to the sale of the Company's outside-operated Eagle Ford shale assets in the first quarter of 2017 and the decrease in the value of the Company's proved reserves at December 31, 2016. The borrowing base redetermination process considers the value of both the Company's (a) proved oil and gas properties reflected in the Company's most recent reserve report and (b) commodity derivative contracts, each as determined by the Company's lender group. Additionally, the Ninth Amendment modified the Credit Agreement to allow the Company to enter into derivative contracts for an increased percentage of projected production volumes. As a result of the reduction to the Company's borrowing base and aggregate lender commitments, the Company recorded approximately \$1.1 million of expense related to the acceleration of unamortized deferred financing costs for the three months ended March 31, 2017. The next scheduled redetermination date is October 1, 2017.

The Company must comply with certain financial and non-financial covenants under the terms of the Credit Agreement, including covenants limiting dividend payments and requiring the Company to maintain certain financial ratios, as defined by the Credit Agreement. Financial covenants under the Credit Agreement require, as of the last day of each of the Company's fiscal quarters, the Company's (a) ratio of senior secured debt to 12-month trailing adjusted EBITDAX to be not more than 2.75 to 1.0; (b) adjusted current ratio to be not less than 1.0 to 1.0; and (c) ratio of 12-month trailing adjusted EBITDAX to interest expense to be not less than 2.0 to 1.0. The Company was in compliance with all financial and non-financial covenants under the Credit Agreement as of March 31, 2017, and through the filing of this report.

Interest and commitment fees are accrued based on a borrowing base utilization grid set forth in the Credit Agreement and presented in Note 5 – Long-Term Debt to the Company's consolidated financial statements in its 2016 Form 10-K. Eurodollar loans accrue interest at the London Interbank Offered Rate plus the applicable margin from the utilization table, and Alternate Base Rate and swingline loans accrue interest at the prime rate, plus the applicable margin from the utilization table. Commitment fees are accrued on the unused portion of the aggregate lender commitment amount and are included in interest expense in the accompanying statements of operations.

The following table presents the outstanding balance, total amount of letters of credit outstanding, and available borrowing capacity under the Credit Agreement as of April 26, 2017, March 31, 2017, and December 31, 2016:

	As of	April 26, 2017	As of March 31, 2017	As of December 31, 2016
	·		(in thousands)	
Credit facility balance (1)	\$	— \$	_	\$ —
Letters of credit (2)		200	200	200
Available borrowing capacity		924,800	924,800	1,164,800
Total aggregate lender commitment amount	\$	925,000 \$	925,000	\$ 1,165,000

<sup>(1)</sup> Unamortized deferred financing costs attributable to the credit facility are presented as a component of other noncurrent assets on the accompanying balance sheets and totaled \$4.3 million and \$5.9 million as of March 31, 2017, and December 31, 2016, respectively.

#### Senior Notes

The Company's Senior Notes consist of 6.50% Senior Notes due 2021, 6.125% Senior Notes due 2022, 6.50% Senior Notes due 2023, 5.0% Senior Notes due 2024, 5.625% Senior Notes due 2025, and 6.75% Senior Notes due 2026 (collectively referred to as "Senior Notes"). The Senior Notes, net of unamortized deferred financing costs line on the accompanying balance sheets as of March 31, 2017, and December 31, 2016, consisted of the following:

	As of March 31, 2017							As of December 31, 2016						
		Principal Amount		Unamortized Principal Deferred			of	nior Notes, Net Unamortized Deferred nancing Costs		Principal Amount		Unamortized Deferred nancing Costs	Senior Notes, No of Unamortized Deferred Financing Costs	
						(in tho	usano	ds)						
6.50% Senior Notes due 2021 (1) (2)	\$	344,611	\$	3,176	\$	341,435	\$	346,955	\$	3,372	\$	343,583		
6.125% Senior Notes due 2022 (2)		561,796		6,684		555,112		561,796		6,979		554,817		
6.50% Senior Notes due 2023 (2)		394,985		4,254		390,731		394,985		4,436		390,549		
5.0% Senior Notes due 2024		500,000		6,302		493,698		500,000		6,533		493,467		
5.625% Senior Notes due 2025		500,000		7,393		492,607		500,000		7,619		492,381		
6.75% Senior Notes due 2026		500,000		7,869		492,131		500,000		8,078		491,922		
Total	\$	2,801,392	\$	35,678	\$	2,765,714	\$	2,803,736	\$	37,017	\$	2,766,719		

<sup>(1)</sup> During the first quarter of 2017, the Company repurchased a total of \$2.3 million in aggregate principal amount of 6.50% Senior Notes due 2021 in open market transactions at a slight premium. The Company canceled all of these repurchased Senior Notes upon cash settlement.

<sup>(2)</sup> Letters of credit outstanding reduce the amount available under the credit facility on a dollar-for-dollar basis.

During the first quarter of 2016, the Company repurchased a total of \$46.3 million in aggregate principal amount of certain of its Senior Notes in open market transactions for a settlement amount of \$29.9 million, excluding interest, of which \$19.9 million was paid during the three months ended March 31, 2016, with the remaining \$10.0 million settlement amount being accrued at March 31, 2016, and paid in the second quarter of 2016. The Company recorded a net gain on extinguishment of debt of approximately \$15.7 million for the three months ended March 31, 2016. This amount includes a gain of approximately \$16.4 million associated with the discount realized upon repurchase, which was partially offset by approximately \$0.7 million related to the acceleration of unamortized deferred financing costs. The Company canceled all of these repurchased Senior Notes upon cash settlement.

The Senior Notes are unsecured senior obligations and rank equal in right of payment with all of the Company's existing and any future unsecured senior debt, and are senior in right of payment to any future subordinated debt. There are no subsidiary guarantors of the Senior Notes. The Company is subject to certain covenants under the indentures governing the Senior Notes and was in compliance with all covenants as of March 31, 2017, and through the filing of this report. The Company may redeem some or all of its Senior Notes prior to their maturity at redemption prices based on a premium, plus accrued and unpaid interest as described in the indentures governing the Senior Notes.

#### Senior Convertible Notes

On August 12, 2016, the Company issued \$172.5 million in aggregate principal amount of 1.50% Senior Convertible Notes due July 1, 2021 (the "Senior Convertible Notes"). The Senior Convertible Notes are unsecured senior obligations and rank equal in right of payment with all of the Company's existing and any future unsecured senior debt, and are senior in right of payment to any future subordinated debt.

The Senior Convertible Notes mature on July 1, 2021, unless earlier converted. Holders may convert their Senior Convertible Notes at their option at any time prior to January 1, 2021, only under certain circumstances as outlined in the indenture governing the Senior Convertible Notes and in *Note 5 – Long-Term Debt* to the Company's consolidated financial statements in its 2016 Form 10-K. On or after January 1, 2021, until the maturity date, holders may convert their Senior Convertible Notes at any time. The Company may not redeem the Senior Convertible Notes prior to the maturity date. Upon conversion, the Senior Convertible Notes may be settled, at the Company's common stock, cash, or a combination of cash and common stock. Holders may convert their notes based on a conversion rate of 24.6914 shares of the Company's common stock per \$1,000 principal amount of the Senior Convertible Notes, which is equal to an initial conversion price of approximately\$40.50 per share, subject to adjustment.

The Company has initially elected a net-settlement method to satisfy its conversion obligation. Under the net-settlement method, upon conversion, the Senior Convertible Notes will be settled, at the Company's election, in shares of the Company's common stock, cash, or a combination of cash and common stock. The Senior Convertible Notes were not convertible at the option of holders as of March 31, 2017, or through the filing of this report. Notwithstanding the inability to convert, the if-converted value of the Senior Convertible Notes as of March 31, 2017, did not exceed the principal amount.

At issuance, the Company recorded \$132.3 million as the initial carrying amount of the debt component, which approximated its fair value at issuance, and was estimated by using an interest rate for nonconvertible debt with terms similar to the Senior Convertible Notes. The effective interest rate used was 7.25 percent. The \$40.2 million excess of the principal amount of the Senior Convertible Notes over the fair value of the debt component was recorded as a debt discount and a corresponding increase in additional paid-in capital. The debt discount and debt-related issuance costs are amortized to the carrying value of the Senior Convertible Notes as interest expense through the maturity date of July 1, 2021. Interest expense recognized on the Senior Convertible Notes related to the stated interest rate and amortization of the debt discount totaled \$2.4 million for the three months ended March 31, 2017.

The net carrying amount of the liability component of the Senior Convertible Notes, as reflected on the accompanying balance sheets as ofMarch 31, 2017, and December 31, 2016, consisted of the following:

	A	s of March 31, 2017	As o	f December 31, 2016
Principal amount of Senior Convertible Notes	\$	172,500	\$	172,500
Unamortized debt discount		(35,713)		(37,513)
Unamortized deferred financing costs	<u> </u>	(3,898)		(4,131)
Net carrying amount	\$	132,889	\$	130,856

The Company is subject to certain covenants under the indenture governing the Senior Convertible Notes and was in compliance with all covenants as of March 31, 2017, and through the filing of this report.

#### Capped Call Transactions

In connection with the issuance of the Senior Convertible Notes, the Company entered into capped call transactions with affiliates of the underwriters of such issuance. The capped call transactions are generally expected to reduce the potential dilution upon conversion of the Senior Convertible Notes and/or partially offset any cash payments the Company is required to make in excess of the principal amount of converted Senior Convertible Notes in the event that the market price per share of the Company's common stock is greater than the strike price of the capped call transactions, which initially corresponds to the approximate \$40.50 per share conversion price of the Senior Convertible Notes. The cap price of the capped call transactions is initially \$60.00 per share. If the market price per share exceeds the cap price of the capped call transactions, there could be dilution or there would not be an offset of such potential cash payments.

#### Note 6 - Commitments and Contingencies

#### Commitments

During the first quarter of 2017, the Company completed the divestiture of its outside-operated Eagle Ford shale assets. Upon closing of the sale, the Company is no longer subject to gathering, processing, and transportation throughput commitments totaling 514 Bcf of natural gas, 52 MMBbl of oil, and 13 MMBbl of NGLs, or \$501.9 million of the potential undiscounted deficiency payments as of December 31, 2016.

As of March 31, 2017, the Company had total gathering, processing, and transportation throughput commitments with various third parties that require delivery of a minimum amount of 915 Bcf of natural gas, 17 MMBbl of crude oil, and 25 MMBbl of water through 2028. If the Company delivers no product, the aggregate undiscounted deficiency payments total approximately \$458.8 million. As of the filing of this report, the Company does not expect to incur any material shortfalls with regard to its remaining gathering, processing, and transportation throughput commitments.

Additionally, the Company entered into drilling rig contracts during the first quarter of 2017, and subsequent to March 31, 2017. As of the filing of this report, the Company's total drilling rig commitment was \$30.1 million; however, if the Company terminated these rig contracts early, penalties of \$19.0 million would be incurred instead.

There were no other material changes in commitments during the first quarter of 2017. Please refer to Note 6 - Commitments and Contingencies in the Company's 2016 Form 10-K for additional discussion of the Company's commitments.

#### Contingencies

The Company is subject to litigation and claims arising in the ordinary course of business. The Company accrues for such items when a liability is both probable and the amount can be reasonably estimated. In the opinion of management, the expected results of any pending litigation and claims will not have a material effect on the results of operations, the financial position, or the cash flows of the Company.

#### Note 7 - Compensation Plans

Performance Share Units Under the Equity Incentive Compensation Plan

The Company grants performance share units ("PSUs") to eligible employees as part of its long-term equity compensation program. The number of shares of the Company's common stock issued to settle PSUs ranges from 0% to 200% of the number of PSUs awarded and is determined based on certain performance criteria over athree-year measurement period. The performance criteria for PSUs are based on a combination of the Company's annualized Total Shareholder Return ("TSR") for the performance period and the relative performance of the Company's TSR compared with the annualized TSR of certain peer companies for the performance period. For the three months ended March 31, 2017, and 2016, total compensation expense recorded for PSUs was\$2.5 million and \$2.9 million, respectively, within general and administrative and exploration expense. As of March 31, 2017, there was \$14.2 million of total unrecognized compensation expense related to unvested PSU awards, which is being amortized through 2019. There have been no material changes to the outstanding and non-vested PSUs during the three months ended March 31, 2017.

Restricted Stock Units Under the Equity Incentive Compensation Plan

The Company grants restricted stock units ("RSUs") as part of its long-term equity compensation program. Each RSU represents a right to receive one share of the Company's common stock upon settlement of the award at the end of the specified vesting period. For the three months ended March 31, 2017, and 2016, total compensation expense recorded for RSUs was \$2.5 million and \$3.2 million, respectively, within general and administrative expense and exploration expense. As of March 31, 2017, there was \$12.9 million of total unrecognized compensation expense related to unvested RSU awards, which is being amortized through2019. There have been no material changes to the outstanding and non-vested RSUs during the three months ended March 31, 2017.

#### Note 8 - Pension Benefits

Pension Plans

The Company has a non-contributory defined benefit pension plan covering substantially all of its employees who joined the Company prior to January 1, 2015, and who meet age and service requirements (the "Qualified Pension Plan"). The Company also has a supplemental non-contributory pension plan covering certain management employees (the "Nonqualified Pension Plan" and together with the Qualified Pension Plan, the "Pension Plans"). The Company froze the Pension Plans to new participants, effective as of December 31, 2015. Employees participating in the Pension Plans as of December 31, 2015, continue to earn benefits.

Components of Net Periodic Benefit Cost for the Pension Plans

The following table presents the components of the net periodic benefit cost for the Pension Plans:

	For the	For the Three Months Ended March 31,					
		2017					
	(in thousands)						
Service cost	\$	2,050	\$	1,987			
Interest cost		727		624			
Expected return on plan assets that reduces periodic pension cost		(559)		(545)			
Amortization of prior service cost		4		4			
Amortization of net actuarial loss		396		372			
Net periodic benefit cost	\$	2,618	\$	2,442			

Prior service costs are amortized on a straight-line basis over the average remaining service period of active participants. Gains and losses in excess of 10 percent of the greater of the benefit obligation or the market-related value of assets are amortized over the average remaining service period of active participants.

Contributions

The Company contributed \$3.0 million to the Qualified Pension Plan during the three months ended March 31, 2017.

#### Note 9 - Earnings Per Share

Basic net income or loss per common share is calculated by dividing net income or loss available to common stockholders by the basic weighted-average number of common shares outstanding for the respective period. Diluted net income or loss per common share is calculated by dividing adjusted net income or loss by the diluted weighted-average number of common shares outstanding, which includes the effect of potentially dilutive securities. Potentially dilutive securities for this calculation consist of unvested RSUs, contingent PSUs, and shares into which the Senior Convertible Notes are convertible, which are measured using the treasury stock method.

PSUs represent the right to receive, upon settlement of the PSUs after the completion of thethree-year performance period, a number of shares of the Company's common stock that may range from zero to two times the number of PSUs granted on the award date. The number of potentially dilutive shares related to PSUs is based on the number of shares, if any, that would be issuable at the end of the respective reporting period, assuming that date was the end of the contingency period applicable to such PSUs.

On August 12, 2016, the Company issued \$172.5 million in aggregate principal amount of Senior Convertible Notes due 2021. Upon conversion, the Senior Convertible Notes may be settled, at the Company's election, in shares of the Company's common stock, cash, or a combination of cash and common stock. The Company has initially elected a net-settlement method to satisfy its conversion obligation, which allows the Company to settle the principal amount of the Senior Convertible Notes in cash and to settle the excess conversion value in shares, as well as cash in lieu of fractional shares. However, the Company has not made this a formal legal irrevocable election and thereby reserves the right to settle the Senior Convertible Notes in any manner allowed under the indenture as business conditions warrant. Shares of the Company's common stock traded at an average closing price below the \$40.50 conversion price for the three months endedMarch 31, 2017, and therefore, had no dilutive impact. In connection with the offering of the Senior Convertible Notes, the Company entered into capped call transactions with affiliates of the underwriters that would effectively prevent dilution upon settlement up to the \$60.00 cap price. The capped call transactions are not reflected in diluted net income per share, nor will they ever be, as they are anti-dilutive. Please refer to Note 5 - Long-Term Debt for additional discussion.

When the Company recognizes a loss from continuing operations, as was the case forthree months ended March 31, 2016, all potentially dilutive shares are anti-dilutive and are consequently excluded from the calculation of diluted net loss per common share.

The following table details the weighted-average dilutive and anti-dilutive securities for the periods presented:

ı 31,	onths Ended March 3	For the Three Months	31,
	2016	2017	
	thousands)	(in thous	
_	1	71	
49	_	_	49

The following table sets forth the calculations of basic and diluted earnings per share:

	For the Three Months Ended March 31,					
		2017	2016			
	(in t	housands, excep	t per sl	nare amounts)		
Net income (loss)	\$	74,434	\$	(347,210)		
Basic weighted-average common shares outstanding		111,258		68,077		
Add: dilutive effect of unvested RSUs and contingent PSUs		71		_		
Add: dilutive effect of 1.50% Senior Convertible Notes		_				
Diluted weighted-average common shares outstanding		111,329		68,077		
Basic net income (loss) per common share	\$	0.67	\$	(5.10)		
Diluted net income (loss) per common share	\$	0.67	\$	(5.10)		

#### Note 10 - Derivative Financial Instruments

Summary of Oil, Gas, and NGL Derivative Contracts in Place

The Company has entered into various commodity derivative contracts to mitigate a portion of its exposure to potentially adverse market changes in commodity prices and the associated impact on cash flows. As of March 31, 2017, all derivative counterparties were members of the Company's credit facility lender group and all contracts were entered into for other-than-trading purposes. The Company's derivative contracts consist of swap and collar arrangements for oil, gas, and NGLs. In a typical commodity swap agreement, if the agreed upon published third-party index price ("index price") is lower than the swap fixed price, the Company receives the difference between the index price and the agreed upon swap fixed price. If the index price is higher than the swap fixed price, the Company pays the difference. For collar arrangements, the Company receives the difference between an agreed upon index and the floor price if the index price is below the floor price. The Company pays the difference between the agreed upon ceiling price and the index price if the index price is above the ceiling price. No amounts are paid or received if the index price is between the floor and ceiling prices.

Oil Swaps

Contract Period	NYMEX WTI Volumes	Weighted-Average Contract Price						
	(MBbls)		(per Bbl)					
Second quarter 2017	1,444	\$	46.44					
Third quarter 2017	1,340	\$	46.66					
Fourth quarter 2017	1,254	\$	46.35					
Total	4,038							

Oil Collars

Contract Period	Weighted- NYMEX WTI Average Floor ct Period Volumes Price		Average Floor	Weighted- Average Ceiling Price
	(MBbls)		(per Bbl)	(per Bbl)
Second quarter 2017	636	\$	45.00	\$ 54.10
Third quarter 2017	583	\$	45.00	\$ 54.05
Fourth quarter 2017	540	\$	45.00	\$ 54.01
2018	3,343	\$	50.00	\$ 58.55
2019	2,660	\$	50.00	\$ 59.31
Total	7,762			

Oil Basis Swaps

Contract Period	Midland-Cushing Volumes	Weighted-Average Contract Price (1)					
	(MBbls)		(per Bbl)				
2018	2,080	\$	(1.27)				
2019	1,588	\$	(1.45)				
Total	3,668						

<sup>(</sup>I) Represents the price differential between WTI prices at Midland, Texas and WTI prices at Cushing, Oklahoma.

Natural Gas Swaps

Contract Period	Sold Volumes	Weighted-Average Contract Price	Purchased Volumes (1)	 Weighted- Average Contract Price	Net Volumes
	(BBtu)	(per MMBtu)	(BBtu)	(per MMBtu)	(BBtu)
Second quarter 2017	26,205	\$ 3.98	_	\$ _	26,205
Third quarter 2017	23,657	\$ 4.01	_	\$ _	23,657
Fourth quarter 2017	22,001	\$ 3.98	_	\$ _	22,001
2018	75,778	\$ 3.54	(30,606)	\$ 4.27	45,172
2019	32,016	\$ 4.02	(24,415)	\$ 4.34	7,601
Total (2)	179,657		(55,021)		124,636

During 2016, the Company restructured certain of its gas derivative contracts by buying fixed price volumes to offset existing 2018 and 2019 fixed price swap contracts totaling 55.0 million MMBtu. The Company then entered into new 2017 fixed price swap contracts totaling 38.6 million MMBtu with a contract price of \$4.43 per MMBtu. No other cash or other consideration was included as part of the restructuring.

Total net volumes of natural gas swaps are comprised of IF El Paso Permian (2%), IF HSC (95%), and IF NNG Ventura

<sup>(3%).</sup> 

	OPIS Pu Mon				ane Mont on-TET				OPIS Isob Belvieu		OPIS Natural Gasoline Mont Belvieu Non-TET			
Contract Period	Volumes		Veighted- Average Contract Price	Volumes	Weighted- Average Contract Price	Weighted- Average Contract Volumes Price		A		Weighted- Average Contract Price	Volumes		Veighted- Average Contract Price	
	(MBbls)	(	(per Bbl)	(MBbls)	(per Bbl)	(MBbls)		(per Bbl)	(MBbls)		(per Bbl)	(MBbls)	(	per Bbl)
Second quarter 2017	787	\$	8.86	634	\$ 21.90	182	\$	32.53	157	\$	33.38	249	\$	48.47
Third quarter 2017	736	\$	9.14	588	\$ 21.91	163	\$	32.42	140	\$	33.28	222	\$	48.43
Fourth quarter 2017	692	\$	9.10	550	\$ 21.91	149	\$	32.34	128	\$	33.23	203	\$	48.41
2018	2,434	\$	10.18	1,442	\$ 22.86	138	\$	35.41	119	\$	35.44	189	\$	49.40
2019	2,176	\$	11.95	_	\$ _	_	\$	_	_	\$	_	_	\$	_
2020	539	\$	11.13	_	\$ _	_	\$	_	_	\$	_	_	\$	_
Total	7,364			3,214		632			544			863		

Summary of Oil, Gas, and NGL Derivative Contracts Entered Into Subsequent toMarch 31, 2017

Subsequent to March 31, 2017, and through April 26, 2017, the Company entered into various derivative commodity contracts as summarized in the tables below.

## Oil Collars

Contract Period	NYMEX WTI Volumes	Weighted- Average Floor Price	Weighted- Average Ceiling Price
	(MBbls)	(per Bbl)	(per Bbl)
Fourth quarter 2017	546	\$ 50.00	\$ 58.08
2018	1,687	\$ 50.00	\$ 57.12
2019	468	\$ 50.00	\$ 56.20
Total	2.701		

### Oil Basis Swaps

Contract Period	Midland-Cushing Volumes	Weighted-Average Contract Price (1)					
	(MBbls)		(per Bbl)				
Third quarter 2017	566	\$	(1.62)				
Fourth quarter 2017	1,403	\$	(1.55)				
2018	3,584	\$	(1.50)				
2019	2,375	\$	(1.45)				
Total	7,928						

<sup>(1)</sup> Represents the price differential between WTI prices at Midland, Texas and WTI prices at Cushing, Oklahoma.

### Natural Gas Swaps

<b>Contract Period</b>	IF HSC Volumes	Weighted-Average Contract Price			
	(BBtu)		(per MMBtu)		
2018	17,236	\$	2.87		
2019	9,378	\$	2.88		
Total	26,614				

		Ethane Mont lvieu		opane Mont 1 Non-TET	0 0 - 1 - 0 - 1	al Butane Mont Non-TET  OPIS Isobutane Mont Belvieu Non-TET					
Contract Period	Volumes	Weighted- Average Contract Price	Volumes	Weighted- Average Contract Price	Volumes	Weighted- Average Contract Price	Volumes	Weighted- Average Contract Price	Volumes	Weighted- Average Contract Price	
	(MBbls)	(per Bbl)	(MBbls)	(per Bbl)	(MBbls)	(per Bbl)	(MBbls)	(per Bbl)	(MBbls)	(per Bbl)	
Second quarter 2017	105	10.97	_	\$ —	_	\$ —	_	\$ —	_	\$ —	
Third quarter 2017	170 \$	10.98	_	\$ —	_	\$ —	_	\$ —	_	\$ —	
Fourth quarter 2017	274 \$	11.04	_	\$ —	_	\$ —	_	\$ —	_	\$ —	
2018	1,157	12.23	345	\$ 26.04	87	\$ 31.71	69	\$ 30.35	116	\$ 47.36	
Total	1,706		345		87		69	-	116		

Derivative Assets and Liabilities Fair Value

The Company's commodity derivatives are measured at fair value and are included in the accompanying balance sheets as derivative assets and liabilities. The fair value of the commodity derivative contracts was a net asset of \$23.1 million as of March 31, 2017, and a net liability of \$91.7 million as of December 31, 2016.

The following tables detail the fair value of derivatives recorded in the accompanying balance sheets, by category:

		As of March 31, 2017							
	Derivative	<b>Derivative Assets</b>			Liabilities				
	Balance Sheet Classification	Fair Value		Balance Sheet Classification	F	air Value			
	•		(in tho	usands)					
Commodity contracts	Current assets	\$	73,978	Current liabilities	\$	53,809			
Commodity contracts	Noncurrent assets		84,195	Noncurrent liabilities		81,306			
Derivatives not designated as hedging instruments		\$	158,173		\$	135,115			

		As of December 31, 2016								
	Derivativ	<b>Derivative Assets</b>			Liabilities	ties				
	Balance Sheet Classification	Fair Value		Balance Sheet Classification	F	air Value				
			(in tho	usands)						
Commodity contracts	Current assets	\$	54,521	Current liabilities	\$	115,464				
Commodity contracts	Noncurrent assets		67,575	Noncurrent liabilities		98,340				
Derivatives not designated as hedging instruments		\$	122,096		\$	213,804				

### Offsetting of Derivative Assets and Liabilities

As of March 31, 2017, and December 31, 2016, all derivative instruments held by the Company were subject to master netting arrangements with various financial institutions. In general, the terms of the Company's agreements provide for offsetting of amounts payable or receivable between it and the counterparty, at the election of both parties, for transactions that settle on the same date and in the same currency. The Company's agreements also provide that in the event of an early termination, the counterparties have the right to offset amounts owed or owing under that and any other agreement with the same counterparty. The Company's accounting policy is to not offset these positions in its accompanying balance sheets.

The following table provides a reconciliation between the gross assets and liabilities reflected on the accompanying balance sheets and the potential effects of master netting arrangements on the fair value of the Company's derivative contracts:

Derivative Assets					Derivative Liabilities			
<u> </u>	As	_	As of					
March 31, 2017 December 31, 2016				Ma	rch 31, 2017	December 31, 2016		
(in thousands)								
\$	158,173	\$	122,096	\$	(135,115)	\$	(213,804)	
	(88,952)		(118,080)		88,952		118,080	
\$	69,221	\$	4,016	\$	(46,163)	\$	(95,724)	
	<b>Mar</b> \$	March 31, 2017  \$ 158,173 (88,952)	As of  March 31, 2017 Dece  \$ 158,173 \$ (88,952)	March 31, 2017         December 31, 2016           \$ 158,173         \$ 122,096           (88,952)         (118,080)	As of	As of     As       March 31, 2017     December 31, 2016     March 31, 2017       (in thousands)       \$ 158,173     \$ 122,096     \$ (135,115)       (88,952)     (118,080)     88,952	As of       March 31, 2017     December 31, 2016     March 31, 2017     December 31, 2016       (in thousands)       \$ 158,173     \$ 122,096     \$ (135,115)     \$ (88,952)       (88,952)     (118,080)     88,952	

The following table summarizes the components of the net derivative gain presented in the accompanying statements of operations:

	For the Three Months Ended March 31,							
		2017		2016				
		(in thou	usano	ds)				
Derivative settlement (gain) loss:								
Oil contracts	\$	9,084	\$	(99,992)				
Gas contracts		(17,506)		(41,053)				
NGL contracts		8,415		(5,983)				
Total derivative settlement gain	\$	(7)	\$	(147,028)				
Total net derivative (gain) loss:								
Oil contracts	\$	(49,590)	\$	(10,432)				
Gas contracts		(44,468)		(24,023)				
NGL contracts		(20,716)		20,227				
Total net derivative gain	\$	(114,774)	\$	(14,228)				

Credit Related Contingent Features

As of March 31, 2017, and through the filing of this report, all of the Company's derivative counterparties were members of the Company's credit facility lender group. Under the Credit Agreement and derivative contracts, the Company is required to secure mortgages on assets having a value equal to at least 90 percent of the total PV-9 of the Company's proved oil and gas properties evaluated in the most recent reserve report.

#### Note 11 - Fair Value Measurements

The Company follows fair value measurement accounting guidance for all assets and liabilities measured at fair value. This guidance defines fair value as the price that would be received to sell an asset or paid to transfer a liability (an exit price) in an orderly transaction between market participants at the measurement date. Market or observable inputs are the preferred sources of values, followed by assumptions based on hypothetical transactions in the absence of market inputs. The fair value hierarchy for grouping these assets and liabilities is based on the significance level of the following inputs:

- Level 1 quoted prices in active markets for identical assets or liabilities
- Level 2 quoted prices in active markets for similar assets or liabilities, quoted prices for identical or similar instruments in markets that are not active, and
  model-derived valuations whose inputs are observable or whose significant value drivers are observable
- Level 3 significant inputs to the valuation model are unobservable

The following table summarizes the Company's assets and liabilities that are measured at fair value in the accompanying balance sheets and where they are classified within the fair value hierarchy as of March 31, 2017:

	Le	evel 1	Level 2			Level 3			
	·	(in thousands)							
Assets:									
Derivatives (1)	\$	_	\$	158,173	\$	_			
Total property and equipment, net (2)	\$	_	\$	_	\$	443,772			
<u>Liabilities:</u>									
Derivatives (1)	\$	_	\$	135,115	\$	_			

- (1) This represents a financial asset or liability that is measured at fair value on a recurring basis
- (2) This represents a non-financial asset that is measured at fair value on a nonrecurring basis.

The following table summarizes the Company's assets and liabilities that are measured at fair value in the accompanying balance sheets and where they were classified within the fair value hierarchy as of December 31, 2016:

	Level 1		Level 2		Level 3			
	 (in thousands)							
Assets:								
Derivatives (1)	\$ _	\$	122,096	\$	_			
Total property and equipment, net (2)	\$ _	\$	_	\$	88,205			
<u>Liabilities:</u>								
Derivatives (1)	\$ _	\$	213,804	\$	_			

- (1) This represents a financial asset or liability that is measured at fair value on a recurring
- (2) This represents a non-financial asset that is measured at fair value on a nonrecurring basis.

Both financial and non-financial assets and liabilities are categorized within the above fair value hierarchy based on the lowest level of input that is significant to the fair value measurement. The following is a description of the valuation methodologies used by the Company as well as the general classification of such instruments pursuant to the above fair value hierarchy.

#### Derivatives

The Company uses Level 2 inputs to measure the fair value of oil, gas, and NGL commodity derivatives. Fair values are based upon interpolated data. The Company derives internal valuation estimates taking into consideration forward commodity price curves, counterparties' credit ratings, the Company's credit rating, and the time value of money. These valuations are then compared to the respective counterparties' mark-to-market statements. The considered factors result in an estimated exit-price that management believes provides a reasonable and consistent methodology for valuing derivative instruments. The derivative instruments utilized by the Company are not considered by management to be complex, structured, or illiquid. The oil, gas, and NGL commodity derivative markets are highly active.

Generally, market quotes assume that all counterparties have near zero, or low, default rates and have equal credit quality. However, an adjustment may be necessary to reflect the credit quality of a specific counterparty to determine the fair value of the instrument. The Company monitors the credit ratings of its counterparties and may require counterparties to post collateral if their ratings deteriorate. In some instances, the Company will attempt to novate the trade to a more stable counterparty. All of the Company's derivative counterparties are members of the Company's credit facility lender group.

Valuation adjustments are necessary to reflect the effect of the Company's credit quality on the fair value of any derivative liability position. This adjustment takes into account any credit enhancements, such as collateral margin that the Company may have posted with a counterparty, as well as any letters of credit between the parties. The methodology to determine this adjustment is consistent with how the Company evaluates counterparty credit risk, taking into account the Company's credit rating, current credit facility margins, and any change in such margins since the last measurement date.

The methods described above may result in a fair value estimate that may not be indicative of net realizable value or may not be reflective of future fair values and cash flows. While the Company believes that the valuation methods utilized are appropriate and consistent with authoritative accounting guidance and with other marketplace participants, the Company recognizes that third parties may use different methodologies or assumptions to determine the fair value of certain financial instruments that could result in a different estimate of fair value at the reporting date.

Refer to Note 10 - Derivative Financial Instruments for more information regarding the Company's derivative instruments.

Proved and Unproved Oil and Gas Properties and Other Property and Equipment

Property and equipment, net measured at fair value within the accompanying balance sheets totaled\$443.8 million as of March 31, 2017, and related to assets held for sale as further discussed below. Property and equipment, net measured at fair value totaled \$88.2 million as of December 31, 2016, and related primarily to downward performance reserve revisions on the Company's Powder River Basin assets at year-end.

Proved oil and gas properties. Proved oil and gas property costs are evaluated for impairment and reduced to fair value when there is an indication the carrying costs may not be recoverable. The Company uses Level 3 inputs and the income valuation technique, which converts future amounts to a single present value amount, to measure the fair value of proved properties through an application of discount rates and price forecasts representative of the current operating environment, as selected by the Company's management. The calculation of the discount rates are based on the best information available and the rates used ranged from 10 percent to 15 percent based on the reservoir specific weightings of future estimated proved and unproved cash flows as of March 31, 2017, and December 31, 2016. The Company believes the discount rates are representative of current market conditions and consider estimates of future cash payments, reserve categories, expectations of possible variations in the amount and/or timing of cash flows, the risk premium, and nonperformance risk. The prices for oil and gas are forecast based on NYMEX strip pricing, adjusted for basis differentials, for the first five years, after which a flat terminal price is used for each commodity stream. The prices for NGLs are forecast using OPIS Mont Belvieu pricing, for as long as the market is actively trading, after which a flat terminal price is used. Future operating costs are also adjusted as deemed appropriate for these estimates. There were no impairments on proved properties during the three months ended March 31, 2017. For the three months ended March 31, 2016, the Company recorded \$269.8 million of proved property impairments related primarily to the Company's outside-operated Eagle Ford shale assets and the decline in expected reserve cash flows driven by commodity price declines during the first quarter of 2016.

Unproved oil and gas properties. Unproved oil and gas property costs are evaluated for impairment and reduced to fair value when there is an indication that the carrying costs may not be recoverable. To measure the fair value of unproved properties, the Company uses a market approach, which takes into account the following significant assumptions: remaining lease terms, future development plans, risk weighted potential resource recovery, estimated reserve values, and estimated acreage value based on price(s) received for similar, recent acreage transactions by the Company or other market participants. There were no material abandonments or impairments of unproved properties during the three months ended March 31, 2017, or 2016.

Oil and gas properties held for sale. Proved and unproved properties and other property and equipment classified as held for sale, including the corresponding asset retirement obligation liability, are valued using a market approach based on an estimated net selling price, as evidenced by the most current bid prices received from third parties, if available, or by recent, comparable market transactions. If an estimated selling price is not available, the Company utilizes the various income valuation techniques discussed above. Any initial write-down and subsequent changes to the fair value less estimated cost to sell is included within the net gain (loss) on divestiture activity line item in the accompanying statements of operations. For the three months ended March 31, 2017, write-downs to fair value less estimated costs to sell on the Divide County assets held for sale totaled \$359.6 million. For the three months ended March 31, 2016, write-downs to fair value less estimated costs to sell on certain assets held for sale totaled \$68.3 million. Certain of these assets were subsequently sold in the third quarter of 2016 for a small net gain due to successful marketing efforts. Please refer to Note 3 – Divestitures, Assets Held for Sale, and Acquisitions for additional discussion.

# Long-Term Debt

The following table reflects the fair value of the Senior Notes and Senior Convertible Notes measured using Level 1 inputs based on quoted secondary market trading prices. These notes were not presented at fair value on the accompanying balance sheets as of March 31, 2017, or December 31, 2016, as they were recorded at carrying value, net of any unamortized discounts and deferred financing costs. Please refer to *Note 5 - Long-Term Debt* for additional discussion.

		As of Mar	, 2017		As of December 31, 2016					
	Princ	Principal Amount		Fair Value		incipal Amount		Fair Value		
		(in thousands)								
6.50% Senior Notes due 2021	\$	344,611	\$	353,657	\$	346,955	\$	354,546		
6.125% Senior Notes due 2022	\$	561,796	\$	572,330	\$	561,796	\$	570,925		
6.50% Senior Notes due 2023	\$	394,985	\$	403,378	\$	394,985	\$	403,134		
5.0% Senior Notes due 2024	\$	500,000	\$	472,500	\$	500,000	\$	475,975		
5.625% Senior Notes due 2025	\$	500,000	\$	479,550	\$	500,000	\$	485,000		
6.75% Senior Notes due 2026	\$	500,000	\$	503,750	\$	500,000	\$	516,565		
1.50% Senior Convertible Notes due 2021	\$	172,500	\$	171,355	\$	172,500	\$	202,189		

The carrying value of the Company's credit facility approximates its fair value, as the applicable interest rates are floating and based on prevailing market rates.

#### ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

This discussion includes forward-looking statements. Please refer to Cautionary Information about Forward-Looking Statements at the end of this item for important information about these types of statements.

#### Overview of the Company, Highlights, and Outlook

#### General Overview

We are an independent energy company engaged in the acquisition, exploration, development, and production of oil, gas, and NGLs in onshore North America. We currently have development positions in the Midland Basin, Eagle Ford shale, and Bakken/Three Forks resource plays. Our strategic objective is to become a premier operator of top tier assets. During 2016, and continuing into 2017, we cored up our portfolio through proved and unproved property acquisitions in the Midland Basin. We were able to accomplish this through the divestiture of non-core assets and successful financing transactions. We plan to continue coring up our portfolio so we can concentrate on our highest return programs and provide value through accelerated development activity. Our Midland Basin and Eagle Ford shale assets have high operating margins and significant opportunities for additional economic investment. We seek to maximize the value of our assets by applying industry leading technology and outstanding operational execution. Our portfolio is comprised of unconventional resource prospects with prospective drilling opportunities, which we believe provide for long-term production and reserves growth. We focus on achieving high full-cycle economic returns on our investments and maintaining a simple, strong balance sheet.

#### Oil, Gas, and NGL Prices

Our financial condition and the results of our operations are significantly affected by the prices we receive for our oil, gas, and NGL production, which can fluctuate dramatically. When we refer to realized oil, gas, and NGL prices below, the disclosed price represents the average price for the respective period, before the effects of derivative settlements, unless otherwise indicated. While quoted NYMEX oil and gas and OPIS NGL prices are generally used as a basis for comparison within our industry, the prices we receive are affected by quality, energy content, location, and transportation differentials for these products.

The following table summarizes commodity price data, as well as the effects of derivative settlements, for thefirst quarter of 2017, as well as the fourth and first quarters of 2016:

		For the Three Months Ended								
	M	arch 31, 2017		December 31, 2016		March 31, 2016				
Crude Oil (per Bbl):										
Average NYMEX contract monthly price	\$	51.91	\$	49.29	\$	33.41				
Realized price, before the effect of derivative settlements	\$	47.55	\$	43.58	\$	25.67				
Effect of oil derivative settlements	\$	(2.58)	\$	5.38	\$	24.27				
Natural Gas:										
Average NYMEX monthly settle price (per MMBtu)	\$	3.32	\$	2.98	\$	1.96				
Realized price, before the effect of derivative settlements (per Mcf)	\$	2.98	\$	2.86	\$	1.87				
Effect of natural gas derivative settlements (per Mcf)	\$	0.52	\$	0.35	\$	1.15				
NGLs (per Bbl):										
Average OPIS price (1)	\$	26.74	\$	24.11	\$	15.99				
Realized price, before the effect of derivative settlements	\$	22.06	\$	20.02	\$	11.76				
Effect of NGL derivative settlements	\$	(2.88)	\$	(3.10)	\$	1.78				

<sup>(1)</sup> Average OPIS prices per barrel of NGL, historical or strip, are based on a product mix of37% Ethane, 32% Propane, 6% Isobutane, 11% Normal Butane, and 14% Natural Gasoline for all periods presented. This product mix represents the industry standard composite barrel and does not necessarily represent our product mix for NGL production. Realized prices reflect our actual product mix.

We expect future prices for oil, gas, and NGLs to continue to be volatile. In addition to supply and demand fundamentals, as a global commodity, the price of oil is affected by real or perceived geopolitical risks in all regions of the world as well as the relative strength of the dollar compared to other currencies. Oil markets continue to be unstable due to over-supply, large inventory balances, and uncertainty in global demand. The increase in oil prices at the end of 2016 was primarily attributable to the Organization of Petroleum Exporting Countries ("OPEC") and non-OPEC exporting countries agreeing to cut production. While participating countries have largely adhered to agreed upon production cuts, uncertainty remains concerning whether these cuts will be sustained. Drilling activity in the United States has increased in recent months, putting continued downward pressure on oil prices in the near term.

There has been improvement in natural gas pricing over the last year, largely as a result of demand growth from gas fired power generation and both LNG exports and exports to Mexico exceeding prior expectations. We expect prices to remain near current levels in the near term as drilling rigs in operation have increased in recent months leading to increased supply, which we expect will be offset by continued demand growth from LNG exports and exports to Mexico. We also expect prices to fluctuate with changes in demand resulting from the weather.

NGL prices have also improved over the last year due to oil and natural gas price recovery. We expect NGL prices to remain near current levels through 2017, as we expect increased demand from export and petrochemical markets to be offset by increased drilling activity.

The following table summarizes 12-month strip prices for NYMEX WTI oil, NYMEX Henry Hub gas, and OPIS NGLs (same product mix as discussed under the table above) as of April 26, 2017, and March 31, 2017:

	As of a	April 26, 2017	As of March 31, 2017
NYMEX WTI oil (per Bbl)	\$	50.58	\$ 51.64
NYMEX Henry Hub gas (per MMBtu)	\$	3.39	\$ 3.36
OPIS NGLs (per Bbl)	\$	24.69	\$ 24.84

We use financial derivative instruments as part of our financial risk management program. We have a financial risk management policy governing our use of derivatives. The amount of our production covered by derivatives is driven by the amount of debt on our balance sheet, the level of capital commitments and long-term obligations we have in place, and our ability to enter into favorable derivative commodity contracts. With our current derivative contracts, we believe we have partially reduced our exposure to volatility in commodity prices in the near term. Our use of costless collars for a portion of our derivatives allows us to participate in some of the upward movements in oil prices while also setting a price floor for a portion of our oil production. Please refer to *Note 10 - Derivative Financial Instruments* in Part I, Item 1 of this report and the caption titled *Commodity Price Risk* in *Overview of Liquidity and Capital Resources* below for additional information regarding our oil, gas, and NGL derivatives.

First Quarter 2017 Highlights and Outlook for the Remainder of 2017

Our priorities for 2017 are to:

- demonstrate the value of our 2016 and 2017 acquisitions in the Midland Basin;
- generate high margin production growth from our operated acreage positions in the Midland Basin and Eagle Ford shale:
- successfully execute the sale of our outside-operated Eagle Ford shale and Divide County assets;
   and
- reduce our outstanding debt.

Our capital program for 2017, excluding acquisitions, is expected to be approximately \$875 million. By concentrating our capital on the highest return programs and operating at strong performance levels, we expect to generate higher company-wide margins and cash flow growth while creating value for our stockholders. We successfully closed the sale of our outside-operated Eagle Ford shale assets in the first quarter of 2017 for net divestiture proceeds of \$747.4 million. Additionally, we began marketing our Divide County assets, which we expect to sell by mid-year. We plan to use these divestiture proceeds to support the funding of our 2017 and 2018 capital programs. If market conditions are favorable, we may use a portion of the divestiture proceeds to pay down outstanding debt; however, the amount and timing is uncertain. Please refer to *Overview of Liquidity and Capital Resources* below for additional discussion on how we expect to fund our 2017 capital program.

Operational Activities. In our Midland Basin program, we began 2017 operating four drilling rigs and added two drilling rigs during the first quarter of 2017, one of which is drilling vertical test wells. We placed a seventh drilling rig into operation in mid-April 2017. Of these seven drilling rigs, five are focused on delineating and developing the Lower and Middle Spraberry and Wolfcamp A and B shale intervals on our acreage position in Howard and Martin Counties, Texas, and the other two drilling rigs remain focused on developing the Wolfcamp and Spraberry shale intervals on our Sweetie Peck property in Upton County, Texas. We expect approximately 80 percent of our 2017 capital program to be dedicated to our Midland Basin program.

During the first quarter of 2017, we acquired approximately 2,900 net acres in Howard County, Texas for\$59.6 million, subject to post-closing adjustments, and completed an acreage trade of approximately 760 net acres in Howard County, Texas. Additionally, subsequent to March 31, 2017, we completed another asset trade, primarily leasehold interests, in Howard and Martin Counties, Texas, whereby we received approximately 2,260 net acres in exchange for transferring approximately 1,550 net acres. These trades provide us the opportunity to drill longer lateral wells and increase our working interest in existing drilling units.

In our Eagle Ford shale program, we began running one drilling rig during the first quarter of 2017. We plan to add another drilling rig in the third quarter of 2017, and are focused on reducing our drilled but not completed well count and meeting lease obligations. We expect 20 percent of our 2017 capital program to be dedicated to our Eagle Ford shale program.

In our Powder River Basin program, we continued running one drilling rig during the first quarter of 2017 under an acquisition and development funding agreement with a third party, pursuant to which the third party is carrying our drilling and completion costs.

Costs Incurred in Oil and Gas Producing Activities. Costs incurred in oil and gas property acquisition, exploration and development activities, whether capitalized or expensed, totaled \$281.5 million for the three months ended March 31, 2017, and were incurred primarily in our Midland Basin and operated Eagle Ford shale programs. Of our total costs incurred for the three months ended March 31, 2017, \$85.8 million relates to property acquisitions, primarily unproved, in Howard County, Texas. Included in this acquisition amount is the fair value attributed to the properties surrendered in the non-monetary acreage trade completed during the first quarter of 2017, as discussed above.

Drilling and Completion Activity. The table below provides a summary of changes in our drilled but not completed well count and current year drilling and completion activity in our operated programs during the first quarter of 2017:

	Midland	Basin	Eagle For	rd Shale	Bakken Fork		Total	
	Gross	Net	Gross	Net	Gross	Net	Gross	Net
Wells drilled but not completed at December 31, 2016	17	17	47	47	20	17	84	81
Wells drilled	19	19	5	5	_	_	24	24
Wells completed	(16)	(16)	(17)	(17)			(33)	(33)
Wells drilled but not completed at March 31, 2017	20	20	35	35	20	17	75	72

Uring the first quarter of 2017, we announced plans to sell our Divide County

Production Results. The table below provides a regional breakdown of our production for the first quarter of 2017:

		outh Texas & Gulf	D 1 15 / 1	T
	Permian	Coast	Rocky Mountain	Total
Oil (MMBbl)	1.6	0.9	1.0	3.5
Gas (Bcf)	2.9	30.0	1.0	33.9
NGLs (MMBbl)	_	2.9	_	2.9
Equivalent (MMBOE)	2.1	8.8	1.2	12.1
Avg. daily equivalents (MBOE/d)	23.4	97.6	13.4	134.4
Relative percentage	17%	73 %	10%	100 %

Note: Amounts may not calculate due to rounding.

Production on an equivalent basis decreased 10 percent for the three months ended March 31, 2017, compared with the same period in 2016, primarily as a result of the divestitures of properties across our regions in the last half of 2016 and the first quarter of 2017, specifically our Raven/Bear Den and outside-operated Eagle Ford shale assets. Please refer to A Three-Month Overview of Selected Production and Financial Information, Including Trends and Comparison of Financial Results and Trends Between the Three Months Ended March 31, 2017, and 2016 below for additional discussion on production.

Financial Results. In the first quarter of 2017, we had the following financial results:

- We recorded net income of \$74.4 million, or \$0.67 per diluted share, for the three months ended March 31, 2017, compared with a net loss of \$347.2 million, or \$5.10 per diluted share, for the three months ended March 31, 2016. Net income for the three months ended March 31, 2017, was driven largely by a net gain of \$398.1 million recorded on the sale of our outside-operated Eagle Ford shale assets and a net derivative gain of \$114.8 million, mostly offset by a \$359.6 million write-down to estimated fair value less costs to sell on our Divide County assets held for sale. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended March 31, 2017, and 2016 below for additional discussion regarding the components of net income (loss) for each period.
- We had net cash provided by operating activities of \$135.0 million for the three months ended March 31, 2017, compared with \$118.3 million for the same period in 2016. Please refer to Analysis of Cash Flow Changes Between the Three Months Ended March 31, 2017, and 2016 below for additional discussion.
- Adjusted EBITDAX, a non-GAAP financial measure, for the three months endedMarch 31, 2017, was \$172.2 million, compared with \$182.3 million for the
  same period in 2016. Please refer to Non-GAAP Financial Measures below for additional discussion, including our definition of adjusted EBITDAX and
  reconciliations of our net income (loss) and net cash provided by operating activities to adjusted EBITDAX.
- On March 31, 2017, pursuant to the Ninth Amendment to our Credit Agreement, our borrowing base and aggregate lender commitments were reduced to \$925 million during the regularly scheduled semi-annual redetermination and as a result of closing the sale of our outside-operated Eagle Ford shale assets. Please refer to Note 5 Long-Term Debt in Part I, Item I of this report for additional discussion.

## Financial Results of Operations and Additional Comparative Data

The tables below provide information regarding selected production and financial information. A detailed discussion follows.

	For the Three Months Ended							
		March 31,		December 31,		September 30,		June 30,
		2017		2016		2016		2016
				(in millions, except	for <sub>J</sub>	production data)		
Production (MMBOE)		12.1		13.4		14.2		14.3
Oil, gas, and NGL production revenue	\$	333.2	\$	346.3	\$	329.2	\$	291.1
Oil, gas, and NGL production expense	\$	138.0	\$	151.9	\$	152.5	\$	148.6
Depletion, depreciation, amortization, and asset retirement obligation liability accretion	\$	137.8	\$	171.6	\$	194.0	\$	211.0
Exploration	\$	12.0	\$	23.7	\$	13.5	\$	13.2
General and administrative	\$	29.2	\$	33.3	\$	32.7	\$	28.2
Net income (loss)	\$	74.4	\$	(200.9)	\$	(40.9)	\$	(168.7)

Note: Amounts may not calculate due to rounding.

## **Selected Performance Metrics**

	For the Three Months Ended							
		March 31,		December 31,		September 30,		June 30,
		2017		2016		2016		2016
Average net daily production equivalent (MBOE per day)		134.4		145.6		153.9		157.2
Lease operating expense (per BOE)	\$	3.82	\$	3.67	\$	3.29	\$	3.31
Transportation costs (per BOE)	\$	5.88	\$	6.39	\$	6.24	\$	5.95
Production taxes as a percent of oil, gas, and NGL production								
revenue		4.2%		4.3 %		4.5 %		4.6%
Ad valorem tax expense (per BOE)	\$	0.55	\$	0.17	\$	0.21	\$	0.19
Depletion, depreciation, amortization, and asset retirement obligation								
liability accretion (per BOE)	\$	11.39	\$	12.81	\$	13.70	\$	14.75
General and administrative (per BOE)	\$	2.42	\$	2.49	\$	2.31	\$	1.97

Note: Amounts may not calculate due to rounding.

# A Three-Month Overview of Selected Production and Financial Information, Including Trends

	Fo	For the Three Months Ended March 31,		Amount Change		Percent Change	
		2017		2016		en Periods	Between Periods
Net production volumes (1)	· <u> </u>					,	
Oil (MMBbl)		3.5		4.1		(0.6)	(14)%
Gas (Bcf)		33.9		35.7		(1.8)	(5)%
NGLs (MMBbl)		2.9		3.3		(0.4)	(13)%
Equivalent (MMBOE)		12.1		13.4		(1.3)	(10)%
Average net daily production <sup>(1)</sup>							
Oil (MBbl per day)		39.2		45.3		(6.1)	(13)%
Gas (MMcf per day)		376.6		392.2		(15.6)	(4)%
NGLs (MBbl per day)		32.5		36.8		(4.4)	(12)%
Equivalent (MBOE per day)		134.4		147.5		(13.1)	(9)%
Oil, gas, and NGL production revenue (in millions)							
Oil production revenue	\$	167.6	\$	105.8	\$	61.8	58 %
Gas production revenue		101.2		66.6		34.6	52 %
NGL production revenue		64.4		39.4		25.0	63 %
Total	\$	333.2	\$	211.8	\$	121.4	57 %
Oil, gas, and NGL production expense (in millions)	<del></del>						
Lease operating expense	\$	46.1	\$	50.8	\$	(4.7)	(9)%
Transportation costs		71.1		81.3		(10.2)	(13)%
Production taxes		14.1		8.9		5.2	58 %
Ad valorem tax expense		6.7		3.5		3.2	91 %
Total	\$	138.0	\$	144.5	\$	(6.5)	(4)%
Realized price (before the effect of derivative settlements)							
Oil (per Bbl)	\$	47.55	\$	25.67	\$	21.88	85 %
Gas (per Mcf)	\$	2.98	\$	1.87	\$	1.11	59 %
NGLs (per Bbl)	\$	22.06	\$	11.76	\$	10.30	88 %
Per BOE	\$	27.55	\$	15.78	\$	11.77	75 %
Per BOE data (1)							
Production costs:							
Lease operating expense	\$	3.82	\$	3.79	\$	0.03	1 %
Transportation costs	\$	5.88	\$	6.06	\$	(0.18)	(3)%
Production taxes	\$	1.17	\$	0.66	\$	0.51	77 %
Ad valorem tax expense	\$	0.55	\$	0.27	\$	0.28	104 %
	\$	2.42	\$	2.40	\$	0.02	1 %
General and administrative							
Depletion, depreciation, amortization, and asset retirement obligation liability accretion	\$	11.39	\$	15.96	\$	(4.57)	(29)%
Derivative settlement gain (2)	\$	_	\$	10.96	\$	(10.96)	(100)%
Earnings per share information							
Basic net income (loss) per common share	\$	0.67	\$	(5.10)	\$	5.77	113 %
Diluted net income (loss) per common share	\$	0.67	\$	(5.10)	\$	5.77	113 %
Basic weighted-average common shares outstanding (in thousands)		111,258		68,077		43,181	63 %
Diluted weighted-average common shares outstanding (in thousands)		111,329		68,077		43,252	64 %

- (1) Amount and percentage changes may not calculate due to rounding
- (2) Derivative settlements for the three months ended March 31, 2017, and 2016, respectively, are included within the net derivative gain line item in the accompanying statements of operations.

We present per BOE information because we use this information to evaluate our performance relative to our peers and to identify and measure trends we believe may require additional analysis. Average net daily production for the three months ended March 31, 2017, decreased nine percent compared with the same period in 2016 primarily as a result of the divestitures of properties across our regions in the last half of 2016 and the first quarter of 2017, specifically our Raven/Bear Den and outside-operated Eagle Ford shale assets. We anticipate a further decrease in production upon the expected divestiture of our Divide County assets later in the year; however, as we ramp up our Midland Basin development program, we expect production to increase and begin offsetting this decline. Overall, we expect a decrease in production for full-year 2017 compared with full-year 2016. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended March 31, 2017, and 2016 below for additional discussion.

Changes in production volumes, revenues, and costs reflect the highly volatile nature of our industry. Our realized price before the effects of derivative settlements on a per BOE basis for the three months ended March 31, 2017, increased 75 percent compared with the same period in 2016. Commodity prices were at recent historic lows in early 2016 and began to improve in the last half of 2016 and are holding relatively flat in the first quarter of 2017. For the three months ended March 31, 2017, we had a minimal gain on the settlement of our derivative contracts, which resulted in no gain on a per BOE basis. This compares with a gain of \$10.96 per BOE for the three months ended March 31, 2016.

Lease operating expense ("LOE") on a per BOE basis increased slightly for thethree months ended March 31, 2017, compared with the same period in 2016. We experience volatility in our LOE as a result of the impact industry activity has on service provider costs and seasonality in workover expense. For 2017, we generally expect LOE on a per BOE basis to be relatively flat compared with 2016. We expect that any increase in service provider costs resulting from increased development activity in the Midland Basin will be offset by the executed and planned divestitures of our higher cost Williston Basin properties.

Transportation expense on a per BOE basis decreased slightly for the three months ended March 31, 2017, compared with the same period in 2016 primarily as a result of selling our outside-operated Eagle Ford shale assets in early March 2017. In general, we expect transportation costs on a per BOE basis to decrease further in 2017 with the sale of our outside-operated Eagle Ford shale assets and as our Midland Basin assets become a larger portion of our production mix. The majority of our Midland Basin production is sold at the wellhead under current contracts, and therefore, there is minimal transportation expense separately recorded on the accompanying statements of operations.

Production taxes on a per BOE basis increased 77 percent for the three months ended March 31, 2017, compared with the same period in 2016 in line with the increase in our realized price before the effect of derivative settlements, as there was no change in our production tax rate of 4.2 percent period-over-period. We generally expect production tax expense to trend with oil, gas, and NGL production revenue on an absolute and per BOE basis. Product mix, the location of production, and incentives to encourage oil and gas development can also impact the amount of production tax we recognize. For 2017, we generally expect our production tax rate to decrease year-over-year as a result of our closed and anticipated divestitures; however, we expect an increase in production taxes on a per BOE basis in line with improved commodity prices.

Ad valorem tax expense on a per BOE basis increased 104 percent for the three months ended March 31, 2017, compared with the same period in 2016 primarily as a result of changes in our asset and production base. The majority of our ad valorem tax expense is related to our Texas properties. Since we have acquired producing properties in Texas and divested properties in our Rocky Mountain region, we expect ad valorem tax expense on an absolute and per BOE basis to increase in 2017. Additionally, we expect an increase in commodity price assumptions used in 2017 property tax valuations.

General and administrative ("G&A") expense on a per BOE basis increased slightly for thethree months ended March 31, 2017, compared with the same period in 2016. The decrease in absolute G&A expense was in line with the decrease in production volumes. We expect G&A expense on an absolute basis to remain relatively flat in 2017 compared with 2016 as reduced headcount in 2016 is expected to be offset by headcount changes resulting from recent and anticipated acquisition and divestiture activity and an expected increase in base and short-term incentive compensation. However, we expect an overall increase in G&A expense on a per BOE basis in 2017 due to the decrease in production volumes.

Depletion, depreciation, amortization, and asset retirement obligation liability accretion ("DD&A") expense on a per BOE basis decreased percent for the three months ended March 31, 2017, compared with the same period in 2016, as a result of divested assets and the impact of assets held for sale, specifically our outside-operated Eagle Ford shale assets that were held for sale prior to being sold in March 2017 and our Divide County assets which we classified as held for sale during the first quarter of 2017. These assets were not depleted while classified as held for sale. Our DD&A rate fluctuates as a result of impairments, planned and closed divestitures, and changes in the mix of our production and the underlying proved reserve volumes. In general, we expect DD&A expense on a per BOE basis to decrease in 2017 due to selling our higher cost Raven/Bear Den assets in late 2016 and assets held for sale impacts.

Please refer to Comparison of Financial Results and Trends Between the Three Months Ended March 31, 2017, and 2016 below for additional discussion on operating expenses.

Please refer to *Note 9 - Earnings Per Share* in Part I, Item 1 of this report for discussion on the types of shares included in our basic and diluted net income (loss) per common share calculations. Our basic and diluted weighted-average share count increased for the three months ended March 31, 2017, compared with the same period in 2016 due to the public and private equity offerings of our common stock made in the last half of 2016. Additionally, we recorded net income for the three months ended March 31, 2017, and therefore, our unvested RSUs and contingent PSUs were dilutive for this period.

#### Comparison of Financial Results and Trends Between the Three Months Ended March 31, 2017, and 2016

Oil, gas, and NGL production, production revenues, and production costs

The following table presents the regional changes in our oil, gas, and NGL production, production revenues, and production costs between the three months ended March 31, 2017, and 2016:

	Average Net Daily Production Increase (Decrease)	tion Production Revenues Prod		Production Costs Increase (Decrease)	
	(MBOE/d)		(in millions)		(in millions)
Permian	17.3	\$	79.3	\$	16.3
South Texas & Gulf Coast	(13.7)	\$	52.1	\$	(6.3)
Rocky Mountain	(16.7)	\$	(10.0)	\$	(16.5)
Total	(13.1)	\$	121.4	\$	(6.5)

The 10 percent decrease in net equivalent production volumes primarily due to recent divestitures was partially offset by a75 percent increase in realized prices on a per BOE basis resulting in a 57 percent increase in oil, gas, and NGL production revenues between the three months endedMarch 31, 2017, and 2016. This decrease in net equivalent production volumes, as well as the changes in costs on a per BOE basis discussed above, resulted in a four percent decrease in total production costs for the three months ended March 31, 2017, compared with the same period in 2016. Offsetting the decrease in production volumes and costs in our South Texas & Gulf Coast and Rocky Mountain regions due to recent divestitures is an increase in production volumes, revenues, and costs in our Permian region due to increased drilling and completion activity in our Midland Basin development program. Please refer to A Three-Month Overview of Selected Production and Financial Information, Including Trends above for discussion of trends on a per BOE basis.

Net gain (loss) on divestiture activity

	For the Three	For the Three Months Ended March 31,					
	2017		2016				
		in millions)					
/	\$ 3	37.5 \$	(69.0)				

The net gain on divestiture activity for the three months endedMarch 31, 2017, was a result of the \$398.1 million net gain recorded on the sale of our outside-operated Eagle Ford shale assets, partially offset by a write-down to fair value less estimated selling costs recorded on the Divide County assets held for sale. Please refer to *Note 3* – *Divestitures, Assets Held for Sale, and Acquisitions* in Part I, Item 1 of this report for additional discussion. The net loss on divestiture activity for the three months ended March 31, 2016, was largely due to the write-down to fair value less estimated selling costs recorded on certain assets held for sale that were sold in the third quarter of 2016.

Depletion, depreciation, amortization, and asset retirement obligation liability accretion

	For the Three Months Ended March				
	2017			2016	
		(in m	illions)	_	
Depletion, depreciation, amortization, and asset retirement obligation liability					
accretion	\$	137.8	\$	214.2	

DD&A expense decreased 36 percent for the three months ended March 31, 2017, compared with the same period in 2016 due to the decline in our production volumes and the impact of assets sold and assets held for sale. Please refer to the section A Three-Month Overview of Selected Production and Financial Information, Including Trends above for further discussion of DD&A expense on a per BOE basis.

Exploration

For the	Three Months E	nded March 31,
2	017	2016
	(in millions	s)
\$	12.0 \$	15.3

Exploration expense for the three months ended March 31, 2017, decreased 22 percent compared with the same period in 2016 due to reduced overhead costs as a result of lower headcount. Overall, we expect our geological and geophysical and exploration overhead expenses to be higher for full-year 2017 than full-year 2016 as we focus on testing and delineating our recently acquired Midland Basin acreage.

Impairment of proved properties and abandonment and impairment of unproved properties

	For the Three Months Ended March				
	2	017	2016		
		(in millions)	_		
Impairment of proved properties	\$	— \$	269.8		
Abandonment and impairment of unproved properties	\$	— \$	2.3		

There were no proved or unproved property impairments recorded for thethree months ended March 31, 2017. Proved property impairments recorded for the three months ended March 31, 2016, were a result of continued commodity price declines, largely impacting proved properties in our outside-operated Eagle Ford shale program. These assets were recently sold for a significant gain due to a recovery in commodity prices and successful marketing efforts. Please refer to the caption *Net gain (loss) on divestiture activity* above. Unproved property impairments recorded for the three months ended March 31, 2016, were due to lease expirations on acreage we no longer intended to develop.

We expect proved property impairments to be more likely to occur in periods of declining or depressed commodity prices, and unproved property impairments to fluctuate with the timing of lease expirations, unsuccessful exploration activities, and changing economics associated with volatile commodity prices. Additionally, changes in drilling plans, downward engineering revisions, or unsuccessful exploration efforts may result in proved and unproved property impairments. Any amount of future impairment is difficult to predict, but based on updated commodity price assumptions as of April 26, 2017, we do not expect any material impairments in the second quarter of 2017 due to commodity price impacts.

	1	For the Three Months Ended March 3					
	_	2017		2016		6	
	_		(in m	illions)			
General and administrative	\$		29.2	\$		32.2	

G&A expense decreased nine percent for the three months ended March 31, 2017, compared with the same period in 2016, primarily due to reduced headcount in the last half of 2016. However, we do expect changes in headcount in 2017 resulting from recent and anticipated acquisition and divestiture activity, as well as an increase in base and short-term incentive compensation. Please refer to the section A Three-Month Overview of Selected Production and Financial Information, Including Trends above for further discussion of G&A expense on a per BOE basis.

Net derivative gain

	For the	For the Three Months Ended March 31,				
		2017	2016			
		(in millions)				
Vet derivative gain	\$	(114.8) \$	(14.2)			

We recognized a derivative gain of \$114.8 million for the three months ended March 31, 2017, which is primarily a result of a \$94.3 million mark-to-market gain recorded on contracts remaining as of March 31, 2017, due to a decrease in commodity strip prices from December 31, 2016. For contracts settled during the first quarter of 2017, the fair value was a net liability of \$20.5 million at December 31, 2016, and net cash settlements were minimal, resulting in a \$20.5 million gain. We recognized a \$14.2 million derivative gain for the three months endedMarch 31, 2016, due to favorable cash settlements on contracts settled in the first quarter of 2016, partially offset by a small mark-to-market loss on remaining contracts as of March 31, 2016. Please refer to *Note 10 - Derivative Financial Instruments* in Part I, Item 1 of this report for additional information.

Other operating expenses

	For the T	For the Three Months Ended March 31,				
	20	2017		2016		
		(in millions)				
Other operating expenses	\$	4.9	\$	5.7		

Other operating expenses for the three months ended March 31, 2017, consisted primarily of losses on materials inventory sales, whereas other operating expenses for the same period in 2016 consisted primarily of drilling rig termination and standby fees.

Gain (loss) on extinguishment of debt

	For the T	For the Three Months Ended March 31,		
	201	17	2016	
		(in millions)		
Gain (loss) on extinguishment of debt	\$	— \$	15.7	

During the first quarter of 2016, we recorded a \$15.7 million net gain on the early extinguishment of a portion of our Senior Notes, which included approximately \$16.4 million associated with the discount realized upon repurchase, slightly offset by approximately \$0.7 million related to the acceleration of unamortized deferred financing costs. Please refer to *Note 5 - Long-Term Debt* in Part I, Item 1 of this report for additional information.

Fo	For the Three Months Ended March 31,		
	2017	2	2016
	(in mi	llions)	
\$	(47.0)	\$	(31.1)

There was a 51 percent increase in interest expense for the three months endedMarch 31, 2017, compared with the same period in 2016, primarily due to the additional debt issued in 2016, as presented in *Note 5 - Long-Term Debt* in Part I, Item I of this report, and an increase in our weighted-average interest rate, as discussed and presented in *Overview of Liquidity and Capital Resources*.

Income tax (expense) benefit

	For the	For the Three Months Ended March 31,		
		2017	2016	
	(	(in millions, except tax rate)		
benefit	\$	(44.5) \$	194.9	
		37.4%	35.9 %	

The increase in the effective tax rate for the three months ended March 31, 2017, compared with the same period in 2016, resulted from 2017 state apportionment changes due to divesting our outside-operated Eagle Ford shale assets, 2016 valuation allowance increases correlating from various 2016 projected state net operating losses, which decreased the 2016 effective tax rate, versus 2017 valuation allowance decreases resulting from projected utilization of various state net operating losses, which increased the 2017 effective tax rate. Please refer to *Note 4 - Income Taxes* in Part I, Item 1 of this report for additional discussion.

### Overview of Liquidity and Capital Resources

Based on the current commodity price environment, we believe we have sufficient liquidity and capital resources to execute our business plan for the foreseeable future. We continue to manage the duration and level of our drilling and completion service commitments to maintain the flexibility to adjust our activity and capital expenditures in periods of prolonged weak commodity prices or to respond should commodity prices recover further.

Sources of Cash

We currently expect our 2017 capital program to be funded by cash flows from operations and proceeds from the divestiture of properties.

Although we anticipate cash flows from these sources will be sufficient to fund our expected 2017 capital program, we may also elect to borrow under our Credit Agreement and/or raise funds through debt or equity financings or from other sources or enter into carrying cost funding and sharing arrangements with third parties for particular exploration and/or development programs. See *Credit Agreement* below for discussion of the recent reduction in our borrowing base. Our borrowing base could be further reduced as a result of lower commodity prices, divestitures of proved properties, or newly issued debt. Decreases in commodity prices have limited our industry's access to capital markets in recent periods. If we raise additional funds through the issuance of equity or convertible debt securities, the percentage ownership of our stockholders could be diluted, and these newly-issued securities may have rights, preferences, or privileges senior to those of existing stockholders. Any future downgrades in our credit ratings may make it more difficult or expensive for us to borrow additional funds. All of our sources of liquidity can be impacted by the general condition of the broader economy and by fluctuations in commodity prices, operating costs, and volumes produced, all of which affect us and our industry.

We have no control over the market prices for oil, gas, or NGLs, although we may be able to influence the amount of our realized revenues from our oil, gas, and NGL sales through the use of derivative contracts as part of our commodity price risk management program. Please refer to *Note 10 - Derivative Financial Instruments* in Part I, Item 1 of this report for additional information about our oil, gas, and NGL derivative contracts currently in place and the timing of settlement of those contracts.

Proposals to reform the Internal Revenue Code of 1986, as amended, which include eliminating or reducing current tax deductions for intangible drilling costs, depreciation of equipment acquisition costs, the domestic production activities deduction, percentage depletion, and other deductions that reduce our taxable income, have been discussed in past years. Although we believe this possibility has decreased with the recent congressional discussions on tax reform, we expect that future legislation eliminating these deductions would reduce net operating cash flows over time, thereby reducing funding available for our exploration and development capital programs, as well as funding available to our peers in the industry for similar programs. If enacted, reductions in available deductions could have a significant adverse effect on drilling in the United States for a number of years.

#### Credit Agreement

Our Credit Agreement provides for a maximum loan amount of \$2.5 billion and has a maturity date of December 10, 2019. On March 31, 2017, we entered into a Ninth Amendment to the Credit Agreement. Pursuant to the Ninth Amendment, and as part of the regular, semi-annual borrowing base redetermination process, the borrowing base and current aggregate lender commitments were decreased to \$925 million. This expected decrease was primarily due to the divestiture of our outside-operated Eagle Ford shale properties in the first quarter of 2017. Additionally, as part of the Ninth Amendment, we are now able to enter into derivative contracts for an increased percentage of projected production volumes. We had a zero balance on our credit facility as of March 31, 2017, and as of the filing of this report. We anticipate our borrowing base to be further reduced upon divesting our Divide County assets, which we expect to occur prior to the October 1, 2017, scheduled borrowing base redetermination. No individual bank that is a party to our Credit Agreement represents more than 10 percent of the lender commitments. Please refer to *Note 5 - Long-Term Debt* in Part I, Item 1 of this report for additional discussion

We must comply with certain financial and non-financial covenants under the Credit Agreement, including covenants limiting dividend payments and requiring us to maintain certain financial ratios, as defined by the Credit Agreement. Financial covenants under the Credit Agreement require, as of the last day of each fiscal quarter, our (a) ratio of senior secured debt to 12-month trailing adjusted EBITDAX to be not more than 2.75 to 1.0; (b) adjusted current ratio to be not less than 1.0 to 1.0; and (c) ratio of 12-month trailing adjusted EBITDAX to interest expense to be not less than 2.0 to 1.0. We were in compliance with all financial and non-financial covenants under the Credit Agreement as of March 31, 2017, and through the filing of this report. Please refer to the caption *Non-GAAP Financial Measures* below for the calculation of adjusted EBITDAX and reconciliations of net income (loss) and net cash provided by operating activities to adjusted EBITDAX.

Our daily weighted-average credit facility debt balance was approximately \$52.9 million and \$252.0 million for the three months ended March 31, 2017, and 2016, respectively. Cash flows provided by our operating activities, proceeds received from divestitures of properties, capital markets activities, and the amount of our capital expenditures, including acquisitions, all impact the amount we have borrowed under our credit facility.

#### Weighted-Average Interest Rates

Our weighted-average interest rates include paid and accrued interest, fees on the unused portion of the credit facility's aggregate commitment amount, letter of credit fees, the non-cash amortization of deferred financing costs, and the non-cash amortization of the discount related to the Senior Convertible Notes. Our weighted-average borrowing rates include paid and accrued interest only.

The following table presents our weighted-average interest rates and our weighted-average borrowing rates for thethree months ended March 31, 2017, and 2016:

	For the Three Months I	For the Three Months Ended March 31,		
	2017	2016		
Weighted-average interest rate	6.6%	5.9%		
Weighted-average borrowing rate	5.8%	5.5%		

The increase in our weighted-average interest rate and weighted-average borrowing rate for the three months ended March 31, 2017, compared with the same period in 2016, is largely due to the Senior Convertible Notes and 6.75% Senior Notes due 2026 issued during the third quarter of 2016. Further impacting these rates in 2017 and 2016 is the timing and amount of Senior Notes redemptions, changes in our aggregate lender commitment amount on our credit facility, and the average balance on our credit facility. The rates disclosed in the above table do not reflect amounts associated with the repurchase of Senior Notes, such as the discount realized or premium paid upon repurchase or acceleration of unamortized deferred financing costs expensed upon repurchase. Please refer to *Note 5 - Long-Term Debt* in Part I, Item 1 of this report for additional discussion.

#### Uses of Cash

We use cash for the acquisition, exploration, and development of oil and gas properties and for the payment of operating and general and administrative costs, income taxes, dividends, and debt obligations, including interest. Expenditures for the acquisition, exploration, and development of oil and gas properties are the primary use of our capital resources. For the three months ended March 31, 2017, we spent \$229.5 million in capital expenditures and in acquiring proved and unproved oil and gas properties. This amount differs from the costs incurred amount, which is accrual-based and includes asset retirement obligations, geological and geophysical expenses, exploration overhead amounts, and the fair value of assets acquired in a non-monetary trade.

The amount and allocation of future capital expenditures will depend upon a number of factors, including the number and size of acquisition opportunities, our cash flows from operating, investing, and financing activities, and our ability to assimilate acquisitions and execute our drilling program. In addition, the impact of oil, gas, and NGL prices on investment opportunities, the availability of capital, and the timing and results of our exploration and development activities may lead to changes in funding requirements for future development. We periodically review our capital expenditure budget to assess changes in current and projected cash flows, acquisition and divestiture activities, debt requirements, and other factors.

We may from time to time repurchase certain amounts of our outstanding debt securities for cash and/or through exchanges for other securities. Such repurchases or exchanges may be made in open market transactions, privately negotiated transactions, or otherwise. Any such repurchases or exchanges will depend on prevailing market conditions, our liquidity requirements, contractual restrictions, compliance with securities laws, and other factors. The amounts involved in any such transaction may be material. If market conditions are favorable, we may use a portion of 2017 divestiture proceeds to pay down debt; however, the amount and timing is uncertain. Please refer to *Note 5 - Long-Term Debt* in Part I, Item 1 of this report for additional discussion of previously repurchased Senior Notes.

As of the filing of this report, we could repurchase up to 3,072,184 shares of our common stock under our stock repurchase program, subject to the approval of our Board of Directors. Shares may be repurchased from time to time in the open market, or in privately negotiated transactions, subject to market conditions and other factors, including certain provisions of our Credit Agreement, the indentures governing our Senior Notes and Senior Convertible Notes, compliance with securities laws, and the terms and provisions of our stock repurchase program. Our Board of Directors periodically reviews this program as part of the allocation of our capital. We currently do not plan to repurchase any outstanding shares during 2017.

Analysis of Cash Flow Changes Between the Three Months Ended March 31, 2017, and 2016

The following tables present changes in cash flows between thethree months ended March 31, 2017, and 2016, for our operating, investing, and financing activities. The analysis following each table should be read in conjunction with our condensed consolidated statements of cash flows in Part I, Item 1 of this report.

Operating Activities

	 For the Three Months Ended March 31,			Amount Change		Percent Change	
	2017 20		2016	Between Periods		Between Periods	
			(in millions)				
Net cash provided by operating activities	\$ 135.0	\$	118.3	\$	16.7	14%	

Cash received from oil, gas, and NGL production revenues, net of transportation costs and production taxes, including derivative cash settlements, decreased\$21.1 million for the three months ended March 31, 2017, compared with the same period in 2016, as a result of the decline in production volumes, partially offset by a slight increase in our realized price, after the effects of derivative settlements. Interest paid increased \$18.4 million for the three months ended March 31, 2017, compared with the same period in 2016 due to the issuance of our 6.75% Senior Notes due 2026 and Senior Convertible Notes in 2016. These decreases in operating cash flow are offset by an \$11.7 million decrease in cash paid for LOE, an increase in working capital balances, and a decrease in cash G&A expense and exploration overhead for the three months ended March 31, 2017, compared to the same period in 2016. The decrease in LOE for the three months ended March 31, 2017, compared with the same period in 2016, is being driven by a decrease in production volumes and the sale of our higher cost Raven/Bear Den assets in 2016.

	For	For the Three Months Ended March 31,			Amount Change		Percent Change	
		2017		2016	Between Periods		Between Periods	
				(in millions)		_		
Net cash provided by (used in) investing activities	\$	517.3	\$	(189.3)	\$	706.6	373 %	

The increase in cash flow from investing activities for the three months ended March 31, 2017, compared with the same period in 2016 is largely due to divestiture cash proceeds of \$744.3 million received in the first quarter of 2017 primarily from the sale of our outside-operated Eagle Ford shale assets and a\$22.0 million decrease in capital expenditures, partially offset by a \$60.1 million increase in proved and unproved property acquisitions in the Midland Basin during the first quarter of 2017.

#### Financing activities

	For	For the Three Months Ended March 31,			Amount Change		Percent Change	
		2017		2016		ween Periods	Between Periods	
			(in m	illions)			_	
Net cash provided by (used in) financing activities	\$	(2.5)	\$	71.1	\$	(73.6)	(104)%	

We had a zero balance on our credit facility as of December 31, 2016, and March 31, 2017. Due to the proceeds received on the sale of our outside-operated Eagle Ford shale assets during the first quarter of 2017, we had a cash balance of \$659.1 million as of March 31, 2017. This compares to net borrowings of \$91.0 million during the three months ended March 31, 2016. Additionally, during the three months ended March 31, 2016, we paid \$19.9 million for the repurchase of a portion of our Senior Notes. Please refer to *Note 5 - Long-Term Debt* in Part I, Item 1 of this report for additional discussion.

#### Interest Rate Risk

We are exposed to market risk due to the floating interest rate on our revolving credit facility; however, as of March 31, 2017, and through the filing of this report, we had a zero balance on our credit facility. Our Credit Agreement allows us to fix the interest rate for all or a portion of the principal balance of our revolving credit facility for a period up to six months. Changes in interest rates do not impact the amount of interest we pay on our fixed-rate Senior Notes or fixed-rate Senior Convertible Notes, but can impact their fair market values. As of March 31, 2017, our outstanding fixed-rate debt totaled\$3.0 billion. Please refer to *Note 11 - Fair Value Measurements* in Part I, Item 1 of this report for additional discussion on the fair value of our Senior Notes and Senior Convertible Notes.

#### Commodity Price Risk

The prices we receive for our oil, gas, and NGL production directly impact our revenue, overall profitability, access to capital, and future rate of growth. Oil, gas, and NGL prices are subject to wide fluctuations in response to changes in supply and demand and other factors. The markets for oil, gas, and NGLs have been volatile, especially over the last several years, and these markets will likely continue to be volatile in the future. The prices we receive for our production depend on numerous factors beyond our control. Based on our production for the three months ended March 31, 2017, a 10 percent decrease in our average realized oil, gas, and NGL prices, before the effects of derivative settlements, would have reduced our oil, gas, and NGL production revenues by approximately \$16.8 million, \$10.1 million, and \$6.4 million, respectively. If commodity prices had been 10 percent lower, our derivative settlements would have been higher, partially offsetting the decrease in production revenues as discussed in the next paragraph.

We enter into commodity derivative contracts in order to reduce the impact of fluctuations in commodity prices. The fair value of our commodity derivative contracts are largely determined by estimates of the forward curves of the relevant price indices. For the three months ended March 31, 2017, a 10 percent decrease in the contract settlement prices would have increased our oil, gas, and NGL derivative settlement gain by approximately \$8.6 million, \$9.3 million, and \$4.5 million, respectively.

#### Off-Balance Sheet Arrangements

As part of our ongoing business, we have not participated in transactions that generate relationships with unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities ("SPEs"), which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes.

We evaluate our transactions to determine if any variable interest entities exist. If we determine that we are the primary beneficiary of a variable interest entity, that entity is consolidated into our consolidated financial statements. We have not been involved in any unconsolidated SPE transactions in 2017.

#### **Critical Accounting Policies and Estimates**

Please refer to the corresponding section in Part II, Item 7 and to Note 1 - Summary of Significant Accounting Policies included in Part II, Item 8 of our 2016 Form 10-K for discussion of our accounting policies and estimates.

#### **New Accounting Pronouncements**

Please refer to Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards under Part I, Item 1 of this report for new accounting matters.

#### **Non-GAAP Financial Measures**

Adjusted EBITDAX represents net income (loss) before interest expense, other non-operating income and expense, income taxes, depletion, depreciation, amortization and asset retirement obligation liability accretion expense, exploration expense, property impairments, non-cash stock-based compensation expense, derivative gains and losses net of settlements, change in the Net Profits Plan liability, gains and losses on divestitures, gains and losses on extinguishment of debt, and materials inventory impairments and losses on sale. Adjusted EBITDAX excludes certain items that we believe affect the comparability of operating results and can exclude items that are generally one-time in nature or whose timing and/or amount cannot be reasonably estimated. Adjusted EBITDAX is a non-GAAP measure that we present because we believe it provides useful additional information to investors and analysts, as a performance measure, for analysis of our ability to internally generate funds for exploration, development, acquisitions, and to service debt. We are also subject to financial covenants under our Credit Agreement based on adjusted EBITDAX ratios as further described in Note 5 - Long-Term Debt in Part I, Item 1 of this report. In addition, adjusted EBITDAX is widely used by professional research analysts and others in the valuation, comparison, and investment recommendations of companies in the oil and gas exploration and production industry, and many investors use the published research of industry research analysts in making investment decisions. Adjusted EBITDAX should not be considered in isolation or as a substitute for net income (loss), income (loss) from operations, net cash provided by operating activities, or profitability or liquidity measures prepared under GAAP. Because adjusted EBITDAX excludes some, but not all items that affect net income (loss) and may vary among companies, the adjusted EBITDAX amounts presented may not be comparable to similar metrics of other companies. Our credit facility provides a material source of liquidity for us. Under the terms of our Credit Agreement, if we fail to comply with the covenants that establish a maximum permitted ratio of senior secured debt to adjusted EBITDAX and a minimum permitted ratio of adjusted EBITDAX to interest, we will be in default, an event that would prevent us from borrowing under our credit facility and would therefore materially limit our sources of liquidity. In addition, if we default under our credit facility and are unable to obtain a waiver of that default from our lenders, lenders under that facility and under the indentures governing our outstanding Senior Notes and Senior Convertible Notes would be entitled to exercise all of their remedies for default.

The following table provides reconciliations of our net income (loss) and net cash provided by operating activities to adjusted EBITDAX for the periods presented:

	For	For the Three Months Ended March		
		2017		2016
		(in thou		,
Net income (loss) (GAAP)	\$	. , -	\$	(347,210
Interest expense		46,953		31,088
Other non-operating income, net		(335)		(6)
Income tax expense (benefit)		44,506		(194,875)
Depletion, depreciation, amortization, and asset retirement obligation liability accretion		137,812		214,207
Exploration (1)		10,570		13,611
Impairment of proved properties		_		269,785
Abandonment and impairment of unproved properties		_		2,311
Stock-based compensation expense		5,455		6,868
Net derivative gain		(114,774)		(14,228
Derivative settlement gain		7		147,028
Net (gain) loss on divestiture activity		(37,463)		69,021
(Gain) loss on extinguishment of debt		35		(15,722)
Other		4,986		432
Adjusted EBITDAX (Non-GAAP)		172,186		182,310
Interest expense		(46,953)		(31,088
Other non-operating income, net		335		6
Income tax (expense) benefit		(44,506)		194,875
Exploration (1)		(10,570)		(13,611
Amortization of discount and deferred financing costs		4,946		(920
Deferred income taxes		33,225		(195,039
Plugging and abandonment		(1,191)		(604
Other, net		(432)		(1,583
Changes in current assets and liabilities		27,926		(16,070
Net cash provided by operating activities (GAAP)	\$	134,966	\$	118,276

<sup>(1)</sup> Stock-based compensation expense is a component of exploration expense and general and administrative expense on the accompanying statements of operations. Therefore, the exploration line items shown in the reconciliation above will vary from the amount shown on the accompanying statements of operations for the component of stock-based compensation expense recorded to exploration expense.

#### **Cautionary Information about Forward-Looking Statements**

This report contains "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). All statements, other than statements of historical facts, included in this report that address activities, events, or developments with respect to our financial condition, results of operations, or economic performance that we expect, believe, or anticipate will or may occur in the future, or that address plans and objectives of management for future operations, are forward-looking statements. The words "anticipate," "assume," "believe," "budget," "estimate," "expect," "forecast," "intend," "plan," "project," "will," and similar expressions are intended to identify forward-looking statements. Forward-looking statements appear throughout this report, and include statements about such matters as:

- the amount and nature of future capital expenditures and the availability of liquidity and capital resources to fund capital expenditures;
- our outlook on future oil, gas, and NGL prices, well costs, and service costs;
- the drilling of wells and other exploration and development activities and plans, as well as possible or expected acquisitions or divestitures:
- the possible divestiture or farm-down of, or joint venture relating to, certain properties;
- proved reserve estimates and the estimates of both future net revenues and the present value of future net revenues associated with those proved reserve estimates;
- future oil, gas, and NGL production estimates;
- cash flows, anticipated liquidity, and the future repayment of debt;
- business strategies and other plans and objectives for future operations, including plans for expansion and growth of operations or to defer capital investment, and our outlook on our future financial condition or results of operations; and
- other similar matters such as those discussed in the Management's Discussion and Analysis of Financial Condition and Results of Operations section of this report.

Our forward-looking statements are based on assumptions and analyses made by us in light of our experience and our perception of historical trends, current conditions, expected future developments, and other factors that we believe are appropriate under the circumstances. These statements are subject to a number of known and unknown risks and uncertainties, which may cause our actual results and performance to be materially different from any future results or performance expressed or implied by the forward-looking statements. Some of these risks are described in the *Risk Factors* section in Part I, Item 1A of our 2016 Form 10-K, and include such factors as:

- the volatility of oil, gas, and NGL prices, and the effect it may have on our profitability, financial condition, cash flows, access to capital, and ability to grow
  production volumes and/or proved reserves;
- weakness in economic conditions and uncertainty in financial markets;
- our ability to replace reserves in order to sustain production;
- our ability to raise the substantial amount of capital required to develop and/or replace our reserves:
- our ability to compete against competitors that have greater financial, technical, and human resources:
- our ability to attract and retain key personnel;
- the imprecise estimations of our actual quantities and present value of proved oil, gas, and NGL reserves;
- the uncertainty in evaluating recoverable reserves and estimating expected benefits or liabilities;

- the possibility that exploration and development drilling may not result in commercially producible reserves:
- our limited control over activities on outside-operated properties;
- our reliance on the skill and expertise of third-party service providers on our operated properties;
- the possibility that title to properties in which we claim an interest may be defective;
- our planned drilling in existing or emerging resource plays using some of the latest available horizontal drilling and completion techniques is subject to drilling and completion risks and may not meet our expectations for reserves or production;
- the uncertainties associated with acquisitions, divestitures, joint ventures, farm-downs, farm-outs and similar transactions with respect to certain assets, including
  whether such transactions will be consummated or completed in the form or timing and for the value that we anticipate;
- the uncertainties associated with enhanced recovery methods:
- our commodity derivative contracts may result in financial losses or may limit the prices we receive for oil, gas, and NGL sales;
- the inability of one or more of our service providers, customers, or contractual counterparties to meet their
  obligations;
- our ability to deliver required quantities of crude oil, natural gas, natural gas liquids, or water to contractual counterparties;
- price declines or unsuccessful exploration efforts resulting in write-downs of our asset carrying values:
- the impact that depressed oil, gas, or NGL prices could have on our borrowing capacity under our Credit Agreement;
- the possibility our amount of debt may limit our ability to obtain financing for acquisitions, make us more vulnerable to adverse economic conditions, and make it
  more difficult for us to make payments on our debt;
- the possibility that covenants in our Credit Agreement or the indentures governing the Senior Notes and Senior Convertible Notes may limit our discretion in the operation of our business, prohibit us from engaging in beneficial transactions, or lead to the accelerated payment of our debt;
- operating and environmental risks and hazards that could result in substantial losses:
- the impact of seasonal weather conditions and lease stipulations on our ability to conduct drilling activities:
- our ability to acquire adequate supplies of water and dispose of or recycle water we use at a reasonable cost in accordance with environmental and other applicable rules:
- complex laws and regulations, including environmental regulations, that result in substantial costs and other risks:
- the availability and capacity of gathering, transportation, processing, and/or refining facilities:
- our ability to sell and/or receive market prices for our oil, gas, and NGLs:
- new technologies may cause our current exploration and drilling methods to become obsolete:
- the possibility of security threats, including terrorist attacks and cybersecurity breaches, against, or otherwise impacting, our facilities and systems;
- litigation, environmental matters, the potential impact of legislation and government regulations, and the use of management estimates regarding such matters.

We caution you that forward-looking statements are not guarantees of future performance and actual results or performance may be materially different from those expressed or implied in the forward-looking statements. The forward-looking statements in this report speak as of the filing of this report. Although we may from time to time voluntarily update our prior forward-looking statements, we disclaim any commitment to do so except as required by securities laws.

#### ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The information required by this item is provided under Interest Rate Risk and Commodity Price Risk in Item 2 above and is incorporated herein by reference. Please also refer to the information under Interest Rate Risk and Commodity Price Risk in Management's Discussion and Analysis of Financial Condition and Results of Operations in Part II. Item 7 of our 2016 Form 10-K.

#### ITEM 4. CONTROLS AND PROCEDURES

#### **Evaluation of Disclosure Controls and Procedures**

We maintain a system of disclosure controls and procedures that is designed to reasonably ensure that information required to be disclosed in our SEC reports is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and to reasonably ensure that such information is accumulated and communicated to our management, including the Chief Executive Officer and the Chief Financial Officer, as appropriate, to allow for timely decisions regarding required disclosure.

Our management, including our Chief Executive Officer and Chief Financial Officer, does not expect that our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) ("Disclosure Controls") will prevent all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within our company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the control. The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected. We monitor our Disclosure Controls and make modifications as necessary; our intent in this regard is that the Disclosure Controls will be modified as systems change and conditions warrant.

An evaluation of the effectiveness of the design and operation of our Disclosure Controls was performed as of the end of the period covered by this report. This evaluation was performed under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our Disclosure Controls were effective at a reasonable assurance level.

#### **Changes in Internal Control Over Financial Reporting**

There were no changes during the first quarter of 2017 that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### PART II. OTHER INFORMATION

#### ITEM 1. LEGAL PROCEEDINGS

From time to time, we may be involved in litigation relating to claims arising out of our business and operations in the normal course of business. As of the filing of this report, no legal proceedings are pending against us that we believe individually or collectively could have a materially adverse effect upon our financial condition, results of operations or cash flows.

#### ITEM 1A. RISK FACTORS

There have been no material changes to the risk factors as previously disclosed in our 2016 Form 10-K.

#### ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

The following table provides information about purchases by us or any "affiliated purchaser" (as defined in Rule 10b-18(a)(3) under the Exchange Act) during the quarter ended March 31, 2017, of shares of our common stock, which is the sole class of equity securities registered by us pursuant to Section 12 of the Exchange Act:

## PURCHASES OF EQUITY SECURITIES BY ISSUER AND AFFILIATED PURCHASERS

	(a)	(b)	(c)	(d)		
Period	Total Number of Shares Purchased <sup>(1)</sup>	Weighted Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Program	Maximum Number of Shares that May Yet Be Purchased Under the Program <sup>(2)</sup>		
01/01/17 - 01/31/17	108	\$ 34.40	_	3,072,184		
02/01/17 - 02/28/17	66	\$ 26.99		3,072,184		
03/01/17 - 03/31/17	205	\$ 24.39		3,072,184		
Total:	379	\$ 27.69	_	3,072,184		

<sup>(1)</sup> All shares purchased by us in the first quarter of 2017 were to offset tax withholding obligations that occurred upon the delivery of outstanding shares underlying restricted stock units delivered under the terms of grants under our Equity Incentive Compensation Plan.

Our payment of cash dividends to our stockholders is subject to covenants under the terms of our Credit Agreement that limit our annual dividend payments to no more than \$50.0 million per year. We are also subject to certain covenants under the indentures governing our Senior Notes and Senior Convertible Notes that restrict certain payments, including dividends; provided, however, that the first \$6.5 million of dividends paid each year are not restricted by these covenants. We do not anticipate that these restrictions will limit our payment of dividends at our current rate for the foreseeable future if any dividends are declared by our Board of Directors.

In July 2006, our Board of Directors approved an increase in the number of shares that may be repurchased under the original August 1998 authorization to up to 6,000,000 shares as of the effective date of the resolution. Accordingly, as of the filing of this report, we may repurchase up to 3,072,184 shares of common stock on a prospective basis, subject to the approval of our Board of Directors. The shares may be repurchased from time to time in open market transactions or privately negotiated transactions, subject to market conditions and other factors, including certain provisions of our Credit Agreement, the indentures governing our Senior Notes and Senior Convertible Notes, and compliance with securities laws. Stock repurchases may be funded with existing cash balances, internal cash flow, or borrowings under our Credit Agreement. The stock repurchase program may be suspended or discontinued at any time.

#### ITEM 6. EXHIBITS

The following exhibits are filed or furnished with or incorporated by reference into this report:

Exhibit Number	<u>Description</u>
2.1*	Membership Interest Purchase Agreement dated January 1, 2017 between SM Energy Company and Venado EF LLC
2.2*	Second Amendment to Membership Interest Purchase Agreement dated March 4, 2017 between SM Energy Company and Venado EF L.P.
3.1	Restated Certificate of Incorporation of SM Energy Company, as amended through June 1, 2010 (filed as Exhibit 3.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010, and incorporated herein by reference)
3.2	Amended and Restated By-Laws of SM Energy Company, effective as of February 21, 2017 (filed as Exhibit 3.2 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2016, and incorporated herein by reference)
10.1	Ninth Amendment to Fifth Amended and Restated Credit Agreement, dated March 31, 2017, among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on April 3, 2017 and incorporated herein by reference)
12.1*	Computation of Ratio of Earnings to Fixed Charges
31.1*	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes - Oxley Act of 2002
31.2*	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes - Oxley Act of 2002
32.1**	Certification pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes - Oxley Act of 2002
101.INS*	XBRL Instance Document
101.SCH*	XBRL Schema Document
101.CAL*	XBRL Calculation Linkbase Document
101.LAB*	XBRL Label Linkbase Document
101.PRE*	XBRL Presentation Linkbase Document
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document
*	Filed with this report.
**	Furnished with this report.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

#### SM ENERGY COMPANY

May 3, 2017 By: /s/ JAVAN D. OTTOSON

Javan D. Ottoson

President and Chief Executive Officer (Principal Executive Officer)

By: /s/ A. WADE PURSELL

A. Wade Pursell

Executive Vice President and Chief Financial Officer

(Principal Financial Officer)

May 3, 2017 By: /s/ MARK T. SOLOMON

May 3, 2017

Mark T. Solomon

Vice President - Controller and Assistant Secretary

(Principal Accounting Officer)

# MEMBERSHIP INTEREST PURCHASE AGREEMENT BETWEEN

**SM ENERGY COMPANY** 

as Seller

and

VENADO EF LLC

as Buyer

**DATED JANUARY 1, 2017** 

### ARTICLE I DEFINITIONS; INTERPRETATION 1 1.1 Definitions. 1 1.2 Interpretation. 1 ARTICLE II PURCHASE AND SALE OF INTERESTS 2 2.1 Purchase and Sale. 2 2.2 Purchase Price. 2 **2.3 Deposit**. 2 2.4 Adjustments to Purchase Price. 2 2.5 Allocated Value. 5 2.6 Settlement; Disputes. 5 ARTICLE III REPRESENTATIONS AND WARRANTIES OF SM ENERGY 7 3.1 Organization, Existence. 7 3.2 Authorization. 7 3.3 No Conflicts. 8 3.4 Consents and Approvals. 8 3.5 Bankruptcy. 9 3.6 Foreign Person. 9 3.7 The Interests. 9 3.8 Capital Structure of the Company 9 3.9 Subsidiaries; Assets. 10 **3.10** Litigation. 10 **3.11 Liabilities**. 10 3.12 Material Contracts. 10 3.13 No Violation of Laws. 12 **3.14** Royalties, Etc. 12 **3.15** *Imbalances*. 12 3.16 Current Commitments. 12 3.17 Environmental. 13 3.18 Taxes. 13 3.19 Brokers' Fees. 14 3.20 Advance Payments. 14 3.21 Partnerships. 14 3.22 Hedge Contracts 14 **3.23** *Insurance* 14 3.24 Equipment and Personal Property 14 3.25 Wells; Plug and Abandon Notice 15 3.26 Permits 15 **3.27** *Payouts* 15

3.28 Bonds 15

3.29 Preferential Purchase Rights 15

3.30 Availability of Properties. 15
3.31 Employees. 15
3.32 Employee Benefits. 16
3.33 Affiliate Transactions 16
3.34 Unlawful Payments 16
3.35 Contribution. 16
3.36 Springfield Financial Information. 17
3.37 Springfield System Ownership Interests. 17
3.38 No Other Representations or Warranties; Disclosed Materials. 17
3.30 1.00 the Representations of Warrantees, Disclosed Patientals.
ARTICLE IV REPRESENTATIONS AND WARRANTIES OF BUYER 18
4.1 Organization; Existence. 18
4.2 Authorization. 18
4.3 No Conflicts. 18
4.4 Bankruptcy. 19
<b>4.5 Litigation</b> . 19
<b>4.6 Financing</b> . 19
4.7 Independent Evaluation. 19
4.8 Brokers' Fees. 19
4.9 Accredited Investor. 19
ARTICLE V ACCESS / DISCLAIMERS 20
<b>5.1</b> Access. 20
<b>5.2</b> Confidentiality. 22
5.3 Disclaimers. 22
ARTICLE VI TITLE MATTERS; CASUALTIES 24
6.1 The Company's Title. 24
6.2 Notice of Title Defects; Defect Adjustments. 24
· · · · · · · · · · · · · · · · · · ·
6.3 Casualty or Condemnation Loss. 29
6.4 Preferential Rights and Consents to Assign. 30
ARTICLE VII ENVIRONMENTAL MATTERS 32
7.1 Environmental Defects. 32
<b>7.2</b> NORM. 35
ARTICLE VIII CERTAIN AGREEMENTS 35
8.1 Conduct of Business. 35
8.2 Governmental Bonds. 39
8.3 HSR Act 39
8.4 Financial Statements and Cooperation 39
<b>8.5</b> Names. 40
8.6 Hedging Matters 40
8.7 Financing Cooperation. 42
8.8 Equity Commitment 43
ADTICLE IV CONDITIONS TO CLOSING 42
ARTICLE IX CONDITIONS TO CLOSING 43

	Buyer's Conditions to Closing. 43 SM Energy's Conditions to Closing. 44
	TAX MATTERS 45
	Tax Matters. 45
	Transfer Taxes. 47
	Tax Cooperation. 47
ARTICLE X	I CLOSING 48
11.1	Date of Closing. 48
11.2	Place of Closing. 48
11.3	Closing Obligations. 48
11.4	Records. 49
ARTICLE X	II ACQUISITION TERMINATION AND REMEDIES 49
	Right of Termination. 49
	Effect of Termination. 50
12.3	Return of Documentation and Confidentiality. 52
ARTICLE X	III COMPANY LIABILITY; SURVIVAL; INDEMNIFICATION 52
13.1	Company Liability. 52
	Indemnities of SM Energy. 52
13.3	Indemnities of Buyer. 54
	Limitation on Liability. 54
	Express Negligence. 55
13.6	Exclusive Remedy for Agreement. 55
	Indemnification Procedures. 55
	Survival. 57
13.9	Non-Compensatory Damages. 58
ARTICLE X	IV MISCELLANEOUS 58
14.1	Counterparts. 58
	Notices. 58
	Expenses. 60
	Waivers; Rights Cumulative. 60
	Relationship of the Parties. 60
	Entire Agreement; Conflicts; No Third Party Beneficiaries. 60
	Governing Law; Venue; Jury Waiver. 61
	Further Cooperation 62
14.9	Amendment. 62
	Parties in Interest. 62
	Successors and Permitted Assigns. 62
	Publicity. 62
	Preparation of Agreement. 63
	Conduct of the Parties. 63
14.15	Severability. 63

14.17 Waiver of Claims Against Debt Financing Sources. 64

#### **Schedules**

Schedule 2.5	Allocated Values
Schedule 3.3	No Conflicts
Schedule 3.7	The Interests
Schedule 3.10	Litigation
Schedule 3.11	Liabilities
Schedule 3.12(a)	Material Contracts
Schedule 3.12(b)	Material Contract Matters
Schedule 3.13	Violation of Laws
Schedule 3.14	Suspense Funds
Schedule 3.15	Imbalances
Schedule 3.16	AFEs
Schedule 3.17	Environmental Matters
Schedule 3.18	Taxes
Schedule 3.21	Partnerships
Schedule 3.23	Insurance
Schedule 3.24	<b>Equipment and Personal Property</b>
Schedule 3.25	P&A Obligations
Schedule 3.27	Payouts
Schedule 3.28	Bonds
Schedule 3.35	Contribution Documents
Schedule 6.4	Consents and Preferential Rights
Schedule 8.1	Permitted Operations
Schedule 8.6(a)	Hedges
Schedule 9.1(f)	Required Consents

<b>Exhibits</b>	
Exhibit A	Leases
Exhibit B	Wells and Well Locations
Exhibit C	Form of Assignment
Exhibit D	Target Intervals
Exhibit E	Seismic Data
Exhibit F	Excluded Assets
Exhibit G	Form of Seismic License Agreement
Exhibit H	Form of Mutual Release
Exhibit I	Springfield Gas and Oil Gathering Systems
Exhibit J	Form of Transition Services Agreement
Exhibit K	Form of Consent Notice

**THIS MEMBERSHIP INTEREST PURCHASE AGREEMENT** (this "Agreement") is made as of January 1, 2017 (the "Execution Date") between SM ENERGY COMPANY, a Delaware corporation ("SM Energy") and VENADO EF LLC, a Delaware limited liability company ("Buyer"). SM Energy and Buyer shall sometimes be referred to herein together as the "Parties", and each individually as a "Party".

#### **RECITALS**

WHEREAS, effective as of the Effective Time (as defined below), SM Energy desires to sell and convey, and Buyer desires to purchase and pay for, all of the issued and outstanding membership interests of SM Energy ST LLC, a Delaware limited liability company (the "Company," and such membership interests, the "Interests"); and

WHEREAS, the Parties have reached agreement regarding the sale and purchase of the Interests.

NOW THEREFORE, in consideration of the mutual agreements herein contained, the Parties hereby agree as follows:

## ARTICLE I DEFINITIONS; INTERPRETATION

- 1.1 Definitions. Capitalized terms used in this Agreement shall have the meanings given such terms in Annex I.
- 1.2 Interpretation. All references in this Agreement to Exhibits, Schedules, Annexes, Articles, Sections, subsections and other subdivisions refer to the corresponding Exhibits, Schedules, Annexes, Articles, Sections, subsections and other subdivisions of or to this Agreement unless expressly provided otherwise. Titles appearing at the beginning of any Articles, Sections, subsections and other subdivisions of this Agreement are for convenience only, do not constitute any part of this Agreement, and shall be disregarded in construing the language hereof. The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular Article, Section, subsection or other subdivision unless expressly so limited. The words "this Article," "this Section" and "this subsection," and words of similar import, refer only to Article, Section or subsection hereof in which such words occur. The word "including" (in its various forms) means "including without limitation." All references to "\$" or "dollars" shall be deemed references to United States Dollars. Each accounting term not defined herein will have the meaning given to it under GAAP, as in effect on the Execution Date. The word "or" is not exclusive and shall have the same meaning as "and/or" unless the context requires otherwise. Pronouns in masculine, feminine or neuter genders shall be construed to state and include any other gender, and words, terms and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires. Appendices, Annexes and Exhibits referred to herein are attached to and made a part of this Agreement. Unless expressly stated otherwise, references to any Law or contract shall mean such Law or contract as it may be amended from time to time. References to any date or time shall mean such date or time, as applicable, at the location of the Assets and for purposes of

calculating the time period in which any notice or action is to be given or undertaken hereunder, such period shall be deemed to begin at 12:01 a.m. on the applicable date at the location of the Assets.

#### ARTICLE II PURCHASE AND SALE OF INTERESTS

- **2.1** Purchase and Sale. Subject to the terms and conditions of this Agreement, Buyer agrees to purchase from SM Energy, and SM Energy agrees to sell, assign and deliver to Buyer the Interests for the consideration specified in this ARTICLE II.
- **2.2 Purchase Price**. Subject to the other terms and conditions of this Agreement, the purchase price for the Interests shall be Eight Hundred Million Dollars (\$800,000,000.00) (the "**Purchase Price**"). The Purchase Price shall be payable as follows:
- (a) Within three (3) Business Days of the Execution Date, Buyer shall deposit by wire transfer in immediately available funds with the Escrow Agent, to be held, invested and disbursed in accordance with the terms of this Agreement and the Escrow Agreement, the sum of One Hundred Twenty Million Dollars (\$120,000,000.00) (such amount, together with any interest earned thereon. the "*Deposit*").
  - (b) At Closing, Buyer shall pay SM Energy the Closing Amount by wire transfer in immediately available funds.
- 2.3 Deposit. The Deposit shall be held in an escrow account that will be administered in accordance with the Escrow Agreement. At and in connection with the Closing, SM Energy and Buyer shall execute and deliver joint written instructions to the Escrow Agent authorizing the Escrow Agent to distribute the Deposit to SM Energy. If this Agreement is terminated, SM Energy and Buyer shall execute and deliver joint written instructions to the Escrow Agent to distribute the Deposit in accordance with Section 12.2. If Closing occurs, the Deposit shall be applied toward the Purchase Price.

#### 2.4 Adjustments to Purchase Price.

(a) For purposes of determining the amounts of the adjustments to the Purchase Price provided for in this Section 2.4, the principles set forth in this Section 2.4(a) shall apply. The Company, on behalf of Buyer, shall be entitled to all production of Hydrocarbons from or attributable to the Company's Units, Leases, and Wells at and after the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Assets at or after the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Operating Expenses incurred at and after the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Assets prior to the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Operating Expenses incurred prior to the Effective Time. "Earned" and "incurred", as used in the Agreement shall be interpreted

in accordance with GAAP and Council of Petroleum Accountants Society standards, except as otherwise specified herein. For purposes of allocating production (and proceeds and accounts receivable with respect thereto), under this Section 2.4, (i) liquid Hydrocarbons shall be deemed to be "from or attributable to" the Units, Leases and Wells when they pass through the pipeline connecting into the storage facilities into which they are run and (ii) gaseous Hydrocarbons shall be deemed to be "from or attributable to" the Units, Leases and Wells when they pass through the royalty measurement meters, delivery point sales meters or custody transfer meters on the gathering lines or pipelines through which they are transported (whichever meter is closest to the well). SM Energy shall utilize reasonable interpolative procedures, consistent with industry practice, to arrive at an allocation of production when exact meter readings or gauging and strapping data are not available. As part of the Preliminary Settlement Statement, Buyer shall provide to SM Energy such data as is reasonably necessary to support any estimated allocation, for purposes of establishing the Closing Amount.

- (b) The Purchase Price shall be adjusted upward by the following amounts (without duplication):
- (i) an amount equal to the value of all Hydrocarbons produced from or attributable to the Assets in storage or existing in stock tanks, pipelines or plants (including inventory) as of the Effective Time, the value to be based upon the contract price in effect as of the Effective Time (or if there is no contract price, then the market price in effect as of the Effective Time in the field in which such Hydrocarbons were produced or if actually sold prior to the date of determination, the proceeds actually recovered by SM Energy or the Company attributable to such sale), net of (A) amounts payable as royalties, overriding royalties and other burdens upon, measured by or payable out of such production (that are deducted by the Third Party purchaser of such production), (B) marketing fees, (C) gravity adjustments for which there is no payment in connection with the sale of such Hydrocarbons, and (D) Asset Taxes;
- (ii) an amount equal to all Operating Expenses (other than Taxes) paid by SM Energy that are attributable to the Assets during the Interim Period, whether paid before or after the Effective Time;
- (iii) an amount equal to all pre-Effective Time revenue received and retained (without distribution to SM Energy) by the Company and all post-Effective Time costs paid by SM Energy that are credited or charged under the COO Agreement with respect to ownership of the Springfield Gathering System;
  - (iv) Subject Transfer Taxes paid by SM Energy with respect to the transactions contemplated by this Agreement;
  - (v) Asset Taxes allocated to Buyer pursuant to Section 10.1, but paid or economically borne by SM Energy; and
  - (vi) any other amount provided for elsewhere in this Agreement or otherwise agreed upon by the Parties.

- (c) The Purchase Price shall be adjusted downward by the following amounts (without duplication):
- (i) an amount equal to all proceeds actually received by SM Energy, the Company or its or their Affiliates attributable to the sale of Hydrocarbons (d) produced from or attributable to the Assets during the Interim Period or (e) contained in storage or existing in stock tanks, pipelines or plants (including inventory) as of the Effective Time for which an upward Purchase Price adjustment was made pursuant to *Section 2.4(b)*;
  - (i) any reduction to the Purchase Price pursuant to ARTICLE VI;
  - (ii) any reduction to the Purchase Price pursuant to ARTICLE VII;
  - (iii) the value of any Casualty Loss pursuant to Section 6.3(b);
- (iv) an amount equal to all Operating Expenses paid by Buyer or the Company that are attributable to the Assets prior to the Effective Time, whether paid before or after the Effective Time;
- (v) an amount equal to all post-Effective Time revenue received by SM Energy and pre-Effective Time costs paid by the Company, in each case that are charged or credited under the COO Agreement with respect to ownership of the Springfield Gathering System;
- (vi) an amount equal to all proceeds from sales of Hydrocarbons relating to the Assets and payable to owners of Working Interests, royalties, overriding royalties and other similar interests (in each case) that are held by SM Energy in suspense as of the Closing Date;
  - (vii) Asset Taxes allocated to SM Energy pursuant to Section 10.1, but paid or economically borne by Buyer; and
  - (viii) any other amount provided for elsewhere in this Agreement or otherwise agreed upon by the Parties.
- (f) If, prior to determination of the Final Settlement Statement, Buyer or SM Energy discovers an error in the Imbalances set forth in *Schedule 3.13*, then the Purchase Price shall be adjusted (based on a value of \$1.50 per Mcf), and *Schedule 3.13* will be deemed amended to reflect the Imbalances for which the Purchase Price is so adjusted.
- (g) Should Buyer or the Company (after the transfer of the Interests to Buyer hereunder) receive, at any time, any proceeds or other income to which SM Energy is entitled under *Section 2.4(a)*, Buyer shall (or shall cause the Company to) promptly remit the same to SM Energy. If SM Energy receives any proceeds or other income to which SM Energy is not entitled under *Section 2.4(a)* and which has not been accounted for under any other adjustment in this *Section 2.4* in connection with the Final Settlement Statement under *Section 2.6(b)*, SM Energy shall fully disclose, account for and promptly remit same to the Company or its successor.

- (h) Should Buyer or the Company (after the transfer of the Interests to Buyer hereunder), pay, at any time, any Operating Expenses for which SM Energy is responsible under *Section 2.4(a)*, SM Energy shall reimburse the Company or its successor, on behalf of the Company, promptly after receipt of such Person's invoice, accompanied by copies of the relevant vendor or other invoice and proof of payment. Should SM Energy pay after the Effective Time any Operating Expenses for which SM Energy is not responsible under *Section 2.4(a)* and which is not already accounted for under any other adjustment in this *Section 2.4*, the Company or its successor shall reimburse SM Energy promptly after receipt of such Person's invoices, accompanied by copies of the relevant vendor or other invoice and proof of payment.
- (i) Notwithstanding the foregoing, *Sections 2.4(e)* and *2.4(f)* shall not apply after the date that is five hundred forty (540) days from the Closing Date.
- 2.5 Allocated Value. SM Energy and Buyer have agreed upon an allocation of the unadjusted Purchase Price among each of the Wells and Well Locations, and such allocations are set forth on Schedule 2.5 (the "Allocated Value"). SM Energy and Buyer agree that the Allocated Values shall be used in calculating adjustments to the Purchase Price as provided herein. Any adjustments to the Purchase Price, other than the adjustments provided for in Section 2.4, shall be applied on a pro rata basis to the amounts set forth in Schedule 2.5, as applicable, for all Assets. After all such adjustments are made, any adjustments to the Purchase Price made pursuant to Section 2.4 shall be applied to the amounts set forth on Schedule 2.5, as applicable, for the particular affected Assets.
- (a) For tax purposes, the Parties agree to report the transactions contemplated by this Agreement in a manner consistent with the terms of this Agreement. The Parties agree that the Purchase Price and any liabilities properly treated as consideration for U.S. federal income Tax purposes shall be allocated among the Assets for U.S. federal and applicable state income Tax purposes in accordance with Section 1060 of the Code and to use commercially reasonable efforts to jointly prepare Form 8594 (Asset Acquisition Statement under Section 1060). Buyer shall prepare a draft of the Form 8594 and provide such draft to SM Energy at least sixty (60) days prior to the Form 8594 filing due date so that SM Energy may review and comment prior to the Parties' filing of the Form 8594. SM Energy shall deliver any written comments to Buyer within fifteen (15) days of receipt of the Form 8594 from Buyer. Buyer and SM Energy shall work in good faith to resolve any disagreements to the Form 8594. In the event that an agreement is reached on the allocation, neither Party will take any position inconsistent therewith, including in any tax return, refund claim, litigation, arbitration, or otherwise.

#### 2.6 Settlement; Disputes.

(a) Not less than five (5) Business Days prior to the Closing, SM Energy shall prepare and submit to Buyer for review a draft settlement statement using the best information available to SM Energy (the "*Preliminary Settlement Statement*") that shall set forth the adjusted Purchase Price reflecting each adjustment made in good faith in accordance with this Agreement as of the date of preparation of such Preliminary Settlement Statement and the itemized calculation and reasonable supporting documentation of the adjustments used to determine such amount. Within three (3) Business Days of receipt of the Preliminary Settlement Statement, Buyer has the right,

but not the obligation, to deliver to SM Energy a written report containing all changes with the explanation therefor that Buyer proposes to be made to the Preliminary Settlement Statement. The Preliminary Settlement Statement, as agreed upon by the Parties, will be used to adjust the Purchase Price at Closing (such adjusted price, the "*Preliminary Purchase Price*"). If the Parties cannot agree on the Preliminary Settlement Statement prior to the Closing, the Preliminary Settlement Statement as presented by SM Energy will be used to adjust the Purchase Price at Closing.

- (b) On or before the later of (i) one hundred twenty (120) days after the Closing and (ii) final resolution of all Title Defects, Title Benefits and Environmental Defects, a final settlement statement (the "Final Settlement Statement") will be prepared by SM Energy. based on actual income and expenses during the Interim Period and which takes into account all final adjustments made to the Purchase Price and shows the resulting final adjusted Purchase Price (the "Final Purchase Price"). SM Energy shall, at Buyer's request, supply reasonable documentation in its or its Affiliates' possession available to support the actual revenue, expenses and other items for which adjustments are made. The Final Settlement Statement shall set forth the actual proration of the amounts required by this Agreement and reasonable supporting documentation of the adjustments used to determine such amounts. As soon as practicable, and in any event within forty-five (45) days, after receipt of the Final Settlement Statement, Buyer has the right, but not the obligation, to return a written report containing any proposed changes to the Final Settlement Statement and an explanation of any such changes and the reasons therefor (the "Dispute Notice"). If the Final Purchase Price set forth in the Final Settlement Statement is mutually agreed upon by SM Energy and Buyer, the Final Settlement Statement and the Final Purchase Price shall be final and binding on the Parties (except with respect to Asset Taxes). Any difference in the adjusted Purchase Price as paid at Closing pursuant to the Preliminary Settlement Statement and the Final Purchase Price shall be paid by the owing Party on or before the date that is ten (10) days following agreement or deemed agreement pursuant to this Section 2.6(b) (or determination by the Accounting Arbitrator pursuant to Section 2.6(c), as applicable) to the owed Party. All amounts paid or transferred pursuant to this Section 2.6(b) shall be delivered in United States currency by wire transfer of immediately available funds to the account specified in writing by the relevant Party.
- (c) If the Parties are unable to resolve the matters addressed in the Dispute Notice, each of Buyer and SM Energy shall, within fourteen (14) Business Days following the delivery of such Dispute Notice, summarize its position with regard to such dispute in a written document and submit such summaries to the Dallas, Texas office of KPMG LLP or such other Person as the Parties may mutually select (the "Accounting Arbitrator"), together with the Dispute Notice, the Final Settlement Statement and any other documentation such Party may desire to submit. The Accounting Arbitrator shall conduct the arbitration proceedings in accordance with the Commercial Arbitration Rules of the AAA, to the extent such rules do not conflict with the terms of this Section 2.6(c). The Accounting Arbitrator, once appointed, shall have no ex parte communications with the Parties concerning the expert determination or the underlying dispute. All communications between any Party and the Accounting Arbitrator shall be conducted in writing, with copies sent simultaneously to the other Party in the same manner, or at a meeting to which all Parties have been invited and of which such Parties have been provided at least five (5) Business Days' notice. Within twenty (20) Business Days after receiving the Parties' respective submissions, the Accounting Arbitrator shall render a decision choosing either SM Energy's position or Buyer's position with respect to each

matter addressed in any Dispute Notice, based on the materials described above. Any decision rendered by the Accounting Arbitrator pursuant hereto shall be final, conclusive and binding on the Parties and will be enforceable against any of the Parties in any court of competent jurisdiction (except with respect to Asset Taxes). The Accounting Arbitrator shall act as an expert for the limited purpose of determining the specific disputed aspects of the adjusted Purchase Price adjustments submitted by each Party and may not award damages, interest or penalties to any Party with respect to any matter. SM Energy and Buyer shall each bear their own legal fees and other costs of presenting its case. The costs and expenses of such Accounting Arbitrator shall be borne one-half by Buyer and one-half by SM Energy. In the event that KPMG declines to serve as the Accounting Arbitrator and the Parties are unable to mutually agree on another Accounting Arbitrator, then the Dallas, Texas office of the AAA shall appoint the Accounting Arbitrator under such conditions as the AAA in its sole discretion deems necessary or advisable.

## ARTICLE III REPRESENTATIONS AND WARRANTIES OF SM ENERGY

SM Energy represents and warrants to Buyer as of the Execution Date and the Closing Date (except as expressly set forth in this ARTICLE III) as follows:

#### 3.1 Organization, Existence.

- (a) SM Energy is a corporation duly formed and validly existing under the Laws of the State of Delaware. SM Energy has all requisite power and authority to own and operate its property (including without limitation, the Interests) and to carry on its business as now conducted. SM Energy is duly licensed or qualified to do business as a foreign corporation and is in good standing in all jurisdictions in which such qualification is required by Law, except where the failure to qualify or be in good standing would not have a Material Adverse Effect.
- (b) The Company is a limited liability company duly organized, validly existing and in good standing under the Laws of Delaware. The Company has full limited liability company power and authority to own, lease or otherwise hold the Assets and to conduct its business in the manner presently conducted and is duly qualified to conduct business in the jurisdictions in which such qualification is required by Law, except where failure to qualify would not have a Material Adverse Effect. Prior to the Execution Date, copies of the organizational documents of the Company (the "Company Organizational Documents") have been delivered to Buyer and such copies are accurate and complete as of the Execution Date, in full force and effect and not modified, amended or rescinded. The Company is not in default under, or in violation of, any of the Company Organizational Documents.
- 3.2 Authorization. SM Energy has full power and authority to enter into and perform this Agreement, the Transaction Documents and the transactions contemplated herein and therein. The execution, delivery and performance by SM Energy of this Agreement and the Transaction Documents has been, and the execution, delivery and performance by SM Energy of all other documents delivered by it pursuant to this Agreement will be when delivered, duly and validly authorized and approved by all necessary corporate action on the part of SM Energy, and no other proceedings on the part of SM Energy are necessary to authorize the execution, delivery or

performance thereof or to consummate the transactions contemplated hereby or thereby. Assuming the due authorization, execution and delivery by the other parties to such documents, this Agreement and the Transaction Documents constitute, and the other documents delivered pursuant to this Agreement to which SM Energy is a party will constitute when delivered, SM Energy's legal, valid and binding obligations, enforceable against SM Energy in accordance with their respective terms, subject however to the effects of bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar Laws relating to or affecting creditors' rights, as well as to principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).

- (a) The Company has full power and authority to enter into and perform the transactions contemplated herein and in the other Transaction Documents to be entered into and performed by it. The execution, delivery and performance by the Company of the documents to be delivered by it pursuant to this Agreement and the Transaction Documents will be when delivered, duly and validly authorized and approved by all necessary corporate action on the part of the Company, and no other proceedings on the part of the Company are necessary to authorize the execution, delivery or performance thereof or to consummate the transactions contemplated hereby or thereby. Assuming the due authorization, execution and delivery by the other parties to such documents, the documents delivered pursuant to this Agreement and the Transaction Documents to which the Company is a party will constitute when delivered, the Company's legal, valid and binding obligations, enforceable against the Company in accordance with their respective terms, subject however to the effects of bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar Laws relating to or affecting creditors' rights, as well as to principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).
- 3.3 No Conflicts. Assuming the receipt of all Consents and the waiver of all Preferential Rights applicable to the transaction (each of which is set forth on Schedule 3.3), the execution, delivery and performance by SM Energy of this Agreement, the Transaction Documents and the consummation of the transactions contemplated hereby and thereby does not and will not (a) conflict with or result in a breach of any provisions of (i) the Existing SM Energy Credit Agreement and (ii) the organizational documents or other governing documents of SM Energy or the Company, (b) except for Permitted Encumbrances, result in a default or the creation of any Encumbrance or give rise to any right of termination, cancellation or acceleration under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license or other material agreement (other than the Leases or Applicable Contracts) to which SM Energy or the Company is a party or by which SM Energy, the Company or the Assets may be bound or (c) violate any Law applicable to SM Energy, the Company or any of the Assets, except in the case of clauses (b) and (c) where such default, Encumbrance, termination, cancellation, acceleration or violation would not have a material adverse effect upon the ability of SM Energy to consummate the transactions contemplated by this Agreement.
- 3.4 Consents and Approvals. Except for compliance with the HSR Act and the consents referred to in Schedule 6.4, there are no consents or other restrictions on assignment, including requirements for consents from Third Parties to any assignment, in each case, that would be applicable in connection with the consummation by Buyer of the transactions contemplated by this Agreement, including the Contribution. Additionally, except for compliance with the HSR Act and

as set forth on *Schedule 6.4*, no other consent, waiver, approval, order, action, permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Authority is required on the part of SM Energy or the Company in connection with the execution, delivery and performance of this Agreement and the other documents to be delivered by it pursuant to this Agreement, the Transaction Documents or the consummation of the transactions contemplated hereby or thereby, including the Contribution.

- 3.5 Bankruptcy. There are no bankruptcy, reorganization or receivership proceedings pending, being contemplated by or, to SM Energy's Knowledge, threatened in writing against SM Energy or the Company, or any of its or their Affiliates.
- **3.6** Foreign Person. SM Energy is not a "disregarded entity" or a "foreign person" within the meaning of Section 1445 of the Code and its implementing Treasury Regulations.
- 3.7 The Interests. Except as disclosed on Schedule 3.7, SM Energy has legal, beneficial and record title to the Interests, free and clear of any Encumbrances, and there are no other contracts or commitments that could require SM Energy to sell, transfer or otherwise dispose of the Interests, other than this Agreement. SM Energy has the power, authority and legal capacity to sell, transfer, assign and deliver the Interests as provided in this Agreement and, upon consummation of the transactions contemplated hereby, such delivery will convey to Buyer, and Buyer will acquire, good and valid title to the Interests free and clear of any Encumbrances, other than restrictions on transfer that may be imposed by federal or state securities Laws.

#### 3.8 Capital Structure of the Company.

- (a) The authorized Membership Interests of the Company consist solely of the Interests. All outstanding Membership Interests of the Company are duly authorized, validly issued, fully paid and nonassessable. Except for the Interests, there are no outstanding (i) Membership Interests or other voting securities of the Company or any other Person convertible into or exchangeable or exercisable for Membership Interests or other voting securities of, or any other interest in, the Company, and (iii) subscriptions, options, warrants, calls, rights (including preemptive rights), commitments, understandings or agreements to which the Company is a party or by which it is bound obligating the Company to issue, deliver, sell, purchase, redeem or acquire Membership Interests or other voting securities of, or any other interest in, the Company (or securities convertible into or exchangeable or exercisable for Membership Interests or other voting securities of, or any other interest in, the Company) or obligating the Company to grant, extend or enter into any such subscription, option, warrant, call, right, commitment, understanding or agreement.
- (b) No Membership Interests in the Company have been issued in violation of, and none are subject to, any preemptive rights, purchase or call options, subscription rights, rights of first refusal or other similar rights except as set forth in the Company Organizational Documents. At the Closing, there will be no member agreement, irrevocable proxies, voting trust or other agreement or understanding relating to the voting of any Membership Interests of the Company. There are, and there will be as of the Closing, no outstanding stock appreciation, phantom stock, profit participation or similar rights with respect to the Company. There are no obligations,

contingent or otherwise, of the Company to provide funds to, or make any investment in (in the form of a loan, capital contribution or otherwise), or provide any guarantee with respect to the obligations of, any Person. There are no bonds, debentures, notes or other indebtedness of the Company having the right to vote or consent (or, convertible into, or exchangeable for, securities having the right to vote or consent) on any matters on which holders of Membership Interests of the Company may vote.

- 3.9 Subsidiaries; Assets. The Company has no Subsidiaries and holds no direct or indirect equity interest in any Person. Except for outstanding AFEs set forth on Schedule 3.16 as of the Execution Date, there are no outstanding obligations of the Company to provide funds to, or make any investment (in the form of a loan, capital contribution, purchase of an equity interest (whether from the issue or another Person) or otherwise) in, any Person. The Company does not, and has not since its inception, held any right, title or interest in or to any real or personal property or other asset, other than the Assets.
- 3.10 Litigation. Except as set forth in Schedule 3.10, there are no investigations, suits, actions or litigation by any Person by or before any Governmental Authority, and no legal, administrative or arbitration proceedings, (in each case) pending or, to SM Energy's Knowledge, threatened in writing against the Company, against SM Energy or the Company with respect to its or their ownership of the Assets, or against SM Energy with respect to its ownership of the Company. Except as set forth in Schedule 3.10, there is no judgment, settlement, order, decision, direction, writ, injunction, decree, stipulation or legal or arbitration award by any Person or before any Governmental Authority pending or, to SM Energy's Knowledge, threatened in writing against SM Energy, the Company or any of their respective Affiliates questioning the validity of or seeking to prevent the consummation of this Agreement and the transaction contemplated thereby or that materially interferes with or would be reasonably likely to interfere with the business of SM Energy, the Company or any of their respective Affiliates as currently conducted with respect to the Assets. To SM Energy's Knowledge, no condemnations or eminent domain proceedings affecting any of the Assets are pending or threatened in writing by any Governmental Authority.

#### 3.11 Liabilities.

The Company has no Liabilities that are unrelated to or otherwise do not arise from the Oil and Gas Properties or any Applicable Contract (excluding Liabilities related to SM Energy's breach of any such Contracts), whether accrued, absolute, contingent or otherwise, except as set forth on *Schedule 3.11*.

#### 3.12 Material Contracts.

- (a) Schedule 3.12(a) sets forth all Applicable Contracts of the type described below (collectively, the "Material Contracts"):
- (i) any Applicable Contract that can reasonably be expected to result in aggregate payments by or on behalf of the Company of more than \$100,000 during the current or any subsequent fiscal year or \$250,000 in the aggregate over the term of such Applicable Contract (in each case, based solely on the terms thereof);

- (ii) any Applicable Contract that can reasonably be expected to result in aggregate revenues paid to SM Energy or the Company of more than \$100,000 during the current or any subsequent fiscal year or \$250,000 in the aggregate over the term of such Applicable Contract (in each case, based solely on the terms thereof);
- (iii) any Applicable Contract that (A) is a Hydrocarbon purchase and sale, gathering, transportation, processing, compression or similar Contract pursuant to which SM Energy or the Company received annual revenues or makes annual payments in excess of \$100,000 and (B) is not terminable by or on behalf of the Company or its assignee without penalty on sixty (60) days' or less notice;
- (iv) any Applicable Contract that is an indenture, mortgage, loan, credit or sale-leaseback, guaranty of any obligation, bonds, letters of credit or similar financial Contract;
- (v) any Applicable Contract (A) with any former officer, director, equity holder, manager or member of SM Energy or the Company, (B) providing for severance, retention, change in control or other similar payments payable by the Company to any current or former officer, manager or employee of SM Energy, and (C) with independent contractors or consultants (or similar arrangements) that are not cancelable without penalty or further payment without more than thirty (30) days' notice;
  - (vi) any Applicable Contract of guaranty or surety, direct or indirect by the Company;
  - (vii) any Applicable Contract granting a power of attorney with respect to the affairs of the Company;
- (viii) any Applicable Contract that constitutes a lease under which the Company is the lessor or the lessee of real, immovable, personal or movable property which lease (A) cannot be terminated by or on behalf of the Company without penalty upon sixty (60) days or less notice and (B) involves an annual base rental of more than \$100,000;
- (ix) any Applicable Contract between SM Energy and any Affiliate of SM Energy that will be binding on Buyer or the Company after the Effective Time, or between SM Energy and any Affiliate of SM Energy (excluding the Company) on the one hand and the Company on the other;
- (x) any Applicable Contract that is a farmout or farmin agreement, joint venture agreement, term assignment, participation agreement, exploration agreement, development agreement, joint operating agreement, operating agreement, unit agreement, pooling agreement or similar Contract;
- (xi) any Applicable Contract that contains any calls on, or options to purchase, material quantities of Hydrocarbon production, other than pursuant to currently

effective Hydrocarbon purchase and sale contracts to which the Assets will be subject after Closing;

- (xii) any existing Applicable Contract that is a seismic or other geophysical acquisition or sharing agreement or license for which Buyer will be liable;
  - (xiii) any Applicable Contract, the primary purpose of which is to provide indemnity rights;
- (xiv) any Applicable Contract that (A) contains or constitutes an existing area of mutual interest agreement or (B) includes non-competition restrictions, non-solicitation or no-hire obligations; and
  - (xv) any amendments to any of the foregoing.
- (b) Except as set forth on *Schedule 3.12(b)* there exist no material default under the Material Contracts by SM Energy or the Company or, to SM Energy's or the Company's Knowledge, by any other Person that is a party to such Material Contracts and to SM Energy's or the Company's Knowledge the Material Contracts are in full force and effect in accordance with their respective terms. Except for the agreements set forth on *Schedule 3.12(b)*, there are no Applicable Contracts that contain mandatory drilling requirements with respect to the Assets (excluding any continuous drilling provisions included in a Lease or sublease required to hold all or any portion of the lands covered by such Lease or sublease), which obligations have not yet been fulfilled. As of the Execution Date, SM Energy has made available to Buyer complete and accurate copies of all Material Contracts, including any and all amendments and supplements thereto.
- 3.13 No Violation of Laws. Except as set forth on Schedule 3.13, neither SM Energy, the Company, nor, to SM Energy's Knowledge, any Third Party operator, is in material non-compliance with or material violation of any applicable Laws, including with respect to the ownership of the Assets. This Section 3.13 does not include any matters with respect to Environmental Laws, such matters being addressed in Section 3.17.
- 3.14 Royalties, Etc. Except as set forth on Schedule 3.14, SM Energy or the Company, as applicable, has properly and timely paid, or caused to be paid, all Burdens in all material respects due by SM Energy or the Company, as applicable, with respect to the Assets prior to the Execution Date in accordance with Law and the applicable Lease. Schedule 3.14 sets forth an accurate list of all Suspense Funds (including interest owed thereon, if any) held by SM Energy as of the Effective Time.
  - 3.15 Imbalances. Schedule 3.15 sets forth all material Imbalances associated with the Assets as of the Effective Time.
- 3.16 Current Commitments. Schedule 3.16 sets forth, as of the Execution Date, all authorities for expenditures related to the Assets to drill or rework Wells ("AFEs") which are outstanding or for which all of the activities anticipated in such AFEs have not been completed by the Effective Time.

#### 3.17 Environmental.

- (a) With respect to the Assets, neither SM Energy nor the Company, nor its or their Affiliates has entered into, nor is subject to, any agreement, consent, order, decree, judgment, license or permit condition or other directive of any Governmental Authority that (i) are in existence as of the Execution Date, (ii) are based on any Environmental Laws that relate to the future use of any of the Assets and (iii) require any change in the present conditions of any of the Assets.
- (b) Except as set forth in *Schedule 3.17*, as of the Execution Date neither SM Energy nor the Company, with respect to the Assets, has received any unresolved written notice from any Person of any release, disposal, event, condition, circumstance, activity, practice or incident concerning any land, facility, asset or property included in the Assets that: (i) materially interferes with or prevents compliance by SM Energy or the Company, as applicable, with any Environmental Law or the terms of any license or permit issued pursuant thereto or (ii) gives rise to or results in any material common Law or other Liability of SM Energy, the Company or any other Person.
- (c) Except as disclosed on *Schedule 3.17*, (i) there are no actions, suits or proceedings pending, or to SM Energy's Knowledge, threatened before any Governmental Authority or arbitrator with respect to the Assets alleging material violations of, or material liabilities under, Environmental Laws, or claiming Remediation obligations, and (ii) SM Energy has made available to Buyer all Third Party reports or studies commissioned by SM Energy, the Company or any Third Party Operator relating to the Assets, assessing the environmental condition of any of the Assets during the five (5) year period prior to the Execution Date and that are in SM Energy's, the Company's, or its or their Affiliate's possession or control.

#### 3.18 Taxes. Except as set forth in Schedule 3.18:

- (a) the Company has filed all material Tax Returns (including returns for Asset Taxes) required to be filed by or with respect to it or the Assets, all such Tax Returns are true, correct and complete in all material respects and the Company has timely paid all Taxes reflected as due on such Tax Returns;
- (b) no penalty, interest or other charge is or will become due with respect to the late filing of any such Tax Return or late payment of any such Tax;
  - (c) the Company is treated as a disregarded entity for U.S. federal income Tax purposes;
  - (d) there are no Encumbrances (other than Permitted Encumbrances) on any of the Interests or the Assets for Taxes;
- (e) the Company does not have in force any waiver of any statute of limitations in respect of Taxes or any extension of time with respect to a Tax assessment or deficiency;
- (f) neither SM Energy nor the Company has received any written notice of any pending claim (which remains outstanding) from any applicable Tax authority for assessment of

Taxes with respect to the Interests, the Company or the Assets and no such claim has been made or threatened; and

- (g) no audit, administrative, judicial or other proceeding by any Governmental Authority with respect to Taxes with respect to the Interests, the Company or the Assets has been commenced or is presently pending.
- 3.19 Brokers' Fees. SM Energy has incurred no liability, contingent or otherwise, for brokers' or finders' fees relating to the transactions contemplated by this Agreement for which Buyer, the Company, or any Affiliate of Buyer shall have any responsibility.
- **3.20** Advance Payments. The Company is not obligated by virtue of any take-or-pay payment, advance payment or other similar payment (other than royalties, overriding royalties and similar arrangements reflected with respect to the Net Revenue Interests for the Wells set forth in **Exhibit B** and gas balancing arrangements), to deliver Hydrocarbons attributable to the Assets, or proceeds from the sale thereof in accordance with its ownership share, at some future time without receiving payment therefore at or after the time of delivery.
- 3.21 Partnerships. Schedule 3.21 sets forth all of the Assets that are deemed by agreement or applicable Law to be held by a partnership for federal Tax purposes. All such partnerships have in effect an election under Section 754 of the Code and have made the election to deduct intangible drilling costs, if applicable, and none of the partnerships have any adjustment, claim, indemnity arrangement, or other method of allocating risk with respect to the effects of Section 708(b)(1)(B) of the Code.
- **3.22** Hedge Contracts. Except for the Hedge Contracts executed pursuant to the terms of this Agreement, neither SM Energy nor the Company, nor its or their Affiliates has entered into any Hedge Contract pursuant to which any production of Hydrocarbons from any of the Assets is dedicated or committed to as of or after the Effective Time.
- 3.23 Insurance. SM Energy or the Company, as applicable, maintains with respect to the Assets the insurance coverage described on Schedule 3.23.
- 3.24 Equipment and Personal Property. Except as disclosed on Schedule 3.24, to SM Energy's Knowledge, all Third Party Operators of the Assets have all easements, rights of way, licenses and authorizations, including those from Governmental Authorities necessary to access, construct, operate, maintain and repair the Wells and equipment included in the Assets in the ordinary course of business as currently conducted by SM Energy or the Company and in compliance with all Laws, including the operation and maintenance of the Springfield Gathering System. To SM Energy's Knowledge, all equipment used or held for use in connection with the Assets, including the operation and maintenance of the Springfield Gathering System, has been maintained in working order and operating condition in all material respects and is adequate for normal operation of the Assets in all material respects consistent with current practices, ordinary wear and tear excepted.
- 3.25 Wells; Plug and Abandon Notice. As of the Execution Date, except as set forth on Schedule 3.25, (a) there are no Wells in respect of which SM Energy, the Company, or its or their

Affiliates have received a written order from any Governmental Authority or a written demand from any Third Party (in each case) requiring that such Wells be plugged and abandoned, or (b) to SM Energy's Knowledge, in use for purposes of production or injection or suspended or temporarily abandoned in accordance with applicable Laws that (i) are required to be plugged and abandoned in accordance with applicable Laws or any Lease and (ii) have not been or are not in the process of being plugged and abandoned.

- 3.26 Permits. To SM Energy's Knowledge, (a) all Third Party Operators of the Assets, including the Springfield Gathering System, possess all material permits, licenses, orders, approvals, variances, waivers, franchise rights and other authorizations required to be obtained from any Governmental Authority for conducting their business with respect to the Assets and (b) there are no material uncured violations by such Third Party Operators of the terms and provisions of such authorizations.
- 3.27 Payouts. Schedule 3.27 contains a complete and accurate list of the status of any "payout" balance, as of the date indicated on such Schedule, for the Wells that are subject to a reversion or other adjustment at some level of cost recovery or payout (or passage of time or other event other than termination of a Lease by its terms).
- 3.28 Bonds. Schedule 3.28 lists all bonds, letters of credit, guarantees and other similar credit support instruments maintained by SM Energy, the Company or any of its or their Affiliates with respect to the Assets.
- 3.29 Preferential Purchase Rights. Except as set forth on Schedule 6.4, there are no preferential purchase rights, drag-along rights, tag-along rights or other similar rights that are applicable to the consummation of the transactions contemplated by this Agreement by SM Energy, including the Contribution.
- 3.30 Availability of Properties. The Assets constitute and include all of the assets used or held for use by SM Energy or the Company in the conduct of the business of the Company with respect to ownership of the Assets as presently owned by SM Energy and the Company in the ordinary course of business, consistent with past practices. Since the Effective Time, except pursuant to the Contribution, neither SM Energy nor the Company have transferred, conveyed (including any conveyance or reservation of overriding royalties, net profits interests or other leasehold burdens), farmed out, sold, encumbered, removed, abandoned, let lapse or expire, terminated or otherwise disposed of its interest in any of the Assets, or agreed to do any of the foregoing, except for sales and dispositions of Hydrocarbon production in the ordinary course of business.
- 3.31 Employees. The Company does not have, and has never had, any officer, managers or employees. The Company is not party to or obligated under any consulting, independent contractor, employment, change of control, severance, termination or similar arrangement with respect to its current or former consultants or independent contractors, or current or former officers, manager, employees, consultants or independent contractors of SM Energy.
- 3.32 Employee Benefits. The Company does not sponsor, maintain or contribute to or have any obligation with respect to any Benefit Plan.

- 3.33 Affiliate Transactions. No officer, director, manager, equity holder, partner or member of the Company, SM Energy or any of their respective Affiliates, or, with respect to natural persons, any member of his or her immediate family ("Related Persons") (a) owes any amount to the Company nor does the Company owe any amount to, or has the Company committed to make any loan or extend or guarantee credit to or for the benefit of, any Related Person, (b) is involved in any business arrangement with the Company (whether written or oral), (c) owns any property or right, tangible or intangible, that is used by the Company, (d) has any claim or cause of action against the Company or (e) owns any direct or indirect interest of any kind in, or controls or is a manager, director, officer, employee or partner of, or consultant to, or lender to or borrower from or has the right to participate in the profits of, any Person which is a competitor, supplier, service provider, customer, client, landlord, tenant, creditor or debtor of the Company.
- 3.34 Unlawful Payments. The Company has not paid, given or received, and has not offered or promised to pay, give or receive, any bribe or other unlawful payment of money or other thing of value in violation of applicable Law, any unlawful discount, or any other unlawful inducement, to or from any Person or Governmental Authority in connection with or in furtherance of the business of the Company (including any offer, payment or promise to pay money or other thing of value (a) to any foreign official (including all employees or management officials of companies owned or controlled by a foreign government), political party (or official thereof) or candidate for political office for the purposes of influencing any act, decision or omission in order to assist the Company in obtaining or retaining business for or with, or directing business to, any Person, or otherwise obtaining any advantage or (b) to any Person, while knowing that all or a portion of such money or other thing of value will be offered, given or promised to any such official or party for such purposes). The Company has not otherwise taken any action that would cause it to be in violation of the Foreign Corrupt Practices Act of 1977, as amended, or any applicable Law of similar effect. There have been no false or fictitious entries made in the books or records of the Company relating to any such illegal payment or secret or unrecorded fund.
- 3.35 Contribution. As of the Closing Date, except for Assets related to the unobtained Outstanding Contribution Consents, all Contribution Assets have been contributed to the Company. Schedule 3.35 sets forth all documents, instruments, certificates, agreements, assignments or conveyances delivered or to be delivered as part of or in conjunction with the Contribution (collectively, the "Contribution Documents"). The Company has made available or will make available prior to Closing to Buyer true and complete copies of all Contribution Documents. All covenants and agreements of SM Energy and the Company with respect to the Contribution Documents have been fulfilled as of the Closing Date and there are no outstanding obligations of SM Energy, its Affiliate or the Company with respect to the Contribution Documents. Except for the Outstanding Contribution Consents, no Preferential Rights or Consents related to the Contribution remain outstanding.
- 3.36 Springfield Financial Information. SM Energy has provided Buyer with true and complete copies of (i) the annual budgets for 2015, 2016 and 2017 provided to SM Energy under Section 4.3(B) of the COO Agreement, and (ii) the monthly reports for 2015 and 2016 provided to SM Energy under Section 4.3(F) of the COO Agreement.

- 3.37 Springfield System Ownership Interests. As of the Closing Date, the Company has, and after the Closing will still have, good and valid title to, and hold beneficially, free and clear of any Encumbrances created by SM Energy or the Company other than those imposed by the COO Agreement, the Springfield System Ownership Interest, which gives and will after the Closing give the Company a twelve point five percent (12.5%) ownership interest in the Springfield Gathering System. The COO Agreement is in full force and effect and represents the legal, valid and binding obligation of the Company and, to the Knowledge of the Company, the other parties thereto, in each case enforceable in accordance with its terms. The Company is not in breach, violation or default under the COO Agreement, and the execution, delivery, and performance of this Agreement and the consummation of the transactions contemplated hereby will not violate, result in any breach of, constitute a default under, or give to others any rights of termination, purchase, acceleration or cancellation, or other rights or remedies, under, the COO Agreement.
- No Other Representations or Warranties; Disclosed Materials. Except for the representations, warranties and covenants contained in this Agreement (as qualified by the Schedules) and the special warranty of title contained in the Contribution Documents, neither SM Energy nor any other Person makes (and Buyer is not relying upon) any other express or implied representation or warranty with respect to SM Energy (including the value, condition or use of any of the Assets) or the transactions contemplated by this Agreement, and SM Energy disclaims any other representations or warranties not contained in this Agreement, whether made by SM Energy, any Affiliate of SM Energy, or any of their respective officers, directors, managers, employees or agents, Except for the representations and warranties contained in this Agreement (as qualified by the Schedules) and the special warranty of title contained in the Contribution Documents, SM Energy disclaims all liability and responsibility for any representation, warranty, projection, forecast, statement or information made, communicated or furnished (orally or in writing) to Buyer or any of its Affiliates or any of its officers, directors, managers, employees or agents (including any opinion, information, projection or advice that may have been or may be provided to Buyer by any director, officer, employee, agent, consultant or representative of SM Energy, or any of its Affiliates). Buyer acknowledges and represents, warrants and agrees that it has not relied upon the accuracy or completeness of any express or implied representation, warranty, statement or information of any nature made or provided by or on behalf of SM Energy, except for the representations and warranties of SM Energy expressly set forth in this Agreement and the special warranty of title contained in the Contribution Documents, and waives any right Buyer may have against SM Energy with respect to any inaccuracy in any such representation, warranty, statement or information, or with respect to any omission or concealment, on the part of SM Energy or any representative of SM Energy, of any potentially material information. The disclosure of any matter or item in the Schedules shall not be deemed to constitute an acknowledgement that any such matter is required to be disclosed or is material or that such matter would or would reasonably be expected to result in a Material Adverse Effect. From time to time prior to the Closing Date, SM Energy shall have the right (but not the obligation) solely with respect to matters that occur during the period from the date hereof to the Closing Date, to supplement or amend the Schedules hereto to correct any matter that would otherwise constitute a breach of any representation or warranty of SM Energy contained herein (each a "Schedule Supplement"), and each such Schedule Supplement shall be deemed to be incorporated into and supplement and amend the Schedules as of the Closing Date;

provided, however, that any such Schedule Supplement shall be disregarded for purposes of, and shall not affect Buyer's conditions to Closing set forth in Section 9.1.

## ARTICLE IV REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to SM Energy as of the Execution Date and the Closing Date (except as expressly set forth in this *ARTICLE IV*) the following:

- 4.1 Organization; Existence. Buyer is a limited liability company duly formed and validly existing under the Laws of Delaware. Buyer has, or will have as of the Closing Date, all requisite power and authority to own and operate its property (including, at Closing, the Assets) and to carry on its business as now conducted. Buyer is duly licensed or qualified to do business as a foreign limited liability company and is in good standing in all jurisdictions in which such qualification is required by Law except where the failure to qualify or be in good standing would not have a material adverse effect upon the ability of Buyer to consummate the transactions contemplated by this Agreement.
- 4.2 Authorization. Buyer has full power and authority to enter into and perform this Agreement to which it is a party and the transactions contemplated herein. The execution, delivery and performance by Buyer of this Agreement has been, and the execution, delivery and performance by Buyer of all other documents delivered pursuant to this Agreement will be when delivered, duly and validly authorized and approved by all necessary limited liability company action on the part of Buyer. Assuming the due authorization, execution and delivery by the other parties to such documents, this Agreement constitutes, and the other documents delivered pursuant to this Agreement to which Buyer is a party will constitute, Buyer's legal, valid and binding obligations, enforceable against Buyer in accordance with their respective terms, subject however to the effects of bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar Laws relating to or affecting creditors' rights, as well as to principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).
- 4.3 No Conflicts. Assuming the receipt of all Consents and the waiver of all Preferential Rights, the execution, delivery and performance by Buyer of this Agreement and the consummation of the transactions contemplated herein will not (a) conflict with or result in a breach of any provisions of the organizational or other governing documents of Buyer, (b) result in a default or the creation of any Encumbrance or give rise to any right of termination, cancellation or acceleration under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license or other agreement to which Buyer is a party or by which Buyer or any of its property may be bound or (c) violate any Law applicable to Buyer or any of its property, except in the case of clauses (b) and (c) where such default, Encumbrance, termination, cancellation, acceleration or violation would not have a material adverse effect upon the ability of Buyer to consummate the transactions contemplated by this Agreement.
- **4.4 Bankruptcy**. There are no bankruptcy, reorganization or receivership proceedings pending, being contemplated by or, to Buyer's Knowledge, threatened in writing against Buyer.

- **4.5** *Litigation*. There is no investigation, suit, action or litigation by any Person or by or before any Governmental Authority, and no legal, administrative or arbitration proceedings pending, or to Buyer's Knowledge, threatened against Buyer, or to which Buyer is a party, that would affect the ability of Buyer to consummate the transactions contemplated by this Agreement.
- **4.6** *Financing*. Buyer has, and shall have as of the Closing Date, sufficient funds (including available borrowing capacity or available equity commitments) with which to pay the Closing Amount and consummate the transactions contemplated by this Agreement and, following Closing, Buyer will have sufficient funds to pay any adjustments to the Purchase Price and meet its other payment obligations under this Agreement as such obligations are due and payable.
- 4.7 Independent Evaluation. Buyer is sophisticated in the evaluation, purchase, ownership and operation of oil and gas properties and related facilities. In making its decision to enter into this Agreement and to consummate the transaction contemplated herein, Buyer (a) has relied or shall rely solely on its own independent investigation and evaluation of the Assets and the advice of its own legal, tax, economic, environmental, engineering, geological and geophysical advisors and the express provisions of this Agreement and not on any comments, statements, projections or other materials made or given by any representatives or consultants or advisors engaged by SM Energy, and (b) has satisfied or shall satisfy itself through its own due diligence as to the environmental and physical condition of and contractual arrangements and other matters affecting the Assets.
- 4.8 Brokers' Fees. Buyer has incurred no liability, contingent or otherwise, for brokers' or finders' fees relating to the transactions contemplated by this Agreement for which SM Energy or any Affiliate of SM Energy shall have any responsibility.
- 4.9 Accredited Investor. Buyer is acquiring the Interests for its own account for use in its trade or business and not with a view toward any sale or distribution thereof, or with any present intention of making a distribution thereof within the meaning of the Securities Act of 1933, as amended, and the rules and regulations thereunder, any applicable state "blue sky" Laws or other applicable securities Laws. Buyer is an Accredited Investor (as defined in Rule 501 of Regulation D promulgated under the Securities Act of 1933, as amended), and Buyer itself, or through its representatives, has sufficient knowledge and experience in financial and business matters to be capable of evaluating the merits and risks of an investment such as an investment in the Interests. Buyer, either alone or through its representatives, has evaluated the merits and risks of the investment in the Interests and is able to bear the economic risk and lack of liquidity inherent in holding the Interests.

## ARTICLE V ACCESS / DISCLAIMERS

### 5.1 Access.

(a) From and after the Execution Date and up to and including the Closing Date (or earlier termination of this Agreement), but subject to the other provisions of this Agreement (including this *Section 5.1*) and obtaining any required consents of Third Parties, including Third

Party Operators of the Assets (with respect to which consents SM Energy shall use its commercially reasonable efforts obtain or to cause the Company to obtain but SM Energy shall not be required to pay any consideration therefor), SM Energy shall afford to Buyer, its Affiliates and its and their officers, employees, agents, accountants, attorneys, investment bankers, consultants and other authorized representatives (collectively, "Buyer's Representatives") reasonable access, during normal business hours, to the Assets and all Records in SM Energy's or any of its Affiliates' possession; provided, however, that if SM Energy refuses to permit such reasonable access then the Assets with respect to which Buyer requested permission to access may, at Buyer's option, be excluded from the Assets to be conveyed to Buyer at Closing and the Purchase Price shall be adjusted downward by the Allocated Value of such Assets. SM Energy shall also make available to Buyer and Buyer's Representatives, upon reasonable notice during normal business hours, SM Energy's personnel knowledgeable with respect to the Assets in order that Buyer may make such diligence investigation as Buyer considers necessary or appropriate. All investigations and due diligence conducted by Buyer or any Buyer's Representative shall be conducted at Buyer's sole cost, risk and expense; and any conclusions made from any examination done by Buyer or any Buyer's Representative shall result from Buyer's own independent review and judgment. Buyer shall coordinate its access rights and physical inspections of the Assets with SM Energy and any Third Party that serves as an Operator (each a "Third Party Operator") to reasonably minimize any inconvenience to or interruption of the conduct of business by SM Energy, the Company or any Third Party Operator. Buyer acknowledges and agrees that any Third Party Operator may require Buyer to execute and deliver a reasonable and customary boarding or similar agreement prior to obtaining access to the Assets. Buyer shall give SM Energy reasonable prior written notice before entering onto any of the Assets and SM Energy shall have the right to have its representatives present at any time any Buyer's Representative is present on the Assets. Buyer shall, and shall cause all of the Buyer's Representatives to, abide by SM Energy's and any Third Party Operator's safety rules, regulations and operating policies while conducting its due diligence evaluation of the Assets including any environmental or other inspection or assessment of the Assets; provided, that such rules, regulations and operating procedures have been provided to Buyer.

(b) Buyer shall be entitled to conduct a Phase I or other visual environmental property assessment with respect to the Assets. Any Third Party Operator, SM Energy or any of their respective designees shall have the right to accompany Buyer and Buyer's Representatives whenever they are on site on the Assets and also to collect split test samples if any are collected. Notwithstanding anything herein to the contrary, Buyer shall not have access to, and shall not be permitted to conduct, any environmental due diligence (including any Phase I environmental property assessments) with respect to any Assets where SM Energy does not have the authority to grant access for such due diligence (provided, however, SM Energy shall use its commercially reasonable efforts to obtain permission from any Third Party to allow Buyer and Buyer's Representatives such access but SM Energy shall not be required to pay any consideration therefor). If following the conduct of the Phase I or other visual environmental property assessment, Buyer reasonably believes that it is necessary to conduct any sampling, boring, drilling or other invasive investigation activities on or with respect to any of the Assets to determine the existence or magnitude of an Environmental Defect (such actions, an "Invasive Activity"), Buyer shall furnish to SM Energy for its review no later than five (5) Business Days prior to the Defect Claim Date a proposed scope of such Invasive Activity, including a reasonable description of such activity and a description of

the approximate locations of any sampling to be conducted. Following the receipt and review of such proposal by SM Energy, SM Energy shall (x) have five (5) Business Days to consent to such Invasive Activity, in its reasonable discretion, which consent will be considered granted by SM Energy if SM Energy fails to respond within five (5) Business Days after receiving such proposal, and (y) promptly use its commercially reasonable efforts to obtain permission from any Third Party Operator to allow Buyer and Buyer's Representatives access to conduct such Invasive Activity but SM Energy shall not be required to pay any consideration therefor. If SM Energy and any Third Party Operator consent to such Invasive Activity, then notwithstanding the Defect Claim Date, Buyer shall have the right to assert Environmental Defects related to the Assets for which Buyer requested such Invasive Activity by providing SM Energy with a valid Defect Notice covering such Environmental Defect on or before the date that is five (5) Business Days after the Defect Claim Date. If SM Energy or any Third Party Operator refuses to permit the conduct of such Invasive Activity, then the Environmental Condition impacting the Assets with respect to which Buyer requested permission to conduct such Invasive Activity may, at Buyer's option, be asserted as an Environmental Defect, and in such event, subject to SM Energy's right to dispute the existence of the Environmental Condition or the related Remediation Amount, shall be deemed an Environmental Defect.

- (c) Buyer agrees to defend, indemnify and hold harmless each of the Operators of the Assets and the SM Indemnified Parties or Third Party Operators from and against any and all Liabilities attributable to personal injury, death, or damage to or destruction of property and arising out of, resulting from or relating to any field visit, environmental property assessment or other due diligence activity conducted by Buyer or any Buyer's Representative (including an Invasive Activity, if any) with respect to the Assets, EVEN IF SUCH LIABILITIES ARISE OUT OF OR RESULT FROM, SOLELY OR IN PART, THE SOLE, ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY A MEMBER OF THE SM INDEMNIFIED PARTIES, EXCEPTING ONLY LIABILITIES ACTUALLY RESULTING ON THE ACCOUNT OF THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF A MEMBER OF THE SM INDEMNIFIED PARTIES OR OPERATOR.
- (d) Buyer agrees to promptly provide SM Energy with copies, but in any event within the earlier of (x) ten (10) days after its receipt of any report or test results, or (y) the Defect Claim Date, of all final reports and test results prepared by Buyer, any of Buyer's Representatives or any Third Party consultants and which contain data collected or generated from Buyer's due diligence with respect to the Assets (including an Invasive Activity, if any), in each case, to the extent related to any Environmental Defect Buyer asserts or reasonably expects to assert (including such portion that evidences such potential Environmental Defect, if any, and the curative recommendation thereto). Buyer shall not be deemed by its delivery of said documents or otherwise to have made any representation or warranty, expressed, implied or statutory as to the condition to the Assets or to the accuracy of said documents or the information contained therein. Except as required by Law, Buyer and Buyer's Representatives shall keep and maintain confidential (and not disclose to any other Person, in whole or in part) any results of (or reports or other documents arising from) an environmental assessment of the Assets and shall use such results, reports or other documents arising therefrom solely for purposes of Buyer's diligence.

- (e) Upon completion of Buyer's due diligence, Buyer shall, at its sole cost and expense and without any cost or expense to any Third Party Operator or SM Energy or its Affiliates (including the Company) (i) repair all damage done to the Assets resulting from Buyer's due diligence and (ii) remove all equipment, tools or other property brought onto the Assets in connection with Buyer's due diligence. Any disturbance to the Assets (including, without limitation, the real property associated with such Assets) resulting from Buyer's due diligence will be promptly corrected by Buyer.
- (f) During all periods that Buyer or any of Buyer's Representatives are present on the Assets, Buyer or Buyer's Representatives shall maintain, at its or their sole expense, policies of insurance of the types and in the amounts customary for conducting environmental due diligence reviews of oil and gas assets, which in no event will constitute less than \$10,000,000 of general liability insurance. Upon request by SM Energy, Buyer shall provide evidence of such insurance to SM Energy prior to entering the Assets.
- 5.2 Confidentiality. Buyer acknowledges that, pursuant to its right of access to the Records and the Assets, Buyer will become privy to confidential and other information of SM Energy and the Company and that such confidential information shall be held confidential by Buyer and Buyer's Representatives in accordance with the terms of the Confidentiality Agreement. If Closing should occur, the foregoing confidentiality restriction on Buyer, including the Confidentiality Agreement and the confidentiality restriction in Section 5.1(d), shall terminate (except as to (a) any Assets that are excluded from the transactions contemplated hereby pursuant to the provisions of this Agreement, (b) the Excluded Assets, and (c) information related to assets other than the Assets).

#### 5.3 Disclaimers.

- (a) EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN *ARTICLE III*, THE CERTIFICATE DELIVERED BY SM ENERGY PURSUANT TO *SECTION 11.3(H)* OR THE SPECIAL WARRANTY OF TITLE CONTAINED IN THE CONTRIBUTION DOCUMENTS, (I) NEITHER SM ENERGY NOR THE COMPANY MAKES ANY REPRESENTATIONS OR WARRANTIES, EXPRESS, STATUTORY OR IMPLIED, AND (II) SM ENERGY AND THE COMPANY EXPRESSLY DISCLAIM ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, STATEMENT OR INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO BUYER OR ANY BUYER'S REPRESENTATIVE (INCLUDING, WITHOUT LIMITATION, ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN PROVIDED TO BUYER BY ANY OFFICER, DIRECTOR, EMPLOYEE, AGENT, CONSULTANT, REPRESENTATIVE OR ADVISOR OF SM ENERGY OR ANY OF ITS AFFILIATES).
- (b) EXCEPT AS EXPRESSLY SET FORTH IN *ARTICLE III*, THE CERTIFICATE DELIVERED BY SM ENERGY PURSUANT TO *SECTION 11.3(H)* OR THE SPECIAL WARRANTY OF TITLE CONTAINED IN THE CONTRIBUTION DOCUMENTS, AND WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, SM ENERGY AND THE COMPANY EXPRESSLY DISCLAIM ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY OR IMPLIED, AS TO (I) TITLE TO ANY OF THE ASSETS, (II) THE

CONTENTS, CHARACTER OR NATURE OF ANY REPORT OF ANY PETROLEUM ENGINEERING CONSULTANT, OR ANY ENGINEERING, GEOLOGICAL OR SEISMIC DATA OR INTERPRETATION, RELATING TO THE ASSETS, (III) THE OUANTITY, QUALITY OR RECOVERABILITY OF HYDROCARBONS IN OR FROM THE ASSETS, (IV) ANY ESTIMATES OF THE VALUE OF THE ASSETS OR FUTURE REVENUES GENERATED BY THE ASSETS, (V) THE ABILITY TO PRODUCE HYDROCARBONS FROM THE ASSETS, (VI) THE MAINTENANCE, REPAIR, CONDITION, QUALITY, SUITABILITY, DESIGN OR MARKETABILITY OF THE ASSETS, (VII) THE CONTENT, CHARACTER OR NATURE OF ANY INFORMATION MEMORANDUM, REPORTS, BROCHURES, CHARTS OR STATEMENTS PREPARED BY SM ENERGY OR THIRD PARTIES WITH RESPECT TO THE ASSETS, (VIII) ANY OTHER MATERIALS OR INFORMATION THAT MAY HAVE BEEN MADE AVAILABLE TO BUYER OR ANY BUYER'S REPRESENTATIVE IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY DISCUSSION OR PRESENTATION RELATING THERETO AND (IX) ANY IMPLIED OR EXPRESS WARRANTY OF FREEDOM FROM PATENT OR TRADEMARK INFRINGEMENT. EXCEPT AS EXPRESSLY REPRESENTED OTHERWISE IN ARTICLE III, THE CERTIFICATE DELIVERED BY SM ENERGY PURSUANT TO SECTION 11.3(H) OR THE SPECIAL WARRANTY OF TITLE CONTAINED IN THE CONTRIBUTION DOCUMENTS, SM ENERGY AND THE COMPANY FURTHER DISCLAIM ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY OR IMPLIED, OF MERCHANTABILITY, FREEDOM FROM LATENT VICES OR DEFECTS, FITNESS FOR A PARTICULAR PURPOSE OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS OF ANY ASSETS, RIGHTS OF A PURCHASER UNDER APPROPRIATE STATUTES TO CLAIM DIMINUTION OF CONSIDERATION OR RETURN OF CONSIDERATION, IT BEING EXPRESSLY UNDERSTOOD AND AGREED BY THE PARTIES THAT BUYER, SUBJECT TO ARTICLE III, ARTICLE VI, ARTICLE VII, AND ARTICLE XIII SHALL BE DEEMED TO BE OBTAINING THE ASSETS IN THEIR PRESENT STATUS, CONDITION AND STATE OF REPAIR, "AS IS" AND "WHERE IS" WITH ALL FAULTS OR DEFECTS (KNOWN OR UNKNOWN, LATENT, DISCOVERABLE OR UNDISCOVERABLE), AND THAT BUYER HAS MADE OR CAUSED TO BE MADE SUCH INSPECTIONS AS BUYER DEEMS APPROPRIATE.

(c) OTHER THAN THOSE REPRESENTATIONS SET FORTH IN SECTION 3.17, NEITHER SM ENERGY NOR THE COMPANY HAS MADE OR WILL MAKE ANY REPRESENTATION OR WARRANTY REGARDING ANY MATTER OR CIRCUMSTANCE RELATING TO ENVIRONMENTAL LAWS, THE RELEASE OF MATERIALS INTO THE ENVIRONMENT OR THE PROTECTION OF HUMAN HEALTH, SAFETY, NATURAL RESOURCES OR THE ENVIRONMENT, OR ANY OTHER ENVIRONMENTAL CONDITION OF THE ASSETS, AND NOTHING IN THIS AGREEMENT OR OTHERWISE SHALL BE CONSTRUED AS SUCH A REPRESENTATION OR WARRANTY, AND SUBJECT TO BUYER'S LIMITED RIGHTS UNDER ARTICLE VI, ARTICLE VII, AND ARTICLE XIII, BUYER SHALL BE DEEMED TO BE TAKING THE ASSETS "AS IS" AND "WHERE IS" WITH ALL FAULTS FOR PURPOSES OF THEIR ENVIRONMENTAL CONDITION AND THAT BUYER HAS MADE OR CAUSED TO BE MADE SUCH ENVIRONMENTAL INSPECTIONS AS BUYER DEEMS APPROPRIATE.

(d) SM ENERGY AND BUYER AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS SECTION 5.3 ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSE OF ANY APPLICABLE LAW.

## ARTICLE VI TITLE MATTERS; CASUALTIES

6.1 The Company's Title. Without limiting Buyer's remedies for Title Defects set forth in this ARTICLE VI or for any breach by SM Energy of its representations and warranties in Section 3.14 and its remedies under Section 13.2(k), and except as and to the extent of the special warranty of title contained in the Contribution Documents, SM Energy makes no warranty or representation, express, implied, statutory or otherwise with respect to the Company's title to any of the Assets and Buyer hereby acknowledges and agrees that Buyer's sole remedy for any Title Defect, with respect to any of the Assets shall be as set forth in Section 6.2.

#### 6.2 Notice of Title Defects; Defect Adjustments.

- (a) Title Defect Notices. On or before 4:00 p.m. (Mountain Time) on February 20, 2017 (the "Defect Claim Date"), Buyer shall have the right, but not the obligation, to deliver notices to SM Energy meeting the requirements of this Section 6.2(a) (each, a "Title Defect Notice") setting forth any matters which, in Buyer's reasonable opinion, constitute Title Defects and which Buyer asserts as a Title Defect pursuant to this Section 6.2. For all purposes of this Agreement and notwithstanding anything herein to the contrary (except for Buyer's remedies for any breach by SM Energy of its representations and warranties in Section 3.14 and its remedies under Section 13.2(k)), Buyer shall be deemed to have waived, and neither SM Energy nor the Company shall have any liability for, any Title Defect that Buyer fails to assert as a Title Defect pursuant to a Title Defect Notice delivered in compliance with this Section 6.2(a) and received by SM Energy on or before the Defect Claim Date. To be effective, each Title Defect Notice shall be in writing and shall include (i) a description of the alleged Title Defect and the Assets affected by such Title Defect (each a "Title Defect Property"), (ii) the Allocated Value of each Title Defect Property, (iii) supporting documents reasonably necessary for SM Energy to identify the existence of the alleged Title Defect(s), (iv) the amount by which Buyer reasonably believes the Allocated Value of each Title Defect Property is reduced by the alleged Title Defect(s), and (v) the computations upon which Buyer's belief is based. To give SM Energy an opportunity to commence reviewing and curing Title Defects. Buyer agrees to use reasonable efforts to give SM Energy, on or before the end of each two-week period prior to the Defect Claim Date, written notice of all alleged Title Defects discovered by Buyer during the preceding two-week period, which notice (in any format) may be preliminary in nature and supplemented prior to the expiration of the applicable Defect Claim Date; provided, that the failure of Buyer to provide such preliminary notice of any Title Defects shall not be deemed to waive or otherwise prejudice Buyer's right to assert a Title Defect on or before the Defect Claim Date.
- (b) <u>Title Benefit Notices</u>. SM Energy shall have the right, but not the obligation, to deliver to Buyer on or before the Defect Claim Date, a notice setting forth any matters that, in SM Energy's reasonable opinion, constitute Title Benefits and that SM Energy asserts as a Title

Benefit pursuant to this Section 6.2 (each, a "Title Benefit Notice") including (i) a description of the Title Benefit and the Assets affected by the Title Benefit (the "Title Benefit Property"), (i) the Allocated Value of each Title Benefit Property, (iii) supporting documents reasonably necessary for Buyer to identify the existence of the alleged Title Benefit(s), (iv) the amount by which SM Energy reasonably believes the Allocated Value of the Title Benefit Property is increased by the Title Benefit, and (v) the computations upon which SM Energy's belief is based. SM Energy shall be deemed to have waived all Title Benefits of which it, or Buyer pursuant to Section 6.2(b), has not given notice on or before the Defect Claim Date.

- (c) <u>Remedies for Title Defects</u>. Subject to SM Energy's continuing right to dispute the existence of a Title Defect or the Title Defect Amount asserted with respect thereto and subject to the Individual Title Defect Threshold and the Aggregate Deductible, in the event that any Title Defect properly asserted by Buyer in accordance with *Section 6.2(a)* is not waived in writing by Buyer or cured on or before Closing, SM Energy shall, at its sole option, elect to:
  - (i) reduce the Purchase Price by the Title Defect Amount;
  - (ii) notify Buyer that SM Energy elects, for a period of one hundred ten (110) days following the Closing Date (such period, the "Cure Period"), to cure any Title Defect relating to such Title Defect Property in which event the applicable Title Defect Property and all associated Assets shall remain in the Company at the Closing, and (A) without application of the Aggregate Deductible the Purchase Price paid to SM Energy at Closing shall be reduced by the amount asserted by Buyer in the applicable Title Defect Notice attributable to such Title Defect, and (B) Buyer shall deliver to the Escrow Agent at Closing the amount set forth in Section 6.2(c)(ii)(A) above. If, (A) at the expiration of the Cure Period SM Energy has failed to cure all or any portion of a Title Defect then the amounts deposited in the Escrow Account with respect to such failed cure (or portion thereof) shall be released to Buyer and within five (5) Business Days after the expiration of the Cure Period the Parties shall jointly instruct the Escrow Account with respect to such cure, or partial cure, as applicable shall be released to SM Energy and within five (5) Business Days after the expiration of the Cure Period the Parties shall jointly instruct the Escrow Agent to release such amounts to SM Energy; or
  - (iii) solely to the extent Buyer agrees in writing, indemnify Buyer against all Liability resulting from such Title Defect with respect to the Assets pursuant to an indemnity agreement mutually agreeable to the Parties.
- (d) Remedies for Title Benefits. With respect to each Well or Well Location affected by Title Benefits reported under this Section 6.2, the aggregate Title Defect Amounts shall be decreased by an amount equal to the increase in the Allocated Value for such Asset caused by such Title Benefits, as determined pursuant to Section 6.2(g) (the "Title Benefit Amount"). For the avoidance of doubt, Title Benefit Amounts shall only offset Title Defect Amounts.

- (e) Exclusive Remedy. Except for Buyer's (i) rights to terminate this Agreement pursuant to Section 12.1(d) and (ii) remedies for any breach by SM Energy of its representations and warranties in Section 3.14, Section 3.35 (along with Buyer's rights under ARTICLE XIII with respect therewith) and the special warranty of title in the Contribution Documents, the provisions set forth in Section 6.2(c) and Section 6.2(i) shall be the exclusive right and remedy of Buyer with respect to the Company's failure to have Defensible Title with respect to any Asset.
- (f) <u>Title Defect Amount</u>. The amount by which the Allocated Value of the affected Title Defect Property is reduced as a result of the existence of such Title Defect shall be the "*Title Defect Amount*" and shall be determined in accordance with the following terms and conditions:
  - (i) if Buyer and SM Energy agree on the Title Defect Amount, then that amount shall be the Title Defect Amount;
  - (ii) if the Title Defect is an Encumbrance that is undisputed and liquidated in amount, then the Title Defect Amount shall be the amount necessary to be paid to remove the Title Defect from the Title Defect Property;
  - (iii) if the Title Defect represents a discrepancy between (A) the Net Revenue Interest for any Title Defect Property and (B) the Net Revenue Interest for such Title Defect Property stated in **Exhibit B**, and the Working Interest attributable to such Title Defect Property has been reduced proportionately, then the Title Defect Amount shall be the product of the Allocated Value of such Title Defect Property multiplied by a fraction, the numerator of which is the positive difference between such Net Revenue Interest values, and the denominator of which is the Net Revenue Interest for such Title Defect Property stated in **Exhibit B**; provided, that if the Net Revenue Interest discrepancy does not affect the Title Defect Property throughout its entire life, the Title Defect Amount determined under this Section 6.2(f) shall be reduced to take into account the applicable time period only; provided, further, that if the reduction in Net Revenue Interest resulting from a Title Defect is greater than fifteen percent (15%) of the Net Revenue Interest stated on **Exhibit B**, then the Title Defect Amount for such Title Defect shall not be calculated under this Section 6.2(f)(iii), but instead shall be calculated in accordance with the adjustment calculation provided in Section 6.2(f)(iv);
  - (iv) if the Title Defect represents an obligation or Encumbrance upon or other defect in title to the Title Defect Property of a type not described above, including a Title Defect that represents a discrepancy between (A) the Net Revenue Interest for any Title Defect Property and (B) the Net Revenue Interest for such Title Defect Property stated in **Exhibit B**, and the Working Interest attributable to such Title Defect Property has not been reduced proportionately, the applicable Title Defect Amount shall be determined by taking into account the Allocated Value of the Title Defect Property, the portion of the Title Defect Property affected by the Title Defect, the legal effect of the Title Defect, the potential economic effect of the Title Defect over the life of the Title Defect Property, the values placed upon the Title Defect by Buyer and SM Energy and such other reasonable factors as are necessary to make a proper evaluation; *provided, however*, that if such Title Defect is

reasonably capable of being cured, the Title Defect Amount shall not be greater than the Allocated Value of such Title Defect Property;

- (v) the Title Defect Amount with respect to a Title Defect Property shall be determined without duplication of any costs or losses included in any other Title Defect Amount hereunder; and
- (vi) notwithstanding anything to the contrary in this ARTICLE VI (except with respect to any Title Defect as specified in Section 6.2(f)(ii)), the aggregate Title Defect Amounts attributable to the effects of all Title Defects upon any single Title Defect Property shall not exceed the Allocated Value of such Title Defect Property.
- (g) <u>Title Benefit Amount</u>. The Title Benefit Amount resulting from a Title Benefit shall be determined in accordance with the following methodology, terms and conditions:
  - (i) if Buyer and SM Energy agree on the Title Benefit Amount, then that amount shall be the Title Benefit Amount;
  - (ii) if the Title Benefit represents an increase between (A) the Net Revenue Interest for any Title Benefit Property and (B) the Net Revenue Interest for such Title Benefit Property stated in **Exhibit B**, and the Working Interest has increased proportionately, then the Title Benefit Amount shall be the product of the Allocated Value of such Title Benefit Property multiplied by a fraction, the numerator of which is the positive difference between such Net Revenue Interest values, and the denominator of which is the Net Revenue Interest for such Title Benefit Property stated in **Exhibit B**; *provided*, that if the Net Revenue Interest increase does not affect the Title Benefit Property throughout its entire life, the Title Benefit Amount determined under this *Section 6.2(g)* shall be reduced to take into account the applicable time period only;
  - (iii) if the Title Benefit is of a type not described above, then the applicable Title Benefit Amount shall be determined by taking into account the Allocated Value of the Title Benefit Property, the portion of the Title Benefit Property affected by such Title Benefit, the legal effect of the Title Benefit, the potential economic effect of the Title Benefit over the life of such Title Benefit Property, the values placed upon the Title Benefit by Buyer and SM Energy and such other reasonable factors as are necessary to make a proper evaluation; and
  - (iv) the Title Benefit Amount with respect to a Title Benefit Property shall be determined without duplication of any items or amounts included in any other Title Benefit Amount hereunder.
- (h) <u>Title Threshold and Deductibles</u>. Notwithstanding anything to the contrary, in no event shall there be any adjustments to the Purchase Price or other remedies provided by SM Energy hereunder: (i) for any individual Title Defect for which the Title Defect Amount does not exceed the Individual Title Defect Threshold and (ii) for any Title Defect for which the Title Defect Amount exceeds the Individual Title Defect Threshold unless the sum of (A) the aggregate Title

Defect Amounts of all such Title Defects that exceed the Individual Title Defect Threshold (excluding any Title Defect Amounts attributable to Title Defects cured by SM Energy), plus (B) the aggregate Remediation Amounts of all Environmental Defects that exceed the Individual Environmental Threshold (excluding any Remediation Amounts attributable to Environmental Defects cured by SM Energy), minus (C) all Title Benefit Amounts, exceeds Twenty Million Dollars (\$20,000,000) (the "Aggregate Deductible"), after which point Buyer shall be entitled to adjustments to the Purchase Price only with respect to such Title Defects in excess of the Aggregate Deductible. For the avoidance of doubt, if SM Energy elects to cure any Title Defect pursuant to Section 6.2(c)(ii), the Title Defect Amount relating to such Title Defect will not be counted toward the Aggregate Deductible, unless SM Energy is unable to cure such Title Defect during the Cure Period, nevertheless all such amounts shall be deposited in the Escrow Account as set forth in Section 6.2(c)(ii).

(i) <u>Title Dispute Resolution</u>. The Parties agree to resolve disputes concerning the following matters pursuant to this Section 6.2(i): (1) the existence and scope of a Title Defect (including any cure thereof), Title Benefit, Title Defect Amount or Title Benefit Amount and (2) the adequacy of SM Energy's Title Defect curative materials and Buyer's reasonable satisfaction thereof (the "Title Disputed Matters"). The Parties agree to attempt to initially resolve all disputes through good faith negotiations. If the Parties cannot resolve disputes regarding items (1) and (2) on or before Closing, the Closing shall be delayed as to only the Assets subject to the Title Disputed Matters until the Parties finally resolve the dispute pursuant to this Section 6.2(i) (and in such event, (A) the Assets subject to such Title Disputed Matters shall be assigned to SM Energy, (B) the Purchase Price paid to SM Energy at Closing shall be reduced by an amount equal to the aggregate of the Allocated Values of such Assets assigned to SM Energy and (C) Buyer shall deliver to the Escrow Agent at Closing an amount equal to such reduction to be held by the Escrow Agent pending resolution of the Title Disputed Matters, such amounts deposited into the Escrow Account pursuant to (C) above shall be released to the applicable Party pursuant to the resolution of the applicable Title Disputed Matter, including as instructed by the Title Arbitrator, if applicable); provided, however, if either Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied due, in whole or in part, to Title Defects, then the Parties will resolve all Title Disputed Matters pursuant to this Section 6.2(i) prior to Closing. In the event that neither Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied, it is understood and agreed that the Parties shall proceed to Closing as contemplated herein as to all Assets not covered by a Title Disputed Matter. The Title Disputed Matters will be finally determined pursuant to this Section 6.2(i). There shall be a single arbitrator, who shall be an attorney with at least ten (10) years' experience in oil and gas title and transactional matters, as selected by mutual agreement of Buyer and SM Energy within fifteen (15) days after any Party invokes the provisions of this Section 6.2(i) to resolve such Dispute, and absent such agreement, by the Dallas office of the AAA (the "Title Arbitrator"). The Title Arbitrator, once appointed, shall have no ex parte communications with the Parties concerning the expert determination or the underlying dispute. All communications between any Party and the Title Arbitrator shall be conducted in writing, with copies sent simultaneously to the other Party in the same manner, or at a meeting to which all Parties have been invited and of which such Parties have been provided at least five (5) Business Days' notice. The arbitration proceeding shall be held in Dallas, Texas and shall be conducted in accordance with the AAA Rules to the extent such rules do not conflict with the terms of this Section 6.2(i). The Title Arbitrator's

determination shall be made within twenty (20) days after submission by the Parties of the matters in Dispute and shall be final and binding upon both Parties, without right of appeal. In making his determination, the Title Arbitrator shall be bound by the rules set forth in Section 6.2(f) and Section 6.2(g) and, subject to the foregoing, may consider such other matters as in the opinion of the Title Arbitrator are necessary to make a proper determination. The Title Arbitrator, however, may not award (a) Buyer a greater Title Defect Amount than the Title Defect Amount claimed by Buyer in the applicable Title Defect Notice (which such Title Defect Amount shall not exceed the Allocated Value of the applicable Title Defect Property) or (b) SM Energy a greater Title Benefit Amount than the Title Benefit Amount claimed by SM Energy in the applicable Title Benefit Notice. The Title Arbitrator shall act as an expert for the limited purpose of determining the specific disputed Title Defect (or cure thereof), Title Benefit, Title Defect Amount or Title Benefit Amount submitted by either Party and may not award damages, interest or penalties to either Party with respect to any Dispute. SM Energy and Buyer shall each bear its own legal fees and other costs of presenting its case to the Title Arbitrator. Any decision rendered by the Title Arbitrator pursuant hereto shall be final, conclusive and binding on SM Energy and Buyer and will be enforceable against any of the Parties in any court of competent jurisdiction. Each of SM Energy and Buyer shall bear one-half of the costs and expenses of the Title Arbitrator. To the extent that the award of the Title Arbitrator with respect to any Title Defect Amount was not taken into account as an adjustment to the Purchase Price or the aggregate Title Defect Amounts, as applicable at Closing pursuant to Section 2.4 and an adjustment would otherwise be required under the provisions of Section 6.2(c) or Section 6.2(d), as applicable, then, within ten (10) days after the Title Arbitrator delivers written notice to Buyer and SM Energy of its award with respect to such Title Defect Amount or a Title Benefit Amount and subject to Section 6.2(h), the Purchase Price will be adjusted pursuant to Section 2.4 by the amount so awarded by the Title Arbitrator and any Assets subject to Title Disputed Matters assigned to SM Energy pursuant to this Section 6.2(i) shall be conveyed to the Company pursuant to a mutually agreed upon form of assignment and bill of sale.

## 6.3 Casualty or Condemnation Loss.

- (a) Notwithstanding anything herein to the contrary, from and after the Effective Time, if Closing occurs, Buyer shall assume all risk of loss with respect to (i) production of Hydrocarbons from the Assets through normal depletion (including watering out of any well, collapsed casing or sand infiltration of any well) and (ii) the depreciation of personal property due to ordinary wear and tear and, in each case, Buyer shall not assert such matters as Casualty Losses or Title Defects hereunder.
- (b) If, from and after the Effective Time but prior to the Closing Date, any portion of the Assets is damaged or destroyed by fire or other casualty or is taken in condemnation or under right of eminent domain (a "Casualty Loss"), and the resulting loss (net to the interest of SM Energy, the Company or its or their Affiliates in the assets damaged or destroyed) from such Casualty Loss exceeds \$100,000 based on the Allocated Value of the Assets physically impacted by such Casualty Loss and not merely associated with the physically impacted Assets, then (i) Buyer shall nevertheless be required to close the transactions contemplated by this Agreement and (ii) SM Energy shall elect by written notice to Buyer prior to Closing to either (A) cause, at SM Energy's sole cost and as promptly as reasonably practicable prior to Closing, each Asset affected by such Casualty Loss to

be repaired or restored to at least its condition prior to such casualty or taking, or (B) reduce the Purchase Price by the cost to repair or restore each Asset affected by such Casualty Loss to at least its condition prior to such casualty or taking. In each case, SM Energy shall retain all rights to insurance, condemnation awards and other claims against Third Parties with respect to the casualty or taking except to the extent the Parties otherwise agree in writing.

## 6.4 Preferential Rights and Consents to Assign.

- (a) All Consents and Preferential Rights are listed on *Schedule 6.4*. Except for Buyer's remedy for breach of SM Energy's representations contained in *Section 3.29*, the remedies set forth in this *Section 6.4* are the exclusive remedies under this Agreement related to the Consents and Preferential Rights.
- (b) From and after the Execution Date up to Closing, SM Energy and Buyer shall use their commercially reasonable efforts to obtain all Consents and waivers of all Preferential Rights (excluding any Customary Post-Closing Consents); *provided, however*, that neither Party shall be required to incur any Liability or pay any money in order to obtain such Consents and waivers.
- (c) <u>Preferential Rights</u>. SM Energy shall, within ten (10) days after the Execution Date (or ten (10) days after discovery thereof if discovered after the Execution Date), send to each holder of a Preferential Right a notice requesting the election or waiver by each such holder of its applicable Preferential Right, in each case in material compliance with the contractual provisions applicable to such Preferential Right, requesting a waiver of such right. Any Preferential Right must be exercised subject to all terms and conditions set forth in this Agreement, including the successful Closing of this Agreement pursuant to *ARTICLE XI*. The consideration payable under this Agreement for any particular Asset for purposes of Preferential Right notices shall be the Allocated Value of such Asset.
  - (i) All Assets burdened by Preferential Rights for which (A) the applicable Preferential Right has been waived, or (B) the period to exercise such Preferential Right has expired prior to the Closing without the applicable holder of such Preferential Right electing to enforce its Preferential Right, shall, in each case, be assigned to Buyer at the Closing pursuant to the provisions of this Agreement.
  - (ii) If, prior to the Closing (A) any holder of a Preferential Right notifies SM Energy that it intends to consummate the purchase of the portion of the Assets (including any Contribution Asset excluded from the Contribution due to such Preferential Right) (all such Assets and Contribution Assets collectively, the "Affected Assets") to which its Preferential Right applies or (B) the time for exercising a Preferential Right has not expired and the holder of such Preferential Right has not waived such Preferential Right, then, in each case, such Affected Assets shall be excluded from the Assets to be conveyed to Buyer at Closing and assigned by the Company back to SM Energy and the Purchase Price shall be reduced by the Allocated Value of such excluded Affected Assets. SM Energy shall be entitled to all proceeds paid by a Person exercising a Preferential Right prior to the Closing. If, within eighty (80) days following the Closing (1) such holder of such Preferential Right

thereafter fails to consummate the purchase of the Affected Assets or (2) the time for exercising such Preferential Right expires without exercise by the holder thereof, then SM Energy shall (x) so notify Buyer and (y) on or before ten (10) days following delivery of such notice, assign such Affected Assets to Buyer pursuant to a mutually agreed upon form of assignment and bill of sale and the Purchase Price shall be increased by an amount equal to the Allocated Value of such Affected Assets.

- (d) <u>Consents</u>. SM Energy, within ten (10) days after the Execution Date (or ten (10) days after discovery thereof if discovered after the Execution Date), shall send to each holder of a Consent a notice, substantially in the form attached hereto as Exhibit K, seeking such holder's consent to the transactions contemplated hereby; *provided, however*, SM Energy shall not be required to send any notices related to the Contribution to the extent that SM Energy sent such notice prior to the Execution Date.
  - (i) If (A) SM Energy fails to obtain a Consent prior to Closing and the failure to obtain such Consent would cause (1) the assignment to Buyer of any portion of the Assets to be void, (2) the termination of a Lease under the express terms thereof or (3) liquidated damages in excess of \$50,000 or (B) a Consent requested by SM Energy is denied in writing (each such Consent, a "Hard Consent"), then, in each case, prior to Closing, that portion of the Assets affected by such Hard Consent shall be assigned to SM Energy and excluded from the Assets and the Purchase Price shall be reduced by the Allocated Value of such portion of the Assets. In the event that, within eighty (80) days following Closing, a Hard Consent that was not obtained prior to Closing is obtained following Closing, or the requirement to obtain such Hard Consent is waived by Buyer then, within ten (10) days after such Hard Consent is obtained or the Company pursuant to a form of assignment that is mutually agreeable to SM Energy shall assign such excluded portion of the Assets to the Consent is waived by Buyer, Buyer shall have no claim against, and SM Energy shall have no Liability for, the failure to obtain such Hard Consent), and (y) Buyer shall pay to SM Energy by wire transfer of immediately available funds an amount equal to the Allocated Value of such portion of the Assets so assigned.
  - (ii) Notwithstanding anything contained herein to the contrary, in cases where a Hard Consent has not been obtained as of the Closing with respect to a Contract (including any contract excluded from the Contribution due to such Hard Consent) (all such Contracts and contracts collectively, the "Affected Contracts"), then the following shall apply: (A) such Affected Contract, to the extent not excluded from the Contribution, shall be assigned to SM Energy and SM Energy shall continue after Closing to use commercially reasonable efforts to obtain the Hard Consent so that such Affected Contract can be transferred to the Company upon receipt of the Hard Consent, (B) such Affected Contract shall be held by SM Energy for the benefit of Buyer, (C) the Assets to which such Affected Contract relates shall be assigned back to SM Energy and the Purchase Price shall be reduced by the Allocated Value of any such excluded Assets, and (D) once the Hard Consent is obtained, both the Affected Contract and the Assets to which such Contract relates shall be assigned to the Company and any Purchase Price reduction pursuant to subclause (C) above

shall be paid to SM Energy upon such assignment. With respect to any Affected Contract for which the applicable Hard Consent for the assignment or transfer to the Company is obtained following the Closing, SM Energy shall transfer such Affected Contract to the Company by execution and delivery of an instrument of conveyance in the form of the Assignment. In the event any Hard Consent is not obtained within the date that is one hundred eighty (180) days after Closing, any Asset that has been assigned back to SM Energy pursuant to subclause (C) above shall no longer be held for the benefit of Buyer and each Party shall repay to the other Party any amounts previously paid hereunder in respect of such Asset and such Asset shall be deemed to be an "Excluded Asset".

## ARTICLE VII ENVIRONMENTAL MATTERS

## 7.1 Environmental Defects.

- (a) Environmental Defects Notice. On or before the Defect Claim Date, Buyer shall have the right, but not the obligation, to deliver notices to SM Energy meeting the requirements of this Section 7.1(a) (each, an "Environmental Defect Notice") setting forth any matters that, in Buyer's reasonable opinion, constitute Environmental Defects and that Buyer asserts as Environmental Defects pursuant to this Section 7.1. For all purposes of this Agreement, Buyer shall be deemed to have waived any Environmental Defect that Buyer fails to properly assert as an Environmental Defect pursuant to an Environmental Defect Notice delivered in accordance with this Section 7.1(a) and received by SM Energy on or before the Defect Claim Date. To be effective, each Environmental Defect Notice shall be in writing and shall include: (i) a description of the Environmental Condition constituting the alleged Environmental Defect, (ii) each Asset (or portion thereof) affected by the alleged Environmental Defect (the "Environmental Defect Property"), (iii) the Allocated Value of each Environmental Defect Property (to the extent applicable), (iv) supporting documents reasonably necessary for SM Energy to identify the existence of the alleged Environmental Defect, and (v) a calculation of the Remediation Amount (itemized in reasonable detail) that Buyer asserts is attributable to such alleged Environmental Defect. SM Energy shall have the right, but not the obligation, to cure any claimed Environmental Defect to Buyer's reasonable satisfaction on or before Closing. To give SM Energy an opportunity to commence reviewing and curing Environmental Defects, Buyer agrees to use reasonable efforts to give SM Energy, on or before the end of each two-week period prior to the Defect Claim Date, written notice of all alleged Environmental Defects discovered by Buyer during the preceding two-week period, which notice (in any format) may be preliminary in nature and supplemented prior to the expiration of the applicable Defect Claim Date; provided, that the failure of Buyer to provide such preliminary notice of any Environmental Defects shall not be deemed to waive or otherwise prejudice Buyer's right to assert an Environmental Defect on or before the Defect Claim Date.
- (b) <u>Remedies for Environmental Defects</u>. Subject to SM Energy's continuing right to dispute the existence of an Environmental Defect or the Remediation Amount asserted with respect thereto and subject to the Individual Environmental Threshold and the Aggregate Deductible, in the event that any Environmental Defect timely asserted by Buyer in accordance with *Section 7.1*

- (a) is not waived in writing by Buyer or cured to Buyer's satisfaction on or before Closing, SM Energy shall, at its sole option, elect to:
  - (i) reduce the Purchase Price by the Remediation Amount for such Environmental Defect; or
  - (ii) to the extent Buyer agrees in writing, indemnify Buyer against all Liability resulting from such Environmental Defect with respect to the Assets pursuant to an indemnity agreement in a form mutually agreeable to the Parties.

If SM Energy elects the option set forth in clause (i) above, Buyer and the Company shall be deemed to have assumed responsibility for all costs and expenses attributable to the Remediation of the applicable Environmental Defect (net to the interest of SM Energy, the Company or its or their Affiliates) and all Liabilities (net to the interest of SM Energy, the Company or its or their Affiliates) with respect thereto, and Buyer's and the Company's obligations with respect to the foregoing shall be deemed to constitute part of the Assumed Obligations.

- (c) <u>Exclusive Remedy</u>. Subject to Buyer's remedy for a breach of SM Energy's representation contained in *Section 3.17*, *Section 7.1(b)* shall be the exclusive right and remedy of Buyer with respect to any Environmental Defect.
- (d) Environmental Deductibles. Notwithstanding anything to the contrary, except as set forth in this Section 7.1(d) below, there shall be no adjustments to the Purchase Price or other remedies provided by SM Energy for (i) any individual Environmental Defect for which the Remediation Amount does not exceed the Individual Environmental Threshold (or, with respect to any individual Asset, Environmental Defects in the aggregate), or (ii) any Environmental Defect (or, with respect to any individual Asset, Environmental Defects in the aggregate for which the Remediation Amount exceeds the Individual Environmental Threshold, unless the sum of (A) the aggregate Remediation Amounts of all such Environmental Defects that exceed the Individual Environmental Threshold (excluding any Remediation Amounts attributable to Environmental Defects cured by SM Energy), plus (B) the aggregate Title Defect Amounts of all Title Defects that exceed the Individual Title Defect Threshold (excluding any Title Defect Amounts attributable to Title Defects cured by SM Energy), minus (C) all Title Benefit Amounts, exceeds the Aggregate Deductible, after which point Buyer shall only be entitled to adjustments to the Purchase Price with respect to such Environmental Defects in excess of the Aggregate Deductible; provided, however, that in the event the Remediation Amount with respect to any Environmental Defect equals or exceeds Three Million Dollars (\$3,000,000) the Purchase Price shall be reduced by the full amount of the applicable Remediation Amount without regard to the Aggregate Deductible and any reductions pursuant to this proviso shall not be used in calculating whether the Aggregate Deductible has been met with respect to any other Environmental Defects.
- (e) <u>Environmental Dispute Resolution</u>. The Parties agree to resolve disputes concerning the existence and scope of an Environmental Defect or Remediation Amount pursuant to this *Section 7.1(e)* (the "*Environmental Disputed Matters*"). The Parties agree to attempt to initially resolve all disputes through good faith negotiations. If the Parties cannot resolve disputes regarding Environmental Disputed Matters on or before Closing, the Closing shall be delayed as

to only the Assets subject to the Environmental Disputed Matters until the Parties finally resolve the dispute pursuant to this Section 7.1(e) (and in such event, the Assets subject to such Environmental Disputed Matters shall be assigned to SM Energy pending resolution of the Environmental Disputed Matters) and the Purchase Price shall be reduced by the Allocated Value of such Assets; provided, however, if either Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied due, in whole or in part, to Environmental Defects, then the Parties will resolve all Environmental Disputed Matters pursuant to this Section 7.1(e) prior to Closing. In the event that neither Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied, it is understood and agreed that the Parties shall proceed to Closing as contemplated herein. The Environmental Disputed Matters will be finally determined pursuant to this Section 7.1(e). There will be a single arbitrator, who must be an environmental attorney with at least 10 years' experience in environmental matters involving oil and gas properties, as selected by mutual agreement of Buyer and SM Energy within fifteen (15) days after any Party invokes the provisions of this Section 7.1(e) to resolve such Dispute, and absent such agreement, by the Dallas office of the AAA (the "Environmental Arbitrator"). The Environmental Arbitrator, once appointed, shall have no exparte communications with the Parties concerning the expert determination or the underlying dispute. All communications between any Party and the Environmental Arbitrator shall be conducted in writing, with copies sent simultaneously to the other Party in the same manner, or at a meeting to which all Parties have been invited and of which such Parties have been provided at least five (5) Business Days' notice. The arbitration proceeding will be held in Dallas, Texas and conducted in accordance with the AAA Rules to the extent such rules do not conflict with the terms of this Section 7.1(e). The Environmental Arbitrator's determination must be made within twenty (20) days after submission of the matters in Dispute and shall be final and binding upon both Parties, without right of appeal. In making its determination, the Environmental Arbitrator shall be bound by the rules set forth in this Section 7.1 and, subject to the foregoing, may consider such other matters as in the opinion of the Environmental Arbitrator are necessary or helpful to make a proper determination. The Environmental Arbitrator, however, may not award Buyer a greater Remediation Amount than the Remediation Amount claimed by Buyer in the applicable Environmental Defect Notice (which such award shall not exceed the Allocated Value of the applicable Environmental Defect Property). The Environmental Arbitrator will act as an expert for the limited purpose of determining the specific disputed Environmental Defects or Remediation Amounts submitted by either Party and may not award damages, interest or penalties to either Party with respect to any matter. SM Energy and Buyer will each bear its own legal fees and other costs of presenting its case. Any decision rendered by the Environmental Arbitrator pursuant hereto shall be final, conclusive and binding on SM Energy and Buyer and will be enforceable against any of the Parties in any court of competent jurisdiction. Each Party will bear one-half of the costs and expenses of the Environmental Arbitrator. To the extent that the award of the Environmental Arbitrator with respect to any Remediation Amount is not taken into account as an adjustment to the Purchase Price at Closing pursuant to Section 2.4 and Buyer would otherwise be entitled to an adjustment under the provisions of Section 7.1(d), then, within ten (10) days after the Environmental Arbitrator delivers written notice to Buyer and SM Energy of such award and subject to Section 7.1(d), the Purchase Price will be adjusted pursuant to Section 2.4 by such Remediation Amount.

7.2 NORM. Buyer acknowledges that the Assets have been used for exploration, development and production of oil and gas and that there may be NORM located in, on or under or associated with the Assets. Equipment and sites included in the Assets may contain asbestos or NORM. NORM may affix or attach itself to the inside of wells, materials and equipment as scale, or in other forms. The wells, materials and equipment located on the Assets or included in the Assets may contain NORM. NORM containing material may have come in contact with various environmental media, including without limitation, water, soils or sediment. Special procedures may be required for the assessment, remediation, removal, transportation or disposal of asbestos and NORM from the Assets.

## ARTICLE VIII CERTAIN AGREEMENTS

- **8.1** Conduct of Business. Except (x) as set forth in Schedule 8.1, (y) as expressly contemplated by this Agreement or (z) as expressly consented to in writing by Buyer, SM Energy agrees that from and after the Execution Date up to Closing:
  - (a) SM Energy will, and will cause its Affiliates to:
    - (i) conduct the business of the Company in the ordinary and usual course consistent with past practice;
  - (ii) use its commercially reasonable efforts to cause the Assets to be maintained and operated in the usual, regular and ordinary manner consistent with past practice;
  - (iii) use commercially reasonable efforts to pay or cause to be paid all Burdens, Asset Taxes and Operating Expenses, and other payments incurred with respect to the Assets consistent with past practice, except expenses or Burdens being contested in good faith, unless the nonpayment of such contested expenses or royalties could result in the termination of a Lease, in which case SM Energy will notify Buyer and obtain Buyer's approval prior to withholding such payment;
    - (iv) maintain insurance coverage on the Assets in the amounts and types currently in force;
  - (v) use commercially reasonable efforts consistent with past practice to cause to be maintained in full force and effect all Leases, except where any such Lease terminates pursuant to its existing terms or where a reasonable and prudent operator would not maintain the same;
  - (vi) notify Buyer of any election that SM Energy, the Company or its or their Affiliates is required or has the right to make under any joint operating agreement, marketing or purchase contract, area of mutual interest agreement or farmout agreement, specifying the nature and time period associated with such election;

- (vii) give prompt notice (and in any event within one (1) Business Day of receipt of written notice from any Third Party) to Buyer of any emergency requiring immediate action, or any emergency action taken, in the face of serious risk to life, property or the environment, (including prevention of environmental contamination);
- (viii) use commercially reasonable efforts to preserve or cause to be preserved relationships with Third Parties having business dealings relating to the Assets;
- (ix) notify Buyer of any operation proposed by a Third Party that is reasonably estimated to cost SM Energy, the Company or its or their Affiliates in excess of \$150,000;
- (x) maintain the books of account and records relating to the Assets in the usual, regular and ordinary manner, in accordance with its usual accounting practices; and
- (xi) give written notice to Buyer as soon as is practicable of any written notice received or given by SM Energy, the Company or its or their Affiliates with respect to any alleged breach by SM Energy, the Company or its or their Affiliates of any Lease or Material Contract;
  - (b) SM Energy will not, and will cause its Affiliates (including the Company) not to:
- (i) adopt any amendments to, or give any waivers under, the Company Organizational Documents or the Contribution Documents;
- (ii) declare, set aside, make or pay any dividend or other distribution in respect of the Interests or any other Membership Interests in the Company;
  - (iii) repurchase, redeem or otherwise acquire the Interests or any other Membership Interests in the Company;
- (iv) transfer, sell, pledge, encumber or dispose of the Interests or any other Membership Interests in the Company held by SM Energy;
- (v) issue, deliver, sell or pledge, or authorize the issuance, delivery, sale or pledge of Membership Interests in the Company or grant options, warrants calls or other rights to purchase or otherwise acquire Membership Interests in the Company, whether to SM Energy or any of its Affiliates, any Affiliates of the Company or any Third Party;
- (vi) split, combine or reclassify any Membership Interests in the Company, or issue any other securities in respect of, in lieu of or in substitution for, any Membership Interests in the Company;
- (vii) except for (A) emergency operations, (B) operations required under presently existing AFEs described on *Schedule 3.16*, (C) operations undertaken to avoid any

penalty provision of any Contract or Governmental Authority order, and (D) operations undertaken pursuant to an approved annual budget for the Springfield gathering systems, agree to, or propose any operations on the Assets anticipated to cost (net to the interest of SM Energy, the Company or its or their Affiliates) in excess of \$150,000 per operation; *provided*, that with respect to emergency operations, SM Energy shall notify Buyer of such emergency as soon as reasonably practicable following receipt of written notice from any Third Party Operator;

- (viii) enter into an Contract that, if entered into prior to the Execution Date, would be required to be listed in Schedule 3.12(a),
- (ix) terminate (unless such Material Contract terminates pursuant to its stated terms) or amend the terms of any Material Contract;
- (x) hire any Person as an employee, consultant or independent contractor of the Company, or enter into any employment, deferred compensation, severance, special pay, consulting, non-competition or similar agreement or arrangement with any Person to which the Company is party;
- (xi) make, change or revoke any material Tax election or materially change financial accounting policies or procedures, except as required by GAAP or applicable law;
- (xii) enter into or agree to enter into any merger or consolidation with any Person, and not engage in any new business or invest in, make a loan, advance or capital contribution to, or otherwise acquire the securities or substantially all of the assets or business of, any other Person;
- (xiii) (A) except for the matters disclosed on *Schedule 8.1*, with respect to the Company, issue, create, incur, assume, guarantee, endorse or otherwise become liable or responsible with respect to (whether directly, contingently or otherwise) any indebtedness, (B) except in the ordinary course of business consistent with past practice (*provided* that such exception shall apply only to periods prior to the Effective Date) or as specifically provided in this Agreement, pay, repay, discharge, purchase, repurchase or satisfy any indebtedness of the Company out of Company funds, (C) modify the terms of any indebtedness of the Company, or (D) except as specifically provided in this Agreement, modify the terms of any Liability that will remain in effect as an obligation of the Company after the Closing;
- (xiv) cancel or compromise any debt or claim held by the Company, or waive or release any right of the Company except in the ordinary course of business consistent with past practice (*provided* that such exception shall apply only to periods prior to the Effective Date);
- (xv) enter into any contract, understanding or commitment that restrains, restricts, limits or impedes the ability of the Company to compete with or conduct any

business or line of business in any geographic area or solicit the employment of any Person (including, but not limited to, agreements with "area of mutual interest" provisions);

- (xvi) except for the matters disclosed on *Schedule 3.10*, settle any suit or litigation or waive any claims or rights in each case attributable to the Assets and effecting the period after the Effective Time;
  - (xvii) become a non-consenting party to any operation proposed by a Third Party;
- (xviii) except for AFEs included on *Schedule 3.16*, approve any individual AFE or similar request or invoice for funding or participation under any Contract (other than those required under the terms of any Contract) or otherwise which would reasonably be estimated to require expenditures in excess of \$150,000 (net to the interest of SM Energy, the Company or its or their Affiliates);
- (xix) grant or create any Preferential Rights or transfer restriction with respect to the Assets except in connection with the performance by SM Energy of any obligation or agreement existing on the Execution Date or pursuant to this Agreement;
- (xx) transfer, sell, mortgage, pledge or dispose of any material portion of the Assets other than the sale or disposal of Hydrocarbons in the ordinary course of business and sales of equipment that is no longer necessary in the operation of the Assets and for which replacement equipment of equal or greater value has been obtained;
- (xxi) voluntarily abandon any of the Leases or Wells other than as required pursuant to the terms of a Lease or applicable Law;
- (xxii) voluntarily waive or release any material right that could reasonably be expected to adversely affect Buyer's ownership or use of, and the ability to operate, the Assets after Closing;
- (xxiii) make any regulatory or other filings of any kind with any Governmental Authority that could reasonably be expected to adversely affect Buyer's ownership or use of the Assets after Closing; or
  - (xxiv) commit to do any of the foregoing.

Buyer acknowledges that the Company owns an undivided interest in the Assets with respect to which it is not the Operator, and Buyer agrees that the acts or omissions of the other Working Interests owners (including the Third Party Operators) who are not SM Energy or the Company shall not constitute a breach of the provisions of this *Section 8.1*, and no action required pursuant to a vote of Working Interest owners shall constitute a breach of the provisions of this *Section 8.1* so long as SM Energy voted its interest in such a manner that complies with the provisions of this *Section 8.1*.

- **8.2** Governmental Bonds. Buyer acknowledges that none of the bonds, letters of credit and guarantees, if any, posted by SM Energy or its Affiliates (excluding the Company) with Governmental Authorities and relating to the Assets and set forth on Schedule 3.28 are transferable to Buyer. At or prior to Closing, Buyer shall deliver to SM Energy evidence of the posting of bonds or other security with all applicable Governmental Authorities meeting the requirements of such Governmental Authorities to own the Assets.
- 8.3 HSR Act. If applicable, within ten (10) Business Days following the execution by Buyer and SM Energy of this Agreement, Buyer and SM Energy will each prepare and file with the DOJ and the FTC the notification and report form required for the transactions contemplated by this Agreement by the HSR Act and request early termination of the waiting period thereunder. Buyer and SM Energy agree to respond promptly to any inquiries or requests for information or documentary material from the DOJ or the FTC concerning such filings and to comply in all material respects with the filing requirements of the HSR Act. Buyer and SM Energy shall cooperate with each other and, subject to the terms of the Confidentiality Agreement, shall promptly furnish all information to the other Party that is necessary in connection with Buyer's and SM Energy's compliance with the HSR Act. Buyer and SM Energy shall keep each other fully advised with respect to any requests from or communications with the DOJ or FTC concerning such filings and shall consult with each other with respect to all responses thereto. Each of SM Energy and Buyer shall use its commercially reasonable efforts to take all actions reasonably necessary and appropriate in connection with any HSR Act filing to satisfy the conditions to closing and consummate the transactions contemplated hereby as promptly as practicable and in any event not later than the Outside Termination Date; provided, however, nothing in this Agreement shall require Buyer or SM Energy to propose, negotiate, effect or agree to, the sale, divestiture, license or other disposition of any assets or businesses of Buyer or SM Energy (including the Assets) or otherwise take any action that limits the freedom of action with respect to, or its ability to retain or operate any of the businesses of the Buyer or SM Energy or the Assets. The filing fees associated with any such HSR Act filing shall be borne one-half by Buyer and one-half by SM Energy. Notwithstanding any provision of this Section 8.3, no Party shall be required to provide the other Party with information regarding the value of the transaction, in order to comply with contractual arrangements or applicable Laws, or subject to the attorney client privilege, work product doctrine or other similar privilege.

### 8.4 Financial Statements and Cooperation.

(a) SM Energy acknowledges that Buyer or its Affiliates may be required to include statements of revenues and direct operating expenses and other financial information relating to the Assets in documents filed by Buyer or its Affiliates or assignee with the SEC pursuant to the Securities Act, or in other materials (including offering materials for securities offered and sold as eligible for resale under Rule 144A of the Securities Act), and that such financial statements may be required to be independently audited (together with any supplementary oil and gas information required by ASC 932-235, the "*Requisite Financial Statement Information*"). At Buyer's request, from and after the Execution Date and for up to three years after Closing, SM Energy shall, at Buyer's sole cost and expense, provide Buyer (or its applicable Affiliate or assignee) and its auditor reasonable access during normal business hours to such records (to the extent that such information is available) and personnel of SM Energy and its Affiliates (the "*Seller Group*") and use reasonable

efforts to, at Buyer's sole cost and expense, provide reasonable access during normal business hours to Seller Group's accounting firm, in each case, as Buyer (or its applicable Affiliate or assignee) may reasonably request to enable Buyer (or its applicable Affiliate or assignee) and its auditor, and its representatives, for the sole purpose of creating and, to the extent required by Law, auditing, the Requisite Financial Statement Information. Notwithstanding anything to the contrary, (i) SM Energy shall in no event be required to create new records relating to the Assets or financial statements and (ii) the access to be provided to Buyer shall not interfere with SM Energy's ability to prepare its own financial statements or its regular conduct of business and shall be made available during SM Energy's normal business hours.

- (b) SM Energy agrees to provide its written consent for the inclusion or incorporation by reference of the Requisite Financial Statement Information in any filing with the SEC of Buyer or any of its Affiliates in which the Requisite Financial Statement Information is required to be included or incorporated by reference to satisfy any rule or regulation of the SEC or to satisfy relevant disclosure obligations under the Securities Act or the Securities Exchange Act of 1934, as amended. Upon request of Buyer, SM Energy shall request the external audit firm that audits the Seller Group's financial statements (the "Seller Audit Firm") to consent to the inclusion or incorporation by reference of its audit opinion with respect to the audited financial statements in any such registration statement, report or other document and SM Energy shall use its commercially reasonable efforts to, at Buyer's sole cost and expense, provide Buyer and Buyer's independent accountants with access to (i) audit work papers of the Seller Group's independent accountants and (ii) management representation letters delivered by any member of the Seller Group, as applicable, or on behalf of Seller Group to the Seller Group's independent accountants.
- (c) All of the information provided by SM Energy pursuant to this *Section 8.4* is given without any representation or warranty, express or implied, and no member of the SM Energy Indemnified Parties or Seller Audit Firm shall have any liability or responsibility with respect thereto. Buyer shall reimburse the Seller Group for all reasonable and documented out-of-pocket costs and expenses incurred by the Seller Group in complying with this *Section 8.4*.
- 8.5 Names. As soon as reasonably possible after final resolution of all Title Disputed Matters, Buyer shall change the name of the Company or merge the Company into Buyer so as to remove the names and other identifying marks of SM Energy and its Affiliates, and (b) ninety (90) days after Closing, Buyer shall remove or caused to be removed, the names and other identifying marks of SM Energy and its Affiliates, if any, from the Assets. As soon as reasonably possible after the Closing, Buyer shall make the requisite filings with, and provide the requisite notices to, the appropriate Governmental Authorities to accomplish the forgoing and to place the title or other indicia of ownership of the Assets in a name other than any name of SM Energy or any of its Affiliates, or any variations thereof.

#### 8.6 Hedging Matters.

(a) On January 3, 2017 or such later date as directed by email by Sponsor, but in no event later than January 15, 2017, SM Energy shall in consultation with and at the direction of Buyer enter into one or more NYMEX swap contracts (each, a "*Hedge Contract*") intended to reduce or eliminate the risk of fluctuation in the price of Hydrocarbons covering the production set

forth on *Schedule 8.6(a)* through 2021 (the "*Hedging Volumes*") with any of JP Morgan Chase Bank, N.A., Wells Fargo Bank N.A. or Bank of America Merrill Lynch or their respective Affiliates (collectively, the "*Counterparties*"). SM Energy hereby agrees to enter into novations of such Hedge Contracts, and Buyer or the Company hereby agrees to accept such novations upon Closing and will assume all Liabilities and retain all economic value related or attributable to such Hedge Contract (including, for the avoidance of doubt, any such Liabilities or gains attributable to the period on or before the Closing), pursuant to documentation required by the Counterparties. Upon Buyer's request, SM Energy shall provide pricing information for any requested Hedge Contract. If this Agreement is terminated for any reason, SM Energy shall determine whether or not, and if so, when, to terminate any and all Hedge Contracts in its sole discretion.

- (b) If this Agreement is terminated prior to Closing the following shall apply:
- (i) Within three (3) Business Days after the termination of this Agreement, Buyer and SM Energy shall endeavor in good faith to agree on the net aggregate value of all Hedge Contracts to SM Energy as of the termination date. If Buyer and SM Energy are unable to so agree, then within two (2) Business Days thereafter SM Energy shall, with respect to each Hedge Contract, obtain a written quote from the applicable Counterparty for the amount that such Counterparty would pay (expressed as a positive number) or require to be paid (expressed as a negative number) in order to terminate such Hedge Contract as of the date such quotation is provided (including, for the avoidance of doubt, any such losses or gains attributable to the period on or before Closing) and promptly provide a copy thereof to Buyer. The net aggregate value agreed between Buyer and SM Energy or the sum of all such quotations, as applicable, shall be referred to herein as the "Aggregate Net Hedge Value".
- (ii) If this Agreement is terminated by SM Energy pursuant to Section 12.1(a) because the conditions set forth in Sections 9.2(a) and 9.2(b) have not been satisfied as of the Outside Termination Date, then (A) if the Aggregate Net Hedge Value is positive, no payment shall be owing by any Party to the other Party with respect to the Hedge Contracts and (B) if the Aggregate Net Hedge Value is negative, Buyer shall pay the absolute value of such amount to SM Energy within five (5) Business Days of receipt of such quotations.
- (iii) If this Agreement is terminated by Buyer pursuant to Section 12.1(b) because the conditions set forth in Sections 9.1(a) and 9.1(b) have not been satisfied as of the Outside Termination Date, then (A) if the Aggregate Net Hedge Value is positive, SM Energy shall pay the absolute value of such amount to Buyer within five (5) Business Days of receipt of such quotations and (B) if the Aggregate Net Hedge Value is negative, no payment shall be owing by any Party to the other Party with respect to the Hedge Contracts.
- (iv) If this Agreement is terminated pursuant to Section 12.1(e) because the conditions set forth in Section 9.1(f) have not been satisfied as of the Closing Date because a Specified Required Consent has not been obtained, then the Hedge Contracts shall remain with SM Energy and Buyer shall have no liability to SM Energy or the Company

with respect to any Hedge Contracts regardless of the Aggregate Net Hedge Value at such time.

- (v) If this Agreement is terminated other than pursuant to Section 12.1(a) or Section 12.1(b) because the conditions set forth in Sections 9.1(a) and 9.1(b), or 9.2(a) and 9.2(b), as applicable, have not been satisfied as of the Outside Termination Date, or pursuant to Section 12.2(e) because the conditions set forth in Section 9.1(f) have not been satisfied as of the Closing Date because any Required Consent that is a Specified Required Consent has not been obtained, then (A) if the Aggregate Net Hedge Value is positive, SM Energy shall pay such amount to Buyer within five (5) Business Days of receipt of such quotations and (B) if the Aggregate Net Hedge Value is negative, Buyer shall pay such amount to SM Energy within five (5) Business Days of receipt of such quotations.
- (c) If Closing occurs and any Counterparty refuses to novate any Hedge Contract to Buyer or the Company in connection with Closing, then SM Energy shall promptly (and in any event within one (1) Business day of receipt of notice from Buyer) unwind the Hedge Contracts and (A) if the Aggregate Net Hedge Value is positive, SM Energy shall pay such amount to Buyer within five (5) Business Days of receipt of such quotations and (B) if the Aggregate Net Hedge Value is negative, Buyer shall pay such amount to SM Energy within five (5) Business Days of receipt of such quotations.
- 8.7 Financing Cooperation. For the period beginning on the Execution Date and ending on the Closing Date, the Seller Group agrees to use their reasonable best efforts to cooperate (and to cause their Representatives to cooperate) as reasonably requested by Buyer in connection with Buyer's efforts to satisfy the conditions precedent to the Financing by (a) furnishing the Buyer with financial and operational information regarding the Assets, including lease, well and production records and financial statements; (b) reasonably cooperating with the marketing and syndication efforts of the Buyer and the Financing Sources for any portion of the Financing; (c) providing Buyer, at least three (3) Business Days prior to the Closing Date, all documentation and other information required by regulatory authorities and as reasonably requested by Buyer on behalf of the Financing Sources with respect to the Assets in connection with the applicable "know your customer" and anti-money laundering rules and regulations, including the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law October 26, 2001), as amended, provided such request is made at least ten (10) Business Days prior to the Closing Date; and (d) furnishing to the Buyer and the Financing Sources, as promptly as reasonably practicable, such additional financial and other information as the Buyer shall reasonably request in order to consummate the Financing. Notwithstanding anything contained herein to the contrary, no such access or examination will be permitted to the extent that it would require the Seller Group to disclose information subject to attorney-client privilege or attorney work-product privilege, conflict with any Third Party confidentiality obligations to which the Seller Group is bound, or violate any applicable Law. It is understood and agreed by the Parties that the cooperation by the Seller Group described in this Section 8.7 shall not obligate the Seller Group to take any actions that materially adversely interfere with its ongoing busine

No member of the Seller Group shall have any liability or responsibility to (x) any Financing Source with respect to any of its obligations under this Section 8.7 or (y) Buyer with respect to the accuracy

or completeness of any information delivered pursuant to this *Section 8.7*. Buyer shall reimburse the Seller Group for all reasonable and documented out-of-pocket costs and expenses incurred by the Seller Group in complying with this *Section 8.7*.

**8.8** Equity Commitment. On or before 5:00 p.m. CST on January 2, 2016, Buyer will deliver to SM Energy an equity commitment letter in a form mutually acceptable to SM Energy and Buyer that provides a commitment in support of Buyer's obligations pursuant to Section 8.6.

## ARTICLE IX CONDITIONS TO CLOSING

- **9.1** Buyer's Conditions to Closing. The obligations of Buyer to consummate the transactions contemplated by this Agreement are subject to the fulfillment (or waiver by Buyer) on or prior to the Closing of each of the following conditions:
- (a) <u>Representations</u>. The representations and warranties of SM Energy set forth in *ARTICLE III* shall be true and correct in all respects as of the Closing Date as though made on and as of the Closing Date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except for those breaches, if any, of such representations and warranties that in the aggregate would not have a Material Adverse Effect; *provided*, that, notwithstanding anything to the contrary herein, any (i) breach that would (A) cause Buyer to incur any Liability or (B) decrease the value of the Assets, in each case in an aggregate amount equal to or greater than \$200,000,000 or (ii) material breach of the Fundamental Representations shall be deemed a "Material Adverse Effect".
- (b) <u>Performance</u>. SM Energy shall have performed or complied in all material respects with all obligations, agreements and covenants contained in this Agreement as to which performance or compliance by SM Energy is required prior to or at the Closing Date.
- (c) <u>No Legal Proceedings</u>. No material suit, action or other proceeding instituted by a Third Party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement. No order, award or judgment shall have been issued by any Governmental Authority or arbitrator to restrain, prohibit, enjoin, or declare illegal, or awarding substantial damages in connection with, the transactions contemplated by this Agreement.
- (d) <u>Title Defects; Environmental Defects; Preferential Rights; Consents; Casualty Losses</u>. The sum of (i) all (A) Title Defect Amounts determined under *Section 6.2(c)(i)*, *plus* (B) the amount deposited into the escrow account pursuant to *Section 6.2(c)(ii)*, *less* (C) all Title Benefit Amounts determined under *Section 6.2(g)*, *plus* (ii) all Remediation Amounts for Environmental Defects determined under *Section 7.1(b)(i)*, *plus* (iii) all adjustments to the Purchase Price made pursuant to *Section 6.4(c)(ii)* as a result of un-waived or unexpired Preferential Rights and *Section 6.4(d)* in respect of unobtained Hard Consents (other than Required Consents, which are addressed in *Section 9.1(f)* below), *plus* (iv) the amount of all Casualty Losses pursuant to *Section 6.3* shall, in the aggregate, be less than twenty percent (20%) of the unadjusted aggregate Purchase Price.

- (e) <u>HSR Act</u>. Any waiting period applicable to the consummation of the transactions contemplated under the terms of this Agreement under the HSR Act shall have expired or been terminated.
  - (f) Required Consents. SM Energy shall have obtained all Required Consents.
- (g) <u>Indebtedness</u>. The Company shall not have any outstanding Liabilities for Indebtedness, any Encumbrances binding the Company or any of the Assets related to any Indebtedness (other than Permitted Encumbrances or Encumbrances to be released at Closing), or any outstanding Indebtedness or other Liabilities (including trade payables incurred in the ordinary course of business) to SM Energy, its Affiliates or Affiliates of the Company.
  - (h) <u>Hedging</u>. SM Energy shall have performed, in all material respects, its obligations under *Section 8.6*.
- (i) <u>Closing Deliverables</u>. SM Energy shall have delivered (or be ready, willing and able to deliver at Closing) to Buyer the documents and other items required to be delivered by SM Energy under *Section 11.3*.
- 9.2 SM Energy's Conditions to Closing. The obligations of SM Energy to consummate the transactions contemplated by this Agreement are subject to the fulfillment (or waiver by SM Energy) on or prior to the Closing of each of the following conditions:
- (a) <u>Representations</u>. The representations and warranties of Buyer set forth in *ARTICLE IV* shall be true and correct in all material respects (other than those representations and warranties of Buyer that are qualified by materiality, which shall be true and correct in all respects) as of the Execution Date and as of the Closing Date as though made on and as of such date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date).
- (b) <u>Performance</u>. Buyer shall have performed or complied in all material respects with all obligations, agreements and covenants contained in this Agreement as to which performance or compliance by Buyer is required prior to or at the Closing Date.
- (c) No Legal Proceedings. No material suit, action or other proceeding instituted by a Third Party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement. No order, award or judgment shall have been issued by any Governmental Authority or arbitrator to restrain, prohibit, enjoin, or declare illegal, or awarding substantial damages in connection with, the transactions contemplated by this Agreement.
- (d) <u>Title Defects; Environmental Defects; Preferential Rights; Consents; Casualty Losses</u>. The sum of (i) all (A) Title Defect Amounts determined under *Section 6.2(c)(i)*, *plus* (B) the amount deposited into the escrow account pursuant to *Section 6.2(c)(ii)*, *less* (C) all Title Benefit Amounts determined under *Section 6.2(g)*, *plus* (ii) all Remediation Amounts for Environmental Defects determined under *Section 7.1(b)(i)*, *plus* (iii) all adjustments to the Purchase

Price made pursuant to Section 6.4(c)(ii) as a result of un-waived or unexpired Preferential Rights and Section 6.4(d) in respect of unobtained Hard Consents (other than Required Consents, which are addressed in Section 9.19.1(f) above), plus (iv) the amount of all Casualty Losses pursuant to Section 6.3 shall, in the aggregate, be less than twenty percent (20%) of the unadjusted aggregate Purchase Price.

- (e) <u>HSR Act</u>. Any waiting period applicable to the consummation of the transactions contemplated under the terms of this Agreement under the HSR Act shall have expired or been terminated.
- (f) <u>Closing Deliverables</u>. Buyer shall have delivered (or be ready, willing and able to deliver at Closing) to SM Energy the documents and other items required to be delivered by Buyer under *Section 11.3*.

## ARTICLE X TAX MATTERS

- 10.1 Tax Matters. Up to the Closing Date, SM Energy shall be responsible for filing with the applicable taxing authorities each Company Tax Return that is required to be filed on or before the Closing Date and paying any Company Taxes due and owing; provided, that to the extent such Taxes relate to the periods from and after the Effective Time, as determined pursuant to Section 10.1(c), such payment shall be on behalf of Buyer, and promptly following the Closing Date, Buyer shall pay to SM Energy any such Taxes; but only to the extent that such amounts have not already been accounted for under Section 2.4. Following the Closing Date, SM Energy shall be responsible for paying all additional Company Taxes to the extent such Taxes relate to the periods on or before the Effective Time, as determined pursuant to Section 10.1(c) and shall reimburse Buyer therefor.
- (a) Buyer shall be responsible for filing with the applicable taxing authorities each Company Tax Return that is required to be filed after the Closing Date and paying the Taxes reflected on such Tax Returns as due and owing; *provided, however*, that in the event that SM Energy is required by Law to file a Tax Return with respect to such Taxes after the Closing Date that includes all or a portion of a Tax period for which Buyer is liable for such Taxes, following SM Energy's request, Buyer shall promptly pay to SM Energy all such Taxes allocable to the period or portion thereof beginning at or after the Effective Time (but only to the extent that such amounts have not already been accounted for under *Section 2.4*), whether such Taxes arise out of the filing of an original Tax Return or a subsequent audit or assessment of Taxes. SM Energy shall be entitled to all Tax credits and Tax refunds that relate to any such Taxes allocable to any Tax period, or portion thereof, ending before the Effective Time.
- (b) Asset Taxes with respect to the Current Tax Period, but excluding Asset Taxes that are based on sales or on the quantity of or the value of production of Hydrocarbons, shall be apportioned between the Parties as of the Effective Time with (i) SM Energy being obligated to pay a proportionate share of the actual amount of such Taxes for the Current Tax Period determined by *multiplying* such actual Taxes by a fraction (A) the numerator of which is the number of days in the Current Tax Period prior to the Effective Time and (B) the denominator of which is the total number of days in the Current Tax Period, and (ii) Buyer being obligated to pay a proportionate share of

the actual amount of such Taxes for the Current Tax Period determined by multiplying such actual Taxes by a fraction (A) the numerator of which is the number of days (including the day that includes the Effective Time) in the Current Tax Period from and after the Effective Time and (B) the denominator of which is the total number of days in the Current Tax Period. Asset Taxes that are based on quantity of or the value of production of Hydrocarbons shall be apportioned between the Parties based on the number of units or value of production actually produced, as applicable, before, on or after the Effective Time. Asset Taxes that are sales, use, or similar Taxes shall be apportioned between the parties based on whether the transaction giving rise to the Tax took place before or after the Effective Time. Any income Tax Liability of the Company or franchise Tax Liability shall be apportioned between the Parties as of the Effective Time with (i) SM Energy being obligated to pay the amount that would be payable if the Tax period of the Company ended on (but did not include) the Effective Time, and (ii) Buyer being obligated to pay the remainder; provided, that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions) shall be allocated between the period ending on (but not including) the Effective Time and the period beginning from and after the Effective Time in proportion to the number of days in each period. Notwithstanding anything to the contrary herein, any Texas franchise Tax shall be allocated to the period during which the income, operations, assets or capital comprising the base of such Tax is measured, regardless of whether the right to do business for another period is obtained by the payment of such franchise Tax. To the extent the Company is included in a combined Texas franchise tax report with SM Energy or any member of the Seller Group, the allocation provided in this Section 10.1(c) shall be calculated as if the Company left the combined group immediately after the Effective Time. In the event that Buyer or SM Energy makes any payment (directly or indirectly) for which it is entitled to reimbursement under this Section 10.1, the applicable Party shall make such reimbursement promptly but in no event later than ten (10) days after the presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of the reimbursement.

(c) The amount of any refunds or credits of Taxes of the Company for any taxable period ending prior to the date of the Effective Time and for the portion of any Current Tax Period through the Effective Time shall be for the account of SM Energy. The amount of any refunds or credits of Taxes of the Company for any taxable period or the portion of any Current Tax Period beginning on or after the date of the Effective Time shall be for the account of the Buyer. The amount of any refunds or credits of Taxes of the Company for any Current Tax Period shall be equitably apportioned between the Buyer and SM Energy in accordance with the principles set forth in Section 10.1(c). Each Party shall forward, and shall cause its Affiliates to forward, to the Party entitled to receive a refund or credit of Tax pursuant to this Section 10.1(d) the amount of such refund or credit within thirty (30) days after such refund is received, or such credit is recognized, but only to the extent that such amounts have not already been accounted for under Section 2.4, in each case net of any costs or expenses incurred by such Party or its Affiliates in procuring such refund or credit. Buyer shall take all necessary actions, or cause the Company to take all necessary actions, as commercially reasonable (i) to obtain any refunds of Taxes of the Company for the account of SM Energy pursuant to this Section 10.1(d), or (ii) to obtain credits or offsets to Taxes of the Company attributable to taxable periods ending prior to the date of the Effective Time or the portion of any Current Tax Period through (but not including) the Effective Time.

- (d) The Purchase Price shall be increased pursuant to *Section 2.4(b)* by (or Buyer shall otherwise reimburse SM Energy for) the amount of Asset Tax for all Tax periods or portions thereof ending on or before the Closing Date that SM Energy or the Company has paid on behalf of other Working Interest owners, royalty interest owners, overriding royalty interest owners and other interests owners in such Assets and that have not been recouped by SM Energy or the Company before the Closing Date from such other Working Interest owners, royalty interest owners, overriding royalty interest owners and other interest owners in such Assets and Buyer shall be subrogated to all rights with respect thereto. Notwithstanding anything to the contrary, to the extent that Buyer receives any Tax refund or credit to which SM Energy is entitled as described above, or any recoupment of Taxes described in the preceding sentence, Buyer shall cause the Company to immediately pay such amount to SM Energy to the extent the Adjusted Purchase Price has not been increased pursuant to *Section 2.4(b)* on account thereof.
- (e) The Parties agree that any payments made pursuant to Section 2.4, Section 2.6, ARTICLE XII or this Section 10.1 shall be treated for all Tax purposes as an adjustment to the Purchase Price unless otherwise required by Law.
- (f) To the extent allowable under applicable law, SM Energy will use its commercially reasonable efforts to obtain third party consents to and cause an election to be made out of subchapter K within the meaning of Section 761 of the Code effective for the taxable year that includes the Closing Date with respect to the tax partnership set forth in *Schedule 3.21*; provided, however, that in no event will SM Energy or any third party be required to make any payment related to such election. If no such election occurs or is invalidated, then notwithstanding anything to the contrary in this Agreement, SM Energy shall retain and reimburse Buyer for, any indemnity obligations of Section 8.1 of the applicable tax partnership agreement as a result of the transactions contemplated by this Agreement.
- 10.2 Transfer Taxes. All sales, use or other Taxes (other than Taxes on gross income, net income or gross receipts), duties, levies, recording fees or other governmental charges incurred by or imposed with respect to the transfer of the Interests undertaken pursuant to this Agreement ("Subject Transfer Taxes") shall be the responsibility of, and shall be paid by, Buyer. The Parties shall reasonably cooperate in taking steps that would minimize or eliminate any Subject Transfer Taxes. Buyer agrees to file all Subject Transfer Tax Returns relating to such Subject Transfer Taxes.
- 10.3 Tax Cooperation. Buyer and SM Energy shall cooperate fully as and to the extent reasonably requested by the other party, in connection with the filing of any Tax Returns and any audit, litigation or other proceeding (each, a "Tax Proceeding") with respect to Taxes relating to or in connection with the Assets. Such cooperation shall include the retention and (upon the other Party's request) the provision of such records and information which are reasonably relevant to any such Tax Return or Tax Proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

ARTICLE XI CLOSING

- 11.1 Date of Closing. Subject to the conditions stated in this Agreement, the transfer by SM Energy and the acceptance by Buyer of the Interests (the "Closing") shall occur on February 28, 2017, or, if all conditions to Closing in ARTICLE IX (other than those conditions that are only capable of being satisfied at the Closing) have not yet been satisfied or waived by that date, five (5) Business Days after such conditions have been satisfied or waived, or such other date as Buyer and SM Energy may agree upon in writing. The date when Closing actually occurs shall be the "Closing Date."
- 11.2 Place of Closing. Closing shall be held at the offices of SM Energy, at 1775 Sherman Street, Suite 1200, Denver, CO 80203 or such other location as Buyer and SM Energy may agree upon in writing.
- 11.3 Closing Obligations. At Closing, the following documents shall be delivered and the following events shall occur, the execution of each document and the occurrence of each event being a condition precedent to the others and each being deemed to have occurred simultaneously with the others:
- (a) SM Energy and Buyer shall execute and deliver the Assignment in sufficient counterparts to facilitate recording in the applicable counties where the Assets are located;
  - (b) SM Energy and Buyer shall execute and deliver the Preliminary Settlement Statement pursuant to Section 2.6(a);
- (c) Buyer shall deliver to SM Energy, to the accounts designated in the Preliminary Settlement Statement, by direct bank or wire transfer in same day funds, the Closing Amount;
- (d) SM Energy shall deliver on forms supplied by Buyer (and reasonably acceptable to SM Energy) transfer orders or letters in lieu thereof directing all purchasers of production to make payment to Buyer or the Company of proceeds attributable to Hydrocarbon production from the Assets from and after the Effective Time, for delivery by Buyer to each purchaser of such Hydrocarbon production;
- (e) SM Energy shall deliver an executed statement described in Treasury Regulation § 1.1445-2(b)(2) certifying that SM Energy is not a "foreign person" or a "disregarded entity" within the meaning of Section 1445 of the Code and its implementing Treasury Regulations;
  - (f) SM Energy shall deliver evidence to the reasonable satisfaction of Buyer that all Required Consents have been obtained;
- (g) Buyer shall execute and deliver a certificate from an authorized officer of Buyer certifying on behalf of Buyer that the conditions set forth in *Section 9.2(a)* and *Section 9.2(b)* have been fulfilled by Buyer;

- (h) SM Energy shall execute and deliver a certificate from an authorized officer of SM Energy certifying on behalf of SM Energy that the conditions set forth in *Section 9.1(a)* and *Section 9.1(b)* have been fulfilled by SM Energy;
- (i) SM Energy shall deliver a recordable release (in forms reasonably acceptable to Buyer and the Financing Sources) of any trust, mortgages, financing statements, fixture filings and security agreements made by SM Energy, the Company or any of its other Affiliates affecting the Assets, including with respect to any matters listed on *Schedule 3.7* and *Schedule 8.1*;
  - (j) Buyer and SM Energy shall execute and deliver the Mutual Release;
  - (k) SM Energy and the Company shall execute and deliver the Seismic License;
  - (1) SM Energy and Buyer shall execute and deliver the Transition Services Agreement; and
- (m) SM Energy and Buyer shall execute and deliver any other agreements, instruments and documents that are required by other terms of this Agreement to be executed or delivered at Closing.
- 11.4 Records. In addition to the obligations set forth under Section 11.3 above, as soon as reasonably practicable following Closing but in any event within thirty (30) days following the Closing Date, SM Energy shall make available to Buyer, during normal business hours at SM Energy's offices, such copies of the Records to which Buyer is entitled pursuant to the terms of this Agreement.

# ARTICLE XII ACQUISITION TERMINATION AND REMEDIES

- 12.1 Right of Termination. This Agreement and the transactions contemplated herein may be terminated at any time at or prior to Closing:
- (a) by SM Energy, if any of the conditions set forth in *Section 9.2* (other than the conditions set forth in *Section 9.2(d)*) have not been satisfied by Buyer on or before March 31, 2017 (the "*Outside Termination Date*");
- (b) by Buyer, if any of the conditions set forth in Section 9.1 (other than the conditions in Section 9.1(d), 9.1(f) or 9.1(h)) have not been satisfied by SM Energy on or before the Outside Termination Date;
  - (c) by SM Energy, if the condition set forth in Section 9.2(d) is not satisfied on or before the Closing Date;
  - (d) by Buyer, if the condition set forth in Section 9.1(d) has not been satisfied on or before the Closing Date;

- (e) by Buyer, if the condition set forth in Section 9.1(f) has not been satisfied on or before the Closing Date;
- (f) by Buyer if the condition set forth in Section 9.1(h) has not been satisfied on or before the Closing Date; or
- (g) by the mutual written agreement of Buyer and SM Energy;

provided, however, that no Party shall have the right to terminate this Agreement pursuant to clause (a) or (b) above if such Party or its Affiliates are at such time in material breach of any provision of this Agreement.

- 12.2 Effect of Termination. If the obligation to close the transactions contemplated by this Agreement is terminated pursuant to any provision of Section 12.1, then, except for the provisions of (i) Section 2.3, Sections 5.1(c) through 5.1(f), Section 5.2, Section 5.3, Section 8.6, this Section 12.2 and Section 13.9, and (ii) such terms as set forth in this Agreement in order to give context to any of the surviving Sections, this Agreement shall forthwith become void and the Parties shall have no liability or obligation hereunder except and to the extent such termination results from the willful breach by a Party of any of its covenants or agreements hereunder, in which case the non-breaching Party shall be entitled to all remedies available at Law or in equity, and shall be entitled to recover court costs and reasonable attorneys' fees in addition to any other relief to which such Party or Parties may be entitled; provided, that except as set forth in Section 12.2(c) below, neither Party shall be entitled to seek the remedy of specific performance from the other Party and, except as set forth in Section 12.2(c), each Party hereby waives any claims for specific performance such Party may have with respect to such breaches.
- (a) If SM Energy has the right to terminate this Agreement pursuant to Section 12.1(a) because a condition set forth in Sections 9.2(a), 9.2(b), 9.2(c) or 9.2(f) has not been satisfied, (i) all of the conditions in Section 9.1 (excluding conditions that, by their terms, cannot be satisfied until the Closing) have been satisfied (or waived by Buyer) (unless the failure to satisfy such condition arises from the action or inaction by Buyer or any Buyer Indemnified Party), and (ii) SM Energy is ready, willing and able to perform its obligations under Section 11.3 then, in such event, SM Energy shall have the right, as SM Energy's sole and exclusive remedy against Buyer, to terminate this Agreement pursuant to Section 12.1(a) and receive the Deposit from the escrow account as liquidated damages, and not as a penalty, for such termination, free and clear of any claims thereon by Buyer, and SM Energy and Buyer shall, subject to Buyer's rights to dispute the validity of any such termination, within five (5) Business Days of the date that this Agreement is terminated or the settlement of all applicable disputes with respect to such termination, take all necessary actions to cause the Escrow Agent to transfer the Deposit to SM Energy free and clear of any claims thereon by Buyer. For the avoidance of doubt, receipt of the Deposit shall constitute full and complete satisfaction of any and all damages that SM Energy may have against Buyer arising out of this Agreement, including specific performance. The provision for payment of the Deposit as liquidated damages in this Section 12.2 has been included because, in the event of a termination of this Agreement permitting SM Energy to receive the Deposit, the actual damages to be incurred by SM Energy can reasonably be expected to

approximate the amount of liquidated damages called for herein and because the actual amount of such damages would be difficult if not impossible to measure accurately.

- (b) If Buyer has the right to terminate this Agreement pursuant to Section 12.1(b) because a condition set forth in Sections 9.1(a), 9.1(b), 9.1(c) or 9.1(i) has not been satisfied (i) all of the conditions in Section 9.2 (excluding conditions that, by their terms, cannot be satisfied until the Closing) have been satisfied (or waived by SM Energy) (unless the failure to satisfy such condition arises from the action or inaction by SM Energy or any SM Indemnified Party), and (ii) Buyer is ready, willing and able to perform its obligations under Section 11.3, then, in such event, (A) Buyer shall be entitled to receipt of the Deposit, the Parties shall have no liability or obligation hereunder as a result of such termination, and SM Energy and Buyer shall, subject to SM Energy's rights to dispute the validity of any such termination, within five (5) Business Days of the date that this Agreement is terminated, take all necessary actions to cause the Escrow Agent to return the Deposit to Buyer free and clear of any claims thereon by SM Energy, or (B) Buyer shall have the right, in lieu of terminating this Agreement, to seek all remedies available at Law or in equity, including specific performance. SM Energy agrees that irreparable damage would occur if any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. SM Energy agrees to waive any requirement for the posting of a bond in connection with any equitable relief hereunder in favor of Buyer, and SM Energy agrees not to challenge any such equitable relief sought in accordance with this Section 12.2(c).
- (c) Notwithstanding anything contained in this Section 12.2, if Buyer terminates this Agreement pursuant to Section 12.1(e) because the condition set forth in Section 9.1(f) has not been satisfied, then Buyer shall be entitled to receipt of the Deposit, the Parties shall have no liability or obligation hereunder as a result of such termination, and SM Energy and Buyer shall, within five (5) Business Days of the date that this Agreement is terminated, take all necessary actions to cause the Escrow Agent to return the Deposit to Buyer free and clear of any claims thereon by SM Energy.
- (d) If this Agreement is terminated for any reason other than as set forth in *Sections 12.2(b)*, *12.2(c)*, *or 12.2(d)* then Buyer shall be entitled to receipt of the Deposit, the Parties shall have no liability or obligation hereunder as a result of such termination, and SM Energy and Buyer shall, within five (5) Business Days of the date that this Agreement is terminated, take all necessary actions to cause the Escrow Agent to return the Deposit to Buyer free and clear of any claims thereon by SM Energy.
- (e) Subject to the foregoing, upon the termination of this Agreement neither Party shall have any other liability or obligation hereunder.
- (f) It is understood for the avoidance of doubt that under no circumstances shall any Financing Source have any liability to SM Energy or any of its Affiliates in respect of the Deposit or any other liability to SM Energy or any of its Affiliates arising hereunder or in connection herewith.
- 12.3 Return of Documentation and Confidentiality. Upon termination of this Agreement, Buyer shall return to SM Energy or destroy (at Buyer's option) all title, engineering, geological and geophysical data, environmental assessments or reports, maps and other information

furnished by SM Energy to Buyer in connection with its due diligence investigation of the Assets, in each case, in accordance with the Confidentiality Agreement.

# ARTICLE XIII COMPANY LIABILITY; SURVIVAL; INDEMNIFICATION

- 13.1 Company Liability. Without limiting Buyer's rights to indemnity under this ARTICLE XIII and subject to any adjustments to the Purchase Price pursuant to Section 2.4, from and after Closing, Buyer expressly acknowledges that the Company is responsible for, and (except as set forth in Section 13.2) neither the Company nor Buyer shall have any recourse against SM Energy for any obligations and Liabilities, known or unknown, of the Company, including those arising under, related to or in connection with the ownership, operation, use or condition of the Assets or the business of the Company, regardless of whether such obligations or Liabilities arose prior to, on or after the Effective Time, including, but not limited to, obligations and Liabilities relating in any manner to the use, ownership or operation of the Assets, such as obligations to: (a) furnish makeup gas or settle Imbalances attributable to the Assets according to the terms of applicable gas sales, processing, gathering or transportation contracts, (a) pay working interests, royalties, overriding royalties and other interest owners' revenues or proceeds attributable to sales of Hydrocarbons produced from the Assets, including those held in suspense, (a) pay the proportionate share attributable to the Assets to properly plug and abandon any and all wells, including inactive wells or temporarily abandoned wells, located on the Assets, (a) pay the proportionate share attributable to the Assets to replug any well, wellbore or previously plugged well on the Assets to the extent required or necessary, (a) pay the proportionate share attributable to the Assets to dismantle or decommission and remove any Personal Property and other property of whatever kind related to or associated with operations and activities conducted by whomever on the Assets, (a) pay the proportionate share attributable to the Assets to clean up, restore or remediate the Assets in accordance with Contracts and Laws, and (a) pay the proportionate share attributable to the Assets to perform all obligations applicable to or imposed on the lessee, owner or operator under the Company's Leases and the Contracts, or as required by any Law, including the payment of all Taxes related to the Assets (all of said obligations and Liabilities, subject to the exclusions below, herein being referred to as the "Assumed Obligations").
- 13.2 Indemnities of SM Energy. Effective as of the Closing, subject to the limitations set forth in Section 13.4 and otherwise contained in this ARTICLE XIII, notwithstanding anything to the contrary contained in any Contribution Document, SM Energy shall defend, indemnify and hold harmless Buyer and its Affiliates (including, after Closing, the Company), and all of its and their respective stockholders, partners, members, directors, officers, managers, employees, agents and representatives (collectively, "Buyer Indemnified Parties") from and against any and all Liabilities, arising from, based upon, related to or associated with:
- (a) any breach by SM Energy of its representations or warranties contained in *ARTICLE III* or in the certificate delivered by SM Energy pursuant *to Section 11.3(h)*;
  - (b) any breach by SM Energy of its covenants and agreements contained in this Agreement;

- (c) any matter set forth on *Schedule 3.10*, (i) with respect to any assets of SM Energy that are not included in the Contribution Assets for all time periods and (ii) with respect to the Assets, solely to the extent such matters relate to the period prior to the Effective Time;
- (d) the disposal or transportation prior to Closing of any Hazardous Substances by or on behalf of SM Energy or the Company or any of its contractors or subcontractors generated or used upon and taken from the Assets to any location not on the Assets;
- (e) personal injury (including death) claims attributable to the ownership or operation of the Assets and arising out of SM Energy's and its Affiliates' (including the Company) period of ownership of the Assets prior to the Closing Date;
- (f) failure to properly and timely pay, or cause to be paid, in accordance with the terms of any Lease, Applicable Contract and applicable Laws, all Royalties with respect to the Assets that are due by or on behalf of SM Energy or any of its Affiliates (including the Company) and attributable to SM Energy's or any of its Affiliates' (including the Company) ownership of the Assets prior to the Effective Time (other than Suspense Funds or any such failure for which Buyer receives the benefit at or prior to Closing);
- (g) claims, demands, damages, costs and expenses, obligations or other liabilities with respect to Third Party surface owner property damage (other than property damage related to, or arising from, an Environmental Condition, Environmental Defect or other environmental matter) by SM Energy or any of its Affiliates (including the Company), or attributable to its or their ownership of the Assets, arising out of ownership on the Assets (to the extent SM Energy or any of its Affiliates (including the Company) are obligated to pay such amounts) prior to the Effective Time;
- (h) civil fines or penalties, criminal sanctions or injunctive relief attributable to SM Energy's or any of its Affiliates' (including the Company) period of ownership of the Assets prior to the Closing Date;
  - (i) Employee Liabilities;
- (j) Asset Taxes attributable to any period (or portion thereof) prior to the Effective Time (as determined pursuant to *Section* 10.1);
  - (k) attributable to or arise out of the ownership, use or operation of the Excluded Assets; or
  - (l) attributable to any income Tax Liability or franchise Tax Liability;

provided, however, in no event shall SM Energy have any obligation to provide indemnification for any matters to the extent adjustments therefor are taken into account in the Preliminary Settlement Statement or the Final Settlement Statement.

13.3 Indemnities of Buyer. Effective as of the Closing, Buyer and its successors and assigns shall defend, indemnify, hold harmless and forever release SM Energy and its Affiliates, and all of their respective stockholders, partners, members, directors, officers, managers, employees,

agents and representatives (collectively, "SM Indemnified Parties") from and against any and all Liabilities arising from, based upon, related to or associated with:

- (a) any breach by Buyer of its representations or warranties contained in ARTICLE IV or in the certificate delivered by Buyer pursuant to Section 11.3(g);
  - (b) any breach by Buyer of its covenants and agreements contained in this Agreement;
  - (c) Subject Transfer Taxes; or
  - (d) the Assumed Obligations;

provided, however, in no event shall Buyer have any obligation to provide indemnification for any matters to the extent adjustments therefor are taken into account in the Preliminary Settlement Statement or the Final Settlement Statement.

## 13.4 Limitation on Liability.

- (a) Except with respect to Taxes and the obligations set forth in Section 10.1(g), if any, SM Energy shall not have any liability for any indemnification under Section 13.2 (other than Liabilities with respect to (x) the breach of the Fundamental Representations or (y) the breach of the representations and warranties set forth in Section 3.4, Section 3.6, Section 3.29, Section 3.31, Section 3.35 and Section 3.37) unless (i) the individual amount of any Liability for which a Claim Notice is delivered by Buyer to SM Energy under this ARTICLE XIII and for which SM Energy is liable exceeds \$100,000 and (ii) the aggregate amount of such Liabilities for which SM Energy is liable under this Agreement after the application of the provisions of clause (i) above exceeds Twenty Million Dollars (\$20,000,000). Notwithstanding the foregoing, the limitations in this Section 13.4(a) shall not apply to indemnification under Sections 13.2(b), 13.2(c), 13.2(i), 13.2(j), 13.2(k) or 13.2(l).
- (b) Except with respect to Taxes and the obligations set forth in Section 10.1(g), if any, notwithstanding anything to the contrary contained in this Agreement, SM Energy shall not be required to indemnify the Buyer Indemnified Parties under Section 13.2 (other than Liabilities with respect to (x) the breach of the Fundamental Representations, (y) the breach of the representations and warranties set forth in Section 3.4, Section 3.6, Section 3.29, Section 3.31, and Section 3.35 or (z) the indemnifications under Sections 13.2(b), 13.2(j), 13.2(k) or 13.2(l)) for aggregate Liabilities in excess of twenty-five percent (25%) of the aggregate unadjusted Purchase Price. Notwithstanding the foregoing, in no event will SM Energy be required to indemnify the Buyer Indemnified Parties under Section 13.2 for aggregate Liabilities in excess of one hundred percent (100%) of the aggregate unadjusted Purchase Price.
- (c) Notwithstanding anything herein to the contrary, for purposes of Section 13.2(a) and Section 13.3(a), when determining the breach or the amount of Liabilities resulting from a breach of representation or warranty contained in this Agreement or in the certificates delivered by the Parties pursuant to Section 11.3(g) or Section 11.3(h), as applicable, all materiality qualifications

(including Material Adverse Effect and any correlative terms) contained in the representations and warranties in ARTICLE III and ARTICLE IV or in such certificates delivered pursuant to Section 11.3 shall be disregarded.

- 13.5 Express Negligence. EXCEPT AS OTHERWISE PROVIDED IN SECTION 5.1(c), THE INDEMNIFICATION, RELEASE, ASSUMED OBLIGATIONS, WAIVER AND LIMITATION OF LIABILITY PROVISIONS PROVIDED FOR IN THIS AGREEMENT SHALL BE APPLICABLE WHETHER OR NOT THE LIABILITIES, LOSSES, COSTS, EXPENSES AND DAMAGES IN QUESTION AROSE OR RESULTED SOLELY OR IN PART FROM THE SOLE, ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY ANY INDEMNIFIED PERSON, EXCEPT TO THE EXTENT CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF SUCH INDEMNIFIED PARTY. BUYER AND SM ENERGY ACKNOWLEDGE THAT THIS STATEMENT COMPLIES WITH THE EXPRESS NEGLIGENCE RULE AND IS "CONSPICUOUS".
- 13.6 Exclusive Remedy for Agreement. Notwithstanding anything to the contrary contained in this Agreement, from and after Closing, Section 5.1(c), Section 13.2 and Section 13.3 contain the Parties' exclusive remedy against each other with respect to breaches of the representations, warranties, covenants and agreements of the Parties contained in this Agreement and the affirmations of such representations, warranties, covenants and agreements contained in the certificate delivered by each Party at Closing pursuant to Section 11.3(g) or Section 11.3(h), as applicable. Except for (a) the remedies contained in this ARTICLE XIII, and (b) other remedies available to the Parties at Law or in equity for breaches of Section 5.1(c) and Section 5.2, and subject to Buyer's rights pursuant to ARTICLE VI and ARTICLE VII, from and after Closing, SM Energy and Buyer each release, remise and forever discharge the other Party and its Affiliates and all such Persons' stockholders, officers, directors, employees, agents, advisors and representatives from any and all Liabilities in Law or in equity, known or unknown, which such Parties might now or subsequently may have, based on, relating to or arising out of (i) this Agreement or the consummation of the transactions contemplated by this Agreement, (ii) the ownership of the Interests, or the ownership, use or operation of the Assets prior to the Closing, or the condition, quality, status or nature of the Assets prior to the Closing, including rights to contribution under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, (iii) breaches of statutory or implied warranties with respect to this Agreement, (iv) nuisance or other tort actions with respect to this Agreement, (v) rights to punitive damages with respect to this Agreement, and (vi) common Law rights of contribution with respect to this Agreement.
- 13.7 Indemnification Procedures. All claims for indemnification under Section 5.1(c), Section 13.2 and Section 13.3 shall be asserted and resolved as follows:
- (a) For purposes of this *ARTICLE XIII*, the term "*Indemnifying Party*", when used in connection with particular Liabilities, shall mean the Party having an obligation to indemnify another Party or Person(s) with respect to such Liabilities pursuant to this *ARTICLE XIII*, and the term "*Indemnified Party*", when used in connection with particular Liabilities, shall mean the Party

or Person(s) having the right to be indemnified with respect to such Liabilities by another Party pursuant to this ARTICLE XIII.

- (b) To make claim for indemnification under Section 5.1(c), Section 13.2 or Section 13.3, an Indemnified Party shall notify the Indemnifying Party of its claim under this Section 13.7, including the specific details of and specific basis under this Agreement for its claim (the "Claim Notice"). In the event that the claim for indemnification is based upon a claim by a Third Party against the Indemnified Party (a "Third Party Claim"), the Indemnified Party shall provide its Claim Notice promptly after the Indemnified Party has actual knowledge of the Third Party Claim and shall enclose a copy of all papers (if any) served with respect to the Third Party Claim; provided, that the failure of any Indemnified Party to give notice of a Third Party Claim as provided in this Section 13.7 shall not relieve the Indemnifying Party of its obligations under Section 5.1(c), Section 13.2 or Section 13.3 (as applicable) except to the extent such failure results in insufficient time being available to permit the Indemnifying Party to effectively defend against the Third Party Claim or otherwise materially prejudices the Indemnifying Party's ability to defend against the claim. In the event that the claim for indemnification is based upon an inaccuracy or breach of a representation, warranty, covenant or agreement that was inaccurate or breached.
- (c) In the case of a claim for indemnification based upon a Third Party Claim, the Indemnifying Party shall have thirty (30) days from its receipt of the Claim Notice to notify the Indemnified Party whether it admits or denies its liability to defend the Indemnified Party against such Third Party Claim at the sole cost and expense of the Indemnifying Party. The Indemnified Party is authorized, prior to and during such thirty (30) day period, at the expense of the Indemnifying Party, to file any motion, answer or other pleading that it shall deem necessary or appropriate to protect its interests or those of the Indemnifying Party and that is not prejudicial to the Indemnifying Party.
- (d) If the Indemnifying Party admits its obligation to indemnify a Third Party Claim, it shall have the right and obligation to diligently defend, at its sole cost and expense, the Third Party Claim; *provided* that, where the Third Party Claim consists of a civil, criminal or regulatory proceeding, action, indictment or investigation against the Indemnified Party by any Governmental Authority, the Indemnified Party shall at its option have the right to control the defense and proceedings. Except as provided in the preceding sentence, the Indemnifying Party shall have full control of such defense and proceedings, including any compromise or settlement thereof unless the compromise or settlement includes the payment of any amount by (because of the Indemnity Deductible or otherwise), the performance of any obligation by or the limitation of any right or benefit of, the Indemnified Party, in which event such settlement or compromise shall not be effective without the consent of the Indemnified Party, which may be withheld in the Indemnified Party's sole discretion. If requested by the Indemnifying Party, the Indemnified Party agrees to cooperate in contesting any Third Party Claim that the Indemnifying Party elects to contest; *provided, however*, that the Indemnified Party shall not be required to bring any counterclaim or cross-complaint against any Person). The Indemnified Party may participate in, but not control, at its own expense, any defense or settlement of any Third Party Claim controlled by the Indemnifying Party pursuant to this *Section 13.7(d)*. An Indemnifying Party shall not, without the written consent of the Indemnified

Party, (i) settle any Third Party Claim or consent to the entry of any judgment with respect thereto which does not result in a final resolution of the Indemnified Party's Liability in respect of such Third Party Claim (including in the case of a settlement an unconditional written release of the Indemnified Party from all Liability in respect of such Third Party Claim) or (ii) settle any Third Party Claim or consent to the entry of any judgment with respect thereto in any manner that may materially and adversely affect the Indemnified Party (other than as a result of money damages covered by the indemnity).

- (e) If the Indemnifying Party does not admit its obligation to indemnify and bear all expenses associated with a Third Party Claim or admits its obligation to indemnify and bear all expenses associated with a Third Party Claim but fails to diligently prosecute or settle the Third Party Claim, then the Indemnified Party shall have the right to defend against the Third Party Claim at the sole cost and expense of the Indemnifying Party, with counsel of the Indemnified Party's choosing, subject to the right of the Indemnifying Party to admit its obligation to indemnify and bear all expenses associated with a Third Party Claim and assume the defense of the Third Party Claim at any time prior to settlement or final determination thereof. If the Indemnifying Party has not yet admitted its obligation to indemnify and bear all expenses associated with a Third Party Claim, the Indemnified Party shall send written notice to the Indemnifying Party of any proposed settlement and the Indemnifying Party shall have the option for 10 Business Days following receipt of such notice to (i) admit in writing its obligation to indemnify and bear all expenses associated with a Third Party Claim and (ii) if such obligation is so admitted, reject, in its reasonable judgment, the proposed settlement.
- (f) In the case of a claim for indemnification not based upon a Third Party Claim, the Indemnifying Party shall have thirty (30) days from its receipt of the Claim Notice to (i) cure the Liabilities complained of, (ii) admit its obligation to indemnify for and bear all expenses associated with such Liability or (iii) dispute the claim for such Liabilities. If the Indemnifying Party does not notify the Indemnified Party within such thirty (30) day period that it has cured the Liabilities or that it disputes the claim for such Liabilities, the amount of such Liabilities shall conclusively be deemed a liability of the Indemnifying Party hereunder.

### 13.8 Survival.

- (a) The representations and warranties of SM Energy contained in *ARTICLE III* (other than the Fundamental Representations or the representations and warranties set forth in *Section 3.6*, *Section 3.18* and *Section 3.21*) shall survive the Closing until the date that is five hundred forty (540) days from the Closing. The representations and warranties of SM Energy contained in *Section 3.6*, *Section 3.18* and *Section 3.21* shall, in each case, survive the Closing until sixty (60) days after the applicable statute of limitations (taking into account any extensions thereof) with respect to the matters and Liabilities covered by such representations and warranties have expired. The Fundamental Representations and the covenants of SM Energy herein shall, in each case, survive the Closing without time limit.
- (b) Subject to Section 13.8(a) and except as set forth in Section 13.8(c), the remainder of this Agreement shall survive the Closing without time limit. Representations, warranties, covenants and agreements shall be of no further force and effect after the date of their

expiration; provided that there shall be no termination of any bona fide claim asserted pursuant to this Agreement with respect to such a representation, warranty, covenant or agreement prior to its expiration date.

- (c) The indemnities in Section 13.2(a), Section 13.2(b), Section 13.3(a) and Section 13.3(b) shall terminate as of the termination date of each respective representation, warranty, covenant or agreement that is subject to indemnification, except, in each case, as to matters for which a specific written claim for indemnity has been delivered to the Indemnifying Party on or before such termination date. The indemnities in Section 13.2(d,) 13.2(e) and 13.2(g) shall survive the Closing until the date that is five hundred forty (540) days from the Closing. The indemnity in Section 13.2(f), shall survive the Closing until the date that is the second (2nd) anniversary of the Closing. The indemnities in Section 13.2(c), 13.2(h), 13.2(i), 13.2(j), 13.2(k) and 13.2(l) shall survive for thirty (30) days following the expiration of the applicable statute of limitations. The indemnities in Section 13.3(c) and Section 13.3(d) shall survive the Closing without time limit.
- 13.9 Non-Compensatory Damages. None of the Buyer Indemnified Parties or SM Indemnified Parties shall be entitled to recover from SM Energy or Buyer, as applicable, or their respective Affiliates, any indirect, consequential, punitive or exemplary damages or damages for lost profits arising under or in connection with this Agreement or the transactions contemplated by this Agreement, except to the extent any such Party suffers such damages (including costs of defense and reasonable attorney's fees incurred in connection with defending against such damages) to a Third Party, which damages (including costs of defense and reasonable attorney's fees incurred in connection with defending against such damages) shall not be excluded by this provision as to recovery hereunder. Subject to the preceding sentence, Buyer, on behalf of each of the Buyer Indemnified Parties, and SM Energy, on behalf of each of SM Indemnified Parties, each waive any right to recover punitive, special, exemplary and consequential damages, including damages for lost profits, arising in connection with this Agreement or the transactions contemplated by this Agreement. This Section 13.9 shall not restrict any Party's right to obtain specific performance or other equitable remedies (other than rescission) pursuant to Section 12.2. Notwithstanding the foregoing, lost profits shall not be excluded by this provision as to recovery hereunder to the extent constituting direct damages.

# ARTICLE XIV MISCELLANEOUS

- 14.1 Counterparts. This Agreement may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall constitute for all purposes one agreement. Any signature hereto delivered by a Party by facsimile transmission shall be deemed an original signature hereto.
- 14.2 Notices. All notices and communications required or permitted to be given hereunder shall be sufficient in all respects if given in writing and delivered personally, or sent by bonded overnight courier, or mailed by U.S. Express Mail or by certified or registered United States Mail with all postage fully prepaid, or sent by facsimile transmission (provided, any such facsimile transmission is confirmed by written confirmation), or sent by electronic mail transmission (provided that the acknowledgement of the receipt of such electronic mail is requested and received, excluding

automatic receipts), addressed to the appropriate Party at the address for such Party shown below or at such other address as such Party shall have theretofore designated by written notice delivered to the Party giving such notice:

# If to SM Energy:

SM Energy Company

1775 Sherman Street, Suite 1200

Denver, CO 80203

Attention: David W. Copeland - Executive Vice President, General Counsel and Corporate

Secretary

Fax: 303.864.2598

Email: dcopeland@sm-energy.com

SM Energy Company

777 North Eldridge Parkway, Suite 1100

Houston, TX 77079

Attention: Kenneth J. Knott – Senior Vice President – Business Development and

Land

Fax: 281.677.2810

Email: kknott@sm-energy.com

If to Buyer:

Venado EF LLC

13501 Galleria Circle, Suite 350

Austin, Texas 78738

Attention: James F. Murchison - Chief Financial Officer

Fax: 512.518.2910

Email: jmurchison@vogllc.com

With a copy to:

Baker Botts L.L.P. 98 San Jacinto Blvd., Suite 1500 Austin, Texas 78701

Attention: Mike Bengtson Fax: 512.322.8349

Email: mike.bengtson@bakerbotts.com

Any notice given in accordance herewith shall be deemed to have been given when (a) delivered to the addressee in person or by courier, (b) transmitted by electronic mail or facsimile transmission during normal business hours, or (c) upon actual receipt by the addressee after such notice has either

been delivered to an overnight courier or deposited in the United States Mail, as the case may be. The Parties may change the address, electronic mail address and facsimile numbers to which such communications are to be addressed by giving written notice to the other Parties in the manner provided in this *Section 14.2*. Notwithstanding anything to the contrary in this *Section 14.2*, no notice of any breach of this Agreement may be delivered by electronic mail.

- 14.3 Expenses. Except as otherwise specifically provided, all fees, costs and expenses incurred by the Parties in negotiating this Agreement shall be paid by the Party incurring the same, including legal and accounting fees, costs and expenses.
- 14.4 Waivers; Rights Cumulative. Any of the terms, covenants or conditions hereof may be waived only by a written instrument executed by or on behalf of the Party waiving compliance. No course of dealing on the part of any Party or its respective officers, employees, agents or representatives, and no failure by a Party to exercise any of its rights under this Agreement shall, in either case, operate as a waiver thereof or affect in any way the right of such Party at a later time to enforce the performance of such provision. No waiver by any Party of any condition, or any breach of any term or covenant contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of any breach of any other term or covenant. The rights of the Parties under this Agreement shall be cumulative, and the exercise or partial exercise of any such right shall not preclude the exercise of any other right.
- 14.5 Relationship of the Parties. The rights, duties, obligations and liabilities of the Parties under this Agreement shall be individual, not joint or collective. It is not the intention of the Parties to create, and this Agreement shall not be deemed or construed to create, a mining or other partnership, joint venture or association or a trust. This Agreement shall not be deemed or construed to authorize any Party to act as an agent, servant or employee for any other Party for any purpose whatsoever except as explicitly set forth in this Agreement. In their relations with each other under this Agreement, the Parties shall not be considered fiduciaries.
- APPENDICES HERETO COLLECTIVELY CONSTITUTE THE ENTIRE AGREEMENT AMONG THE PARTIES PERTAINING TO THE SUBJECT MATTER HEREOF AND SUPERSEDE ALL PRIOR AGREEMENTS, UNDERSTANDINGS, NEGOTIATIONS AND DISCUSSIONS, WHETHER ORAL OR WRITTEN, OF THE PARTIES PERTAINING TO THE SUBJECT MATTER OF THIS AGREEMENT. IN THE EVENT OF A CONFLICT BETWEEN THE TERMS AND PROVISIONS OF THIS AGREEMENT AND THE TERMS AND PROVISIONS OF ANY EXHIBIT HERETO; THE TERMS AND PROVISIONS OF THIS AGREEMENT SHALL GOVERN AND CONTROL; PROVIDED, HOWEVER, THAT THE INCLUSION IN ANY OF THE EXHIBITS HERETO OF TERMS AND PROVISIONS NOT ADDRESSED IN THIS AGREEMENT SHALL NOT BE DEEMED A CONFLICT, AND ALL SUCH ADDITIONAL PROVISIONS SHALL BE GIVEN FULL FORCE AND EFFECT.
- (a) THE TERMS AND PROVISIONS OF THIS AGREEMENT ARE INTENDED SOLELY FOR THE BENEFIT OF THE PARTIES, THEIR RESPECTIVE

SUCCESSORS OR PERMITTED ASSIGNS, AND IT IS NOT THE INTENTION OF THE PARTIES TO CONFER THIRD PARTY BENEFICIARY RIGHTS UPON ANY OTHER PERSON; *PROVIDED, HOWEVER*, THAT NOTWITHSTANDING THE FOREGOING, THE FINANCING SOURCES, THE FINANCE RELATED PARTIES AND THEIR RESPECTIVE REPRESENTATIVES SHALL BE EXPRESS THIRD PARTY BENEFICIARIES OF, AND SHALL BE ENTITLED TO ENFORCE (AND ENTITLED TO RELY ON), SECTION 12.2(G), SECTION 13.6, THIS SECTION 14.6(B), SECTION 14.7, SECTION 14.16 AND SECTION 14.17.

- 14.7 Governing Law; Venue; Jury Waiver. THIS AGREEMENT AND THE LEGAL RELATIONS AMONG THE PARTIES SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, EXCLUDING ANY CONFLICTS OF LAW RULE OR PRINCIPLE THAT MIGHT REFER CONSTRUCTION OF SUCH PROVISIONS TO THE LAWS OF ANOTHER JURISDICTION; PROVIDED THAT (WITHOUT LIMITING THE PROVISIONS OF SECTION 14.17) ANY CLAIMS OR CAUSES OF ACTION (WHETHER IN CONTRACT OR TORT) IN CONNECTION WITH THIS AGREEMENT AGAINST THE FINANCING SOURCES, OR THE FINANCE RELATED PARTIES IN ANY WAY RELATING TO THE DEBT FINANCING AND THE TRANSACTIONS CONTEMPLATED THEREBY SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS OF THE STATE OF NEW YORK.
- (a) ALL OF THE PARTIES CONSENT TO THE EXERCISE OF JURISDICTION IN PERSONAM BY THE COURTS OF THE STATE OF TEXAS FOR ANY DISPUTE. ALL ACTIONS OR PROCEEDINGS WITH RESPECT TO, ARISING DIRECTLY OR INDIRECTLY IN CONNECTION WITH, OUT OF, RELATED TO OR FROM THIS AGREEMENT OR THE OTHER TRANSACTION DOCUMENTS SHALL BE EXCLUSIVELY LITIGATED IN COURTS IN HOUSTON, HARRIS COUNTY, TEXAS. WITHOUT LIMITING THE PROVISIONS OF SECTION 14.17 AND NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS AGREEMENT, EACH OF THE PARTIES HERETO (A) AGREES THAT IT WILL NOT BRING OR SUPPORT ANY ACTION, CAUSE OF ACTION, CLAIM, CROSS-CLAIM OR THIRD-PARTY CLAIM OF ANY KIND OR DESCRIPTION, WHETHER IN LAW OR IN EQUITY, WHETHER IN CONTRACT OR IN TORT OR OTHERWISE, AGAINST THE FINANCING SOURCES OR THE FINANCE RELATED PARTIES IN ANY WAY RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. INCLUDING BUT NOT LIMITED TO ANY DISPUTE ARISING OUT OF OR RELATING IN ANY WAY TO THE FINANCING, IN ANY FORUM OTHER THAN EXCLUSIVELY IN THE SUPREME COURT OF THE STATE OF NEW YORK, COUNTY OF NEW YORK, OR, IF UNDER APPLICABLE LAW EXCLUSIVE JURISDICTION IS VESTED IN THE FEDERAL COURTS, THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK (AND APPELLATE COURTS THEREOF), (B) SUBMITS FOR ITSELF AND ITS PROPERTY WITH RESPECT TO ANY SUCH ACTION TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS, (C) AGREES THAT SERVICE OF PROCESS, SUMMONS, NOTICE OR DOCUMENT BY REGISTERED MAIL ADDRESSED TO IT AT ITS ADDRESS PROVIDED IN SECTION 14.2 SHALL BE EFFECTIVE SERVICE OF PROCESS AGAINST IT FOR ANY SUCH ACTION BROUGHT IN ANY SUCH COURT, (D) IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY

OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF, AND THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF, ANY SUCH ACTION IN ANY SUCH COURT AND (E) AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW.

(b) EACH PARTY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY DISPUTE. INCLUDING ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION THAT MAY BE BASED UPON, ARISE OUT OF OR RELATE TO THE FINANCING SOURCES OR ANY FINANCE RELATED PARTIES OR THE FINANCING.

### 14.8 Further Cooperation.

After the Closing, SM Energy, Buyer and their respective Affiliates shall execute and deliver, or shall cause to be executed and delivered, from time to time such further documents and instruments of conveyance and transfer, and shall take such other actions as SM Energy or Buyer may reasonably request, for carrying out the purposes of this Agreement or of any document delivered pursuant to this Agreement, to convey and deliver the Interests to Buyer, to perfect Buyer's title to any of the Assets and to accomplish the orderly transfer of the Interests to Buyer in the manner contemplated by this Agreement.

- **14.9 Amendment.** This Agreement may be amended only by an instrument in writing executed by all of the Parties and expressly identified as an amendment or modification hereof.
- 14.10 Parties in Interest. Except as set forth in Section 13.2 or Section 13.3, nothing in this Agreement shall entitle any Person other than the Parties to any claim, cause of action, remedy or right of any kind.
- 14.11 Successors and Permitted Assigns. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. This Agreement may not be assigned by either Party without the prior written consent of the other Party; provided, however, Buyer may, by written notice to SM Energy, assign Buyer's interest in this Agreement or all or a portion of the Interests to one or more Affiliates of Buyer and in such event Buyer shall remain liable for all Buyer obligations under this Agreement.
- 14.12 Publicity. Buyer and SM Energy shall not make or issue any press release or other public announcements concerning the transactions contemplated by this Agreement without the prior consent of the other Party, which consent shall not be unreasonably withheld. If either Party desires to make a public announcement, it shall first give the other Party reasonable advance written notification of its desire to make such a public announcement. The written notification shall include (i) a request for consent to make the announcement, and (ii) a written draft of the text of such public announcement. Notwithstanding anything to the contrary in this Section 14.12, (A) a Party may make any public announcement where such release or statement is deemed in good faith by the releasing Party and upon opinion of counsel to be required by Law or under the rules and regulations

of a recognized stock exchange on which shares of such Party or any of its Affiliates are listed; and (B) any Party or Affiliate of a Party may disclose information regarding the Assets in investor presentations, industry conference presentations or similar disclosures to the extent that such information has previously been publicly released; *provided* that such disclosure shall not include the name of the non-disclosing Party or its Affiliates without such Party's prior written consent.

- 14.13 Preparation of Agreement. Both SM Energy and Buyer and their respective counsel participated in the preparation of this Agreement. In the event of any ambiguity in this Agreement, no presumption shall arise based on the identity of the draftsman of this Agreement.
- 14.14 Conduct of the Parties. Each Party warrants that it and its Affiliates have not made, offered or authorized and agrees that it will not make, offer or authorize with respect to the matters which are the subject of this Agreement, any payment, gift, promise or other advantage, whether directly or through any other Person, to or for the use or benefit of any public official (being any person holding a legislative, administrative or judicial office, including any person employed by or acting on behalf of a public agency, a public enterprise or public international organization) or any political party or political party official or candidate for office, where such payment, gift, promise or advantage would violate any applicable Law.
- 14.15 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any adverse manner to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.
- 14.16 No Recourse. Notwithstanding anything that may be expressed or implied in this Agreement or any document, agreement, or instrument delivered contemporaneously herewith, and notwithstanding the fact that Buyer may be a partnership or limited liability company, but in all events subject to SM Energy's rights to receive the Deposit as liquidated damages pursuant to the terms of this Agreement and subject to Section 14.11, SM Energy, by its acceptance of the benefits of this Agreement, covenants, agrees and acknowledges that no Persons other than Buyer (and their respective successors and assigns, collectively, the "Recourse Parties") shall have any obligation hereunder and that it has no rights of recovery hereunder against, and no recourse hereunder or under any documents, agreements, or instruments delivered contemporaneously herewith or in respect of any oral representations made or alleged to be made in connection herewith or therewith against (a) any former, current or future director, officer, agent, attorney, financing source (including the Financing Sources), Affiliate, manager, incorporator, controlling Person, fiduciary, representative or employee of Buyer (or any of the foregoing Persons successors or permitted assignees), (b) any former, current, or future general or limited partner, owner, manager, stockholder or member of Buyer (or any of the foregoing Persons successors or permitted assignees) or any Affiliate thereof or (c) any former, current or future director, owner, officer, agent, employee, attorney, financing source (including the Financing Sources), Affiliate, manager, incorporator,

controlling Person, fiduciary, representative, general or limited partner, stockholder, manager or member of any of the foregoing, but in each case not including Buyer (each, but excluding for the avoidance of doubt, the Recourse Parties, a "Party Affiliate"), whether by or through attempted piercing of the corporate veil, by or through a claim (whether in tort, contract or otherwise) by or on behalf of Buyer against the Party Affiliates, by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other Law, or otherwise; it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on, or otherwise be incurred by any Party Affiliate, as such, for any obligations of Buyer under this Agreement or the transactions contemplated hereby, under any documents or instruments delivered contemporaneously herewith, in respect of any oral representations made or alleged to be made in connection herewith or therewith, or for any claim (whether in tort, contract or otherwise) based on, in respect of, or by reason of, such obligations or their creation. Notwithstanding anything that may be expressed or implied in this Agreement or any document, agreement, or instrument delivered contemporaneously herewith, Buyer, by its acceptance of the benefits of this Agreement, covenants, agrees and acknowledges that no Persons other than SM Energy shall have any obligation hereunder and that Buyer has no rights of recovery hereunder against, and no recourse hereunder or under any documents, agreements, or instruments delivered contemporaneously herewith or in respect of any oral representations made or alleged to be made in connection herewith or therewith against any of SM Energy's financial advisors whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other Law, or otherwise; it being expressly agreed and acknowledged that no personal liability whatsoever shall attach to, be imposed on, or otherwise be incurred by any of SM Energy's financial advisors, as such, for any obligations of SM Energy under this Agreement or the transactions contemplated hereby, under any documents or instruments delivered contemporaneously herewith, in respect of any oral representations made or alleged to be made in connection herewith or therewith, or for any claim (whether in tort, contract or otherwise) based on, in respect of, or by reason of, such obligations or their creation.

14.17 Waiver of Claims Against Debt Financing Sources. Notwithstanding anything in this Agreement to the contrary, SM Energy agrees, on behalf of itself and its Affiliates, that none of the Financing Sources or any of their respective general or limited partners, stockholders, managers, members, agents, representatives, Affiliates, successors or assigns (collectively, "Finance Related Parties") shall have any Liability to SM Energy or its Affiliates relating to or arising out of this Agreement or the transactions contemplated by this Agreement, including the financing of the transactions contemplated by this Agreement, whether at Law or equity, in contract, in tort or otherwise, and that neither SM Energy nor any of its Affiliates will have any rights or claims against any Finance Related Parties under this Agreement or any other agreement contemplated by, or entered into in connection with, the transactions contemplated by this Agreement, including any commitments by the Finance Related Parties in respect of financing the transactions contemplated by this Agreement.

**IN WITNESS WHEREOF**, the Parties have executed this Agreement by their duly authorized representatives on and as of the Execution Date.

# **SM ENERGY COMPANY**

By: /s/ DAVID W. COPELAND

Name: David W. Copeland

Title: Executive Vice President, General Counsel and Corporate

Secretary

# VENADO EF LLC

By: /s/ R. SCOTT GARRICK
Name: R. Scott Garrick
Title: Chief Executive Officer

SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT

### Annex I

### **Definitions**

- "Accounting Arbitrator" has the meaning set forth in Section 2.6(c).
- "AFE" has the meaning set forth in Section 3.16.
- "Affected Assets" has the meaning set forth in Section 6.4(c)(ii).
- "Affected Contracts" has the meaning set forth in Section 6.4(d)(ii).
- "Affiliate" means, with respect to any Party, a Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by or is under common Control with, such Party.
  - "Aggregate Deductible" has the meaning set forth in Section 6.2(h).
  - "Agreement" has the meaning set forth in the preamble to this Agreement.
  - "Allocated Value" has the meaning set forth in Section 2.5.
- "Applicable Contracts" means all contracts (i) to which the Company or SM Energy is a party that relate to the Oil and Gas Properties or (ii) that will be binding on the Assets or the Company after Closing, including, without limitation; farmin and farmout agreements; bottomhole agreements; crude oil, condensate and natural gas purchase and sale agreements; gathering, transportation and marketing agreements; hydrocarbon storage agreements; acreage contribution agreements; operating agreements (including, for the avoidance of doubt, Applicable Operating Agreements); balancing agreements; pooling declarations or agreements; unitization agreements; processing agreements; crossing agreements and other similar contracts and agreements.
- "Applicable Operating Agreements" means, collectively, the joint operating agreements applicable to the Assets, and "Applicable Operating Agreement" means any of them.
- "Asset Taxes" means ad valorem, property, excise, severance, production, sales, use, or similar taxes (including any interest, fine, penalty or additions to tax imposed by Governmental Authorities in connection with such taxes) based upon operation or ownership of the Assets or the production of Hydrocarbons therefrom, but excluding, for the avoidance of doubt, income, capital gains and franchise taxes.
- "Assets" means, excluding the Excluded Assets, all assets owned by the Company, including the Oil and Gas Properties and the Springfield System Ownership Interests.
  - "Assignment" means the Assignment, substantially in the form attached hereto as Exhibit C.
  - "Assumed Obligations" has the meaning set forth in Section 13.1.

- "Benefit Plan" means: (a) any "employee benefit plan," as such term is defined in Section 3(3) of ERISA, whether or not subject to ERISA, (b) any stock bonus, stock ownership, stock option, stock purchase, stock appreciation rights, phantom stock, or other stock plan (whether qualified or nonqualified), and (c) any bonus or incentive compensation plan.
- "Burdens" means, with respect to any Asset, all royalties, overriding royalties, production payments, carried interests, net profits interests, reversionary interests and other burdens upon, measured by or payable out of, production therefrom.
  - "Business Day" means a day (other than a Saturday or Sunday) on which commercial banks in Texas are generally open for business.
  - "Buyer" has the meaning set forth in the preamble to this Agreement.
  - "Buyer Indemnified Parties" has the meaning set forth in Section 13.2.
  - "Buyer's Representatives" has the meaning set forth in Section 5.1(a).
  - "Casualty Loss" has the meaning set forth in Section 6.3(b).
  - "Claim Notice" has the meaning set forth in Section 13.7(b).
  - "Closing" has the meaning set forth in Section 11.1.
  - "Closing Amount" means the Preliminary Purchase Price less the Deposit.
  - "Closing Date" has the meaning set forth in Section 11.1.
  - "Code" means the Internal Revenue Code of 1986, as amended.
  - "Company" has the meaning set forth in the recitals to this Agreement.
- "Confidentiality Agreement" means that certain Confidentiality Agreement between SM Energy and Venado Oil & Gas, LLC, dated as of October 19, 2016.
- "Consents" means all consents relating to the Assets (or an item that would otherwise constitute an Asset but for the fact it was not included in the Contribution because of a Consent or Preferential Right) or the Interests that would be applicable in connection with (i) the consummation by Buyer of the transactions contemplated by this Agreement, or (ii) the Contribution.
- "Contracts" means all contracts to which the Company is a party that relate to the Assets and that will be binding on the Assets or the Company after Closing, including, without limitation; farmin and farmout agreements; bottomhole agreements; crude oil, condensate and natural gas purchase and sale agreements; gathering, transportation and marketing agreements; hydrocarbon storage agreements; acreage contribution agreements; operating agreements (including, for the avoidance of doubt, Applicable Operating Agreements); balancing agreements; pooling declarations

or agreements; unitization agreements; processing agreements; crossing agreements and other similar contracts and agreements.

"Contribution" means the contribution of the Contribution Assets by SM Energy or its Affiliates to the Company.

"Contribution Assets" means the Oil and Gas Properties and all associated or related assets used or held for use by SM Energy or its Affiliates (excluding the Company) in the ownership and maintenance of the Oil and Gas Properties, including, without limitation, the Springfield System Ownership Interests, in each case as owned and maintained by SM Energy and its Affiliates (excluding the Company) as of the Effective Time.

"Contribution Documents" has the meaning set forth in Section 3.35.

"Control" and its derivatives mean, with respect to any Person, the possession, directly or indirectly, of the power to exercise or determine the voting of more than fifty percent (50%) of the voting rights in a corporation, and, in the case of any other type of entity, the right to exercise or determine the voting of more than fifty percent (50%) of the equity interests having voting rights, or otherwise to direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.

"COO Agreement" means that certain Agreement for the Construction, Ownership and Operation of Midstream Assets in Maverick, Dimmit, Webb and La Salle Counties, Texas effective as of January 1, 2006, by and between SM Energy Company and Springfield Pipeline LLC, and as amended by that First Amendment to Agreement for the Construction, Ownership and Operation of Midstream Assets in Maverick, Dimmit, Webb and La Salle Counties, Texas effective as of January 1, 2016, by and between Springfield Pipeline LLC, SM Energy Company, Eagle Ford MS LLC, and Mitsui E&P Texas LP.

"Counterparties" has the meaning set forth in Section 8.6.

"Cure Period" has the meaning set forth in Section 6.2(c)(ii).

"Current Tax Period" means any Tax period beginning before, and ending after the Effective Time.

"Customary Post-Closing Consents" means those consents and approvals from Governmental Authorities for the assignment of the Assets to the Buyer that are customarily obtained after the assignment of properties similar to the Assets.

"Defect Claim Date" has the meaning set forth in Section 6.2(a).

"Defensible Title" means such title of the Company with respect to the Assets that, subject to Permitted Encumbrances, is deducible of record or title evidenced by unrecorded instruments or elections, in each case, made or delivered pursuant to joint operating agreements, pooling agreements or unitization agreements and:

- (a) with respect to each Well or Well Location shown in **Exhibit B** (but limited to the applicable Target Interval set forth in **Exhibit B** for such Well or Well Location and any currently producing formation (for such Well)), entitles the Company to receive not less than the Net Revenue Interest shown in **Exhibit B**, for such Well or Well Location throughout the duration of the productive life of such Well or Well Location, except for (i) decreases in connection with those operations in which the Company may, from and after the Execution Date, be a non-consenting co-owner (to the extent permitted pursuant to *Section 8.1*), (ii) decreases resulting from the establishment or amendment from and after the Execution Date of pools or units, (iii) decreases required to allow other Working Interest owners to make up past underproduction or pipelines to make up past under deliveries, and (iv) as otherwise stated in **Exhibit B**;
- (b) with respect to each Well or Well Location shown in **Exhibit B** (but limited to the applicable Target Interval set forth in **Exhibit B** for such Well or Well Location and any currently producing formation (for such Well)), obligates the Company to bear the Working Interest for such Well or Well Location not greater than the Working Interest shown in **Exhibit B**, for such Well or Well Location without increase throughout the productive life of such Well or Well Location, except (i) increases resulting from contribution requirements with respect to defaults by co-owners from and after the Execution Date under Applicable Operating Agreements, (ii) increases to the extent that they are accompanied by a proportionate increase in the Net Revenue Interest in the Assets, and (iii) as otherwise stated in **Exhibit B**; and
  - (c) is free and clear of all Encumbrances.
  - "Deposit" has the meaning set forth in Section 2.2(a).
- "Dispute" means any dispute, controversy or claim (of any and every kind or type, whether based on contract, tort, statute, regulation or otherwise) arising out of, relating to or connected with this Agreement or the transactions contemplated hereby, including but not limited to any dispute, controversy or claim concerning the existence, validity, interpretation, performance, breach or termination of this Agreement, the relationship of the Parties arising out of this Agreement or the transactions contemplated hereby.
  - "Dispute Notice" has the meaning set forth in Section 2.6(b).
  - "DOJ" means the Antitrust Division of the U.S. Department of Justice.
  - "Effective Time" means 7:00 a.m. local time at the location of the Assets on November 1, 2016.
- "Encumbrance" means a mortgage, lien, security interest, pledge, charge or other encumbrance, and "Encumber" and other similar derivatives shall be construed accordingly.
  - "Environmental Arbitrator" has the meaning set forth in Section 7.1(e).
- "Environmental Condition" means (a) a condition existing on or prior to the Defect Claim Date with respect to the environment, including air, soil, subsurface, surface waters, ground waters

or sediments that causes any Asset (or SM Energy or the Company with respect to any Asset) not to be in compliance with, or constitutes a violation of, any Environmental Law or (b) the existence as of or prior to the Defect Claim Date with respect to any Asset or the operation thereof of any condition, including any environmental pollution, contamination, degradation, damage or injury caused by, related to such Asset for which remedial or corrective action is presently required (or if known, would be presently required), or which gives rise to a Liability, under Environmental Laws.

"Environmental Defect" means an Environmental Condition with respect to an Asset that is not set forth in Schedule 3.17.

"Environmental Defect Notice" has the meaning set forth in Section 7.1(a).

"Environmental Defect Property" has the meaning set forth in Section 7.1(a).

"Environmental Laws" means all applicable federal, state and local Laws in effect as of or prior to the Execution Date, including common Law, relating to pollution or the protection of the public health, welfare and the environment, including, without limitation, those Laws relating to the generation, storage, handling, use, processing, treatment, transportation, disposal, release or other management of, or exposure to, chemicals and other Hazardous Substances. The term "Environmental Laws" does not include good or desirable operating practices or standards that may be voluntarily employed or adopted by other similarly situated oil and gas well operators or recommended (but not required) by any Governmental Authority to the extent such operating practices or standards are not legally binding or otherwise required.

"Escrow Agent" means JP Morgan Chase Bank, N.A.

"Escrow Agreement" means that certain Escrow Agreement, dated the date hereof, among SM Energy, Buyer and the Escrow Agent, as such may be amended, modified or replaced from time to time.

"Excluded Assets" means (a) all of SM Energy's corporate minute books, financial records and other business records that relate to SM Energy's business generally (including the ownership and operation of the Assets); (b) all trade credits, all accounts, receivables and all other proceeds, income or revenues attributable to the Assets with respect to any period of time prior to the Effective Time; (c) subject to Section 6.3, all rights and interests relating to the Assets under any bond; (d) all Hydrocarbons produced and sold from the Assets with respect to all periods prior to the Effective Time; (e) all claims of SM Energy or its Affiliates for refunds of or loss carry forwards with respect to (i) production or any other Taxes paid by SM Energy or its Affiliates attributable to any period prior to the Effective Time, (ii) income Taxes paid by SM Energy or its Affiliates or (iii) any Taxes attributable to the other Excluded Assets; (f) all personal computers and associated peripherals and all radio and telephone equipment; (g) all of SM Energy's proprietary computer software, patents, trade secrets, copyrights, names, trademarks, logos and other intellectual property; (h) all documents and instruments of SM Energy that may be protected by an attorney-client privilege. excluding any and all title opinions, environmental reports or evaluations and any such documents and instruments that relate to or cover any Assumed Obligations; (i) all data that cannot be disclosed to Buyer as a result of confidentiality arrangements under agreements with Third Parties (provided that SM Energy

has used commercially reasonable efforts to obtain a waiver of such confidentiality from such Third Parties); (j) all audit rights arising under any of the (i)Applicable Contracts or otherwise with respect to any period prior to the Effective Time or (ii) other Excluded Assets, except for any Imbalances; (k) marketing materials prepared or received by SM Energy or its Affiliates with respect to (i) lists of prospective purchasers for the Assets or the Interests, (iii) bids submitted by other prospective purchasers of the Assets or the Interests, (iii) analyses by SM Energy or its Affiliates of any bids submitted by any prospective purchaser, (iv) correspondence between or among SM Energy, its representatives and any prospective purchaser other than Buyer, and (v) correspondence between SM Energy or any of its representatives with respect to any of the bids, the prospective purchasers or the transactions contemplated by this Agreement; (l) a copy of all Records; (m) any Assets and any Applicable Contracts and Records that are related to such Assets that are excluded pursuant to the provisions of *Section 6.4(c)(ii)*, or *Section 6.4(d)*; (n) any contracts that constitute master services agreements or similar contracts; and (o) the assets described in **Exhibit F**.

"Execution Date" has the meaning set forth in the preamble to this Agreement.

"Existing SM Energy Credit Agreement" means that certain Fifth Amended and Restated Credit Agreement dated as of April 12, 2013 (as amended, supplemented, or otherwise modified prior to the date hereof) among SM Energy, as borrower, each of the lenders party thereto and Wells Fargo Bank, N.A., as administrative agent.

"Final Settlement Statement" has the meaning set forth in Section 2.6(b).

"Finance Related Parties" has the meaning set forth in Section 14.17.

"Financing" means one or more transactions entered into by Buyer or its Affiliates with one or more Financing Sources whereby Buyer will obtain debt or equity financing.

"Financing Sources" means the agents, arrangers, lenders and other entities that have committed to provide or arrange or otherwise entered into agreements in connection with all or any part of the Financing, including the parties to any joinder agreements, indentures or credit agreements entered into in connection therewith.

"FTC" means the Federal Trade Commission.

"Fundamental Representations" means the representations and warranties of SM Energy set forth in Sections 3.1, 3.2, 3.3, 3.5, 3.7, 3.11, 3.18, 3.19 and 3.21, and the representation and warranties of Buyer set forth in Sections 4.1, 4.2, 4.3(a), 4.4, and 4.8.

"GAAP" means the generally accepted accounting principles in the United States of America.

"Governmental Authority" means any federal, state, local, municipal, tribal or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitle to exercise any administrative, executive, judicial, legislative, regulatory or taxing authority or power; and any court or governmental tribunal, including any tribal authority having or asserting jurisdiction.

- "Hard Consent" has the meaning set forth in Section 6.4(d).
- "Hazardous Substances" means any pollutants, contaminants, toxics or hazardous or extremely hazardous substances, or any other substances, materials, wastes, constituents, compounds or chemicals that are regulated by, or may form the basis of any liability under, any Environmental Laws, including NORM, Hydrocarbons and other substances referenced in Section 7.2.
  - "Hedge Contract" has the meaning set forth in Section 8.6(a).
  - "Hedging Volumes" has the meaning set forth in Section 8.6(a).
- "HSR Act" means the Hart-Scott-Rodino-Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.
- "Hydrocarbons" means oil, gas and other hydrocarbons (including condensate) produced or processed in association therewith (whether or not such item is in liquid or gaseous form), including all crude oils, condensates and natural gas liquids at atmospheric pressure and all gaseous hydrocarbons (including wet gas, dry gas and residue gas) or any combination thereof, and any minerals produced in association therewith.
- "Imbalance" means any imbalance at the (a) wellhead between the amount of Hydrocarbons produced from a Well and allocable to the interests of the Company therein and the shares of production from the relevant Well to which the Company is entitled or (b) pipeline flange between the amount of Hydrocarbons nominated by or allocated to the Company and the Hydrocarbons actually delivered on behalf of the Company at that point.
- "Indebtedness" means, without duplication, all indebtedness, Liabilities and obligations, now existing or hereafter arising, for money borrowed by a Person (including accrued and unpaid interest), or any contingent liability for or guaranty by a Person of any obligation of any other Person (including the pledge of any collateral or grant of any security interest by a Person in any property as security for any such liability, guaranty or obligation) whether or not any of the foregoing is evidenced by any note, indenture, guaranty or agreement, but excluding all trade payables incurred in the ordinary course of business.
  - "Indemnified Party" has the meaning set forth in Section 13.7(a).
  - "Indemnifying Party" has the meaning set forth in Section 13.7(a).
  - "Individual Environmental Threshold" means \$75,000.
  - "Individual Title Defect Threshold" means \$50,000.
  - "Interests" has the meaning set forth in the recitals to this Agreement.
  - "Interim Period" means that period of time from and after the Effective Time up to Closing.

"Invasive Activity" has the meaning set forth in Section 5.1(b).

"Knowledge" means with respect to SM Energy, the actual knowledge (after due inquiry of direct reports) of the following Persons: Mary Ellen Lutey, Senior Vice President and Regional Manager; Kenneth Knott, Senior Vice President – Business Development and Land; Brad Peterson, Asset Manager; and Kelly Donohoue, Senior Landman, and, with respect to Buyer, the actual knowledge (after due inquiry of direct reports) of the following Persons: R. Scott Garrick, Chief Executive Officer and James F. Murchison, Chief Financial Officer, and, with respect to the Company, the actual knowledge (after due inquiry of direct reports) of the following Persons: Mary Ellen Lutey, Senior Vice President and Regional Manager; Kenneth Knott, Senior Vice President – Business Development and Land; Brad Peterson, Asset Manager; and Kelly Donohoue, Senior Landman.

"Laws" means any constitution, decree, resolution, law, statute, act, ordinance, rule, directive, order, treaty, code or regulation and any injunction or final non-appealable judgment or any interpretation of the foregoing, as enacted, issued or promulgated by any Governmental Authority.

"Leases" means, except for the Excluded Assets, SM Energy's or the Company's right, title and interest in and to the oil and gas leases, oil, gas, and mineral leases and mineral interests (and all leasehold estates created thereby), described in **Exhibit A**, in each case, subject to any depth limitations applicable thereto, if any.

"Liabilities" means any and all claims, causes of actions, payments, charges, judgments, assessments, liabilities, losses, damages, penalties, fines, costs and expenses, including any attorneys' fees and legal or other expenses incurred in connection therewith and including liabilities, costs, losses and damages for personal injury or death or property damage.

"Material Adverse Effect" means any change, inaccuracy, effect, event, result, occurrence, condition or fact (for the purposes of this definition, each, an "event") (whether foreseeable or not and whether covered by insurance or not) that has had or would be reasonably likely to have, individually or in the aggregate with any other event or events, a material adverse effect on the ownership, operation or financial condition of the Assets, taken as a whole; provided, however, that "Material Adverse Effect" shall not include such material adverse effects resulting from (a) general changes in oil and gas prices; (b) general changes in industry, economic or political conditions or markets; (c) changes in conditions or developments generally applicable to the oil and gas industry in any or all of the states where the Assets are located; (d) acts of God, including hurricanes and storms; (e) orders, actions or failures to act of Governmental Authorities, except to the extent arising out of SM Energy's or the Company's action or inaction; (f) civil unrest or similar disorder, terrorist acts or changes in Laws; (g) effects or changes that are cured or no longer exist by the earlier of Closing and the termination of this Agreement pursuant to Section 12.1, without cost to Buyer; (h) changes in GAAP; and (i) changes resulting from the announcement of the transactions contemplated hereby or the performance of the covenants set forth in ARTICLE VIII; provided, that in each case, the changes and effects described in clauses (a), (b) and (c) of this definition do not disproportionately affect the Assets, taken as a whole.

- "Material Contracts" has the meaning set forth in Section 3.12(a).
- "Membership Interests" means, with respect to a Person, the member interests, limited liability company interests, other equity interests, rights to profits or revenue and any other similar interest, and any security or other interest convertible into the foregoing, or any right (contingent or otherwise) to acquire any of the foregoing, of such Person.
  - "Month" means any of the months of the Gregorian calendar.
  - "Mutual Release" means a document in substantially the form attached hereto as Exhibit H.
- "Net Revenue Interest" means, with respect to any Well or Well Location (but limited to the applicable Target Interval set forth in **Exhibit B** for such Well or Well Location and any currently producing formation (for such Well)), the interest in and to all Hydrocarbons produced, saved and sold from or allocated to such Well or Well Location (but limited to the applicable Target Interval set forth in **Exhibit B** for such Well or Well Location and any currently producing formation (for such Well)), after giving effect to all Burdens thereon.
  - "NORM" means naturally occurring radioactive material.
- "Oil and Gas Properties" means the Leases, Wells, Units, Seismic and all infrastructure relating to the Leases, Wells and Units, including the Springfield Gathering System.
- "Operating Expenses" means all operating expenses attributable to the Assets (including Asset Taxes) and capital expenditures incurred in the ownership and operation of the Assets in the ordinary course of business and, where applicable, in accordance with the Applicable Operating Agreement, and overhead costs charged to the Assets as reflected in SM Energy's financial statements, but excluding Liabilities attributable to (a) personal injury or death, property damage or violation of any Law, (b) obligations to plug wells and dismantle or decommission facilities, (c) curing any Title Defect or obligations to remediate any Environmental Condition under applicable Environmental Laws, (d) obligations with respect to Imbalances, or (e) obligations to pay Working Interests, royalties, overriding royalties or other interest owners revenues or proceeds attributable to sales of Hydrocarbons relating to the Assets, including those held in suspense.
  - "Operator" means the Person serving as operator under any Applicable Operating Agreement.
  - "Outside Termination Date" has the meaning set forth in Section 12.1(a).
  - "Outstanding Contribution Consents" means those Consents listed under the heading "Outstanding Hard Consents" on Schedule 6.4.
  - "Party" and "Parties" have the meanings set forth in the preamble to this Agreement.
  - "Party Affiliate" has the meaning set forth in Section 14.16.

"Permits" means any permits, licenses, authorizations, registrations, consents, approvals, allowances or orders granted or issued by any Governmental Authority, in each case as necessary for the conduct of the Company with respect to its business, including with respect to the Assets.

## "Permitted Encumbrances" means:

- (a) lessor's royalties, non-participating royalties, overriding royalties, reversionary interests and similar burdens upon, measured by or payable out of production if the net cumulative effect of such burdens does not (i) interfere with the operation or use of any of the Assets (as currently operated and used), (ii) operate to reduce the Net Revenue Interest of the Company in any Well or Well Location to an amount less than the Net Revenue Interest set forth in **Exhibit B** for such Well or Well Location, or (ii) obligate the Company to bear a Working Interest for such Well or Well Location in any amount greater than the Working Interest set forth in **Exhibit B** for such Well or Well Location (unless the Net Revenue Interest for such Asset is greater than the Net Revenue Interest set forth in **Exhibit B**, in the same proportion as any increase in such Working Interest);
- (b) Preferential Rights or similar agreements with respect to which (i) waivers are obtained from the appropriate parties for the transaction contemplated hereby, or (ii) required notices have been given for the transaction contemplated hereby to the holders of such rights and the appropriate period for making an election has expired without an exercise of such rights;
- (c) required Third Party consents to assignments or similar agreements with respect to which (A) consents have been obtained from the appropriate parties for the transaction contemplated hereby, or (B) required notices have been given for the transaction contemplated hereby to the holders of such rights and the applicable period (as specified in the contract, agreement or other instrument granting or reserving such rights) for giving notice of objection or withholding of consent has expired without an exercise of such rights or the period within which the failure to respond to such notice is considered under the relevant contract, agreement or other instrument as deemed consent has expired without the Company's receipt of a notice of objection or withholding of consent;
  - (d) liens for Taxes or assessments not yet due or delinquent;
  - (e) Customary Post-Closing Consents;
  - (f) other than such rights that have already been triggered, conventional rights of reassignment;
- (g) such Title Defects as Buyer may have waived or is deemed to have waived pursuant to the terms of this Agreement or Title Defects that were not properly asserted by Buyer prior to the Defect Claim Date (other than claims which may be made pursuant to the special warranty of title set forth in the Assignment or the Contribution Documents);
- (h) all applicable Laws, and rights reserved to or vested in any Governmental Authority (i) to control or regulate any Asset in any manner (excluding eminent domain or

condemnation); (ii) by the terms of any right, power, franchise, grant, license or permit, or by any provision of Law, to terminate such right, power, franchise grant, license or permit or to purchase or recapture or to designate a purchaser of any of the Assets; (iii) to use such property in a manner which does not materially impair the use of such property for the purposes for which it is currently owned and operated and (iv) to enforce any obligations or duties affecting the Assets to any Governmental Authority, with respect to any franchise, grant, license or permit;

- (i) rights of a common owner of any interest in rights-of-way or easements currently held by the Company and such common owner as tenants in common or through common ownership (*provided* that such rights do not prevent or substantially impair the present operation, value or present use of the Assets subject thereto for the purpose of oil and gas development and operations);
- (j) easements, conditions, covenants, restrictions, servitudes, permits, rights-of-way, surface leases and other rights in the Assets for the purpose of surface operations, roads, alleys, highways, railways, pipelines, transmission lines, transportation lines, distribution lines, power lines, telephone lines and removal of timber, grazing, logging operations, canals, ditches, reservoirs and other like purposes, or for the joint or common use of real estate, rights-of-way, facilities and equipment, (in each case) that do not (i) materially impair the use, ownership or operation of the Assets (as currently owned and operated), (ii) reduce the Net Revenue Interest of the Company in any Well or Well Location to an amount less than the Net Revenue Interest set forth in **Exhibit B**, for such Well or Well Location, or (iii) obligate the Company to bear a Working Interest for such Well or Well Location (unless the Net Revenue Interest for such Asset is greater than the Net Revenue Interest set forth in **Exhibit B**, in the same proportion as any increase in such Working Interest);
  - (k) zoning and planning ordinances and municipal regulations;
- (l) vendors, carriers, warehousemen's, repairmen's, mechanics, workmen's, materialmen's, construction or other like Encumbrances arising by operation of Law in the ordinary course of business or incident to the construction or improvement of any property in respect of obligations that are not yet due;
- (m) Encumbrances created under Leases or Applicable Operating Agreements or by operation of Law in respect of obligations that are not yet due;
  - (n) any Encumbrance affecting the Assets which is discharged by SM Energy at or prior to Closing;
- (o) any contracts (other than the Leases or other title instruments), restrictions or exclusions set forth in **Exhibit A** or **Exhibit B**, as applicable, and all litigation referenced in *Schedule 3.12(a)*; and
- (p) the Leases and all other Encumbrances, contracts (including the Applicable Contracts), agreements, instruments, obligations, defects and irregularities affecting the Assets that

(in each case) do not (i) materially impair the use, ownership or operation of the Assets (as currently owned and operated), (ii) reduce the Net Revenue Interest of the Company in any Well or Well Location to an amount less than the Net Revenue Interest set forth in **Exhibit B** for such Well or Well Location, or (iii) obligate the Company to bear a Working Interest for such Well or Well Location in any amount greater than the Working Interest set forth in **Exhibit B** for such Well or Well Location (unless the Net Revenue Interest for such Asset is greater than the Net Revenue Interest set forth in **Exhibit B**, in the same proportion as any increase in such Working Interest).

"*Person*" means any individual, firm, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization, Governmental Authority or any other entity.

"Personal Property" means equipment, machinery, fixtures, and other real, immovable, personal, movable and mixed property, including saltwater disposal wells, well equipment, casing, rods, tanks, boilers, buildings, tubing, pumps, motors, fixtures, machinery, compression equipment, flow lines, and separation facilities, structures, materials, and other items used or held for use in the operation thereof and located upstream of the outlet flange of the relevant custody transfer meter (or, in the case of Hydrocarbon liquids, upstream of the outlet flange in the tanks).

"*Preferential Right*" means all preferential rights to purchase relating to the Assets (or an item that would otherwise constitute an Asset but for the fact it was not included in the Contribution because of a Consent or Preferential Right) or the Interests that would be applicable in connection with (i) the consummation by Buyer of the transactions contemplated by this Agreement, or (ii) the Contribution.

"Preliminary Settlement Statement" has the meaning set forth in Section 2.6(a).

"Purchase Price" has the meaning set forth in Section 2.2, as such amount may be adjusted from time to time pursuant to Section 2.4 and Section 2.6.

"Records" means the Company's and SM Energy's books, records and files, including any and all title, Tax, financial, technical, engineering, environmental and safety records and information, to the extent relating to the Assets, including all the files, records and data relating to the foregoing items, which records shall include, without limitation: lease records, well records, division order records, well files, title records (including abstracts of title, title opinions and memoranda, and title curative documents), engineering records, geological and all technical evaluations, interpretative data and technical data and information relating to the Assets, correspondence, electronic data files (if any), maps, production records, electric logs, core data, pressure data, decline curves and graphical production curves, reserve reports, appraisals and accounting and Asset Tax records.

"Recourse Parties" has the meaning set forth in Section 14.16.

"Related Person" has the meaning given in Section 3.33.

- "Remediation" means, with respect to an Environmental Condition, the implementation and completion of any investigation, monitoring, remedial, removal, response, construction, closure, disposal or other corrective actions or cures required under Environmental Laws or by a Governmental Authority to cure, correct or remove such Environmental Condition.
- "Remediation Amount" means, with respect to an Environmental Condition, the present value as of the Closing Date of the cost (net to the interest of SM Energy, the Company or its or their Affiliates) of the most cost effective Remediation of such Environmental Condition that is reasonably available.
  - "Required Consents" means those consents identified on Schedule 9.1(f).
  - "Requisite Financial Statement Information" has the meaning set forth in Section 8.4(a).
  - "Schedules" mean the schedules attached hereto.
  - "Securities Act" has the meaning set forth in Section 4.9.
  - "Seismic" means that geological and geophysical data (including seismic data) included in the Assets, set forth on Exhibit E.
- "Seismic License" means the Seismic License from Buyer, as Licensor, to SM Energy, as Licensee, covering all geological and geophysical data (including seismic data) included in the Assets, substantially in the form attached hereto as **Exhibit G**.
  - "Seller Audit Firm" has the meaning set forth in Section 8.4(b).
  - "Seller Group" has the meaning set forth in Section 8.4(a).
  - "SM Energy" has the meaning set forth in the preamble to this Agreement.
  - "SM Indemnified Parties" has the meaning set forth in Section 13.3.
  - "Specified Required Consent" means those Consents identified as "Specified Required Consents" on Schedule 9.1(f)1.
  - "Sponsor" means Kohlberg Kravis Roberts & Co. L.P.
- "Springfield Assets" means the Springfield Gathering System and the other assets owned, held, used or held for use by SM Energy (or the Company following the Contribution) in connection with the Springfield Gathering System.
- "Springfield Gas Gathering System" means the gas gathering system (including all appurtenances thereto) which is more specifically described on Exhibit I.
  - "Springfield Gathering System" means, collectively, the Springfield Gas Gathering System and the Springfield Oil Gathering System.

- "Springfield Oil Gathering System" means the oil gathering system (including all appurtenances thereto) which is more specifically described on Exhibit I.
- "Springfield Surface Contracts" means all easements, permits, licenses, servitudes, rights-of-way, surface leases, fee interests in real property and other surface rights appurtenant to, and used or held for use in connection with the Springfield Assets.
- "Springfield System Ownership Interests" means the twelve point five percent (12.5%) System Ownership Interest (as defined in the COO Agreement) owned by SM Energy (or the Company following the Contribution).
  - "Subject Transfer Taxes" has the meaning set forth in Section 10.2.
- "Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company or other entity of which a majority of the shares of capital stock or other ownership interests having ordinary voting power to elect a majority of the board of directors or other similar managing body of such corporation, partnership, limited liability company or other entity are owned directly or indirectly by such Person.
  - "Suspense Funds" means all amounts held by SM Energy or the Company in suspense that are attributable to the Assets.
  - "Target Intervals" means the intervals set forth in Exhibit D, as applicable.
- "Tax Return" means any return (including any information return), report, statement, schedule, declaration, notice, form, election, estimated Tax filing, claim for refund or other document (including any attachments thereto and amendments thereof) filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority with respect to or in connection with any Tax.
- "Taxes" means any and all federal, state, local, foreign and other taxes, unclaimed property or escheat obligations, or other assessments, including, without limitation, all net income, gross income, gross receipts, sales, use, ad valorem, transfer, franchise, profits, profit share, license, lease, service, service use, value added, withholding, payroll, employment, excise, estimated severance, stamp, occupation, premium, property, windfall profit or other taxes of any kind whatsoever, together with any interests, penalties, additions to tax, fines or other additional amounts imposed thereon or related thereto, and the term "Tax" means any one of the foregoing Taxes.
  - "Third Party" means any Person other than a Party to this Agreement or an Affiliate of a Party to this Agreement.
  - "Third Party Claim" has the meaning set forth in Section 13.7(b).
  - "Third Party Operator" has the meaning set forth in Section 5.1(a).
  - "Title Arbitrator" has the meaning set forth in Section 6.2(i).

"*Title Benefit*" means any right, circumstance or condition that operates to increase the Net Revenue Interest being assigned to Buyer in any Well or Well Location above that shown for such Well or Well Location in **Exhibit B**, to the extent the same does not cause a greater than proportionate increase in the Working Interest being assigned to Buyer therein above that shown in **Exhibit B**.

- "Title Benefit Amount" has the meaning set forth in Section 6.2(d).
- "Title Benefit Notice" has the meaning set forth in Section 6.2(b).
- "Title Benefit Property" has the meaning set forth in Section 6.2(b).
- "Title Defect" means any Encumbrance, defect or other matter that causes the Company not to have Defensible Title in and to any Asset; provided that the following shall not be considered Title Defects:
- (a) defects in the chain of title consisting of the failure to recite marital status in a document or omissions of successions of heirship or estate proceedings, unless Buyer provides affirmative evidence that such failure or omission could reasonably be expected to result in another Person's superior claim of title to the relevant Asset;
  - (b) defects arising out of lack of survey, unless a survey is expressly required by applicable Laws;
- (c) liens created under deeds of trust, mortgages and similar instruments by the lessor under a Lease (solely to the extent SM Energy's interest derives from the leasehold estate thereunder) covering the lessor's surface and mineral interests in the land covered thereby that are either (i) subordinated to the Lease or (ii) that are not subordinated to such Lease, but only insofar as the same (x) are liquidated in amount and are not in default or subject to foreclosure, (y) do not reduce SM Energy's Net Revenue Interest in such Lease, and (z) do not increase SM Energy's Working Interest in such Lease;
- (d) defects arising out of lack of corporate or other entity authorization unless Buyer provides affirmative evidence that causes Buyer to reasonably believe such corporate or other entity action may not have been authorized and could reasonably be expected to result in another Person's superior claim of title to the relevant Asset; and
  - (e) defects that have been cured by applicable Laws of limitations or prescription.
  - "Title Defect Amount" has the meaning set forth in Section 6.2(f).
  - "Title Defect Notice" has the meaning set forth in Section 6.2(a).
  - "Title Defect Property" has the meaning set forth in Section 6.2(a).
  - "Title Disputed Matters" has the meaning set forth in Section 6.2(i).

"Transaction Documents" means this Agreement and any instrument, certificate or document delivered in connection with this Agreement or the transactions contemplated hereunder together with the Contribution Documents and any instrument, certificate or document delivered in connection with this Agreement or the transactions contemplated thereunder.

"Transition Services Agreement" means the Transition Services Agreement, substantially in the form attached hereto as Exhibit J.

"Treasury Regulations" means the regulations promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Internal Revenue Code. All references herein to sections of the Treasury Regulations shall include any corresponding provision or provisions of succeeding, similar, substitute, proposed or final Treasury Regulations.

"Units" means SM Energy's or the Company's right, title and interest in and to any units or pooling arrangements wherein any Lease is pooled or unitized.

"Wells" means, except for the Excluded Assets, SM Energy's or the Company's right, title and interest in and to any oil and gas well, fresh water well, injection well, salt water disposal well and each other well of every nature and kind located on the Leases or Units, including those described in Exhibit B.

"Well Location" means each well location and the assumed spacing unit or governmental spacing unit associated with such well location, as set forth in Exhibit B. The assumed spacing unit is calculated as follows: (i) the completed lateral length specified on Exhibit B for such location multiplied by (ii) (x) 1,000 for Well Locations in the upper Eagle Ford or (y) 500 for all other Target Intervals divided by (iii) 43,560.

"Working Interest" means, with respect to any Well or Well Location (but limited to the applicable Target Interval set forth in **Exhibit B** for such Well or Well Location and any currently producing formation (for such Well)), the interest in and to such Target Interval for such Well or Well Location or producing formation (for such Well) that is burdened with the obligation to bear and pay costs and expenses of maintenance, development and operations on or in connection with such Target Interval for such Well or Well Location or producing formation (for such Well), but without regard to the effect of any Burdens.

## SECOND AMENDMENT TO MEMBERSHIP INTEREST PURCHASE AGREEMENT

THIS SECOND AMENDMENT TO MEMBERSHIP INTEREST PURCHASE AGREEMENT (this "Second Amendment") is made and entered into as of March 4, 2017, by and between SM ENERGY COMPANY, a Delaware corporation ("SM Energy") and VENADO EF L.P. (f/k/a Venado EF LLC), a Delaware limited partnership ("Buyer"). SM Energy and Buyer shall sometimes be referred to herein together as the "Parties", and each individually as a "Party". Unless otherwise specified herein and as amended by this Second Amendment, capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the MIPA (as defined below).

### RECITALS

- A. SM Energy and Buyer entered into that certain Membership Interest Purchase Agreement, dated as of January 1, 2017 (as amended, the "MIPA").
  - B. SM Energy and Buyer agreed to amend the MIPA as of February 28, 2017.
- C. Section 14.9 of the MIPA provides that the MIPA may be amended by an instrument in writing executed by the Parties and expressly identified as an amendment or modification of the MIPA as contemplated herein.
  - D. SM Energy and Buyer wish to amend the MIPA in accordance with the provisions of this Second Amendment.

## **AGREEMENT**

NOW THEREFORE, for and in consideration of the mutual agreements contained in the MIPA and this Second Amendment and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

1. Section 11.1 of the MIPA is hereby amended and restated in its entirety as follows:

Date of Closing. Subject to the conditions stated in this Agreement, the transfer by SM Energy and the acceptance by Buyer of the Interests (the "Closing") shall occur on March 15, 2017, or, if all conditions to Closing in ARTICLE IX (other than those conditions that are only capable of being satisfied at the Closing) have not yet been satisfied or waived by that date, five (5) Business Days after such conditions have been satisfied or waived, or such other date as Buyer and SM Energy may agree upon in writing. The date when Closing actually occurs shall be the "Closing Date."

- 2. As amended by this Second Amendment, all terms and provisions of the MIPA shall remain unchanged and in full force and effect, and the MIPA, as amended by this Second Amendment, is hereby ratified, acknowledged and reaffirmed by SM Energy and Buyer.
- 3. This Second Amendment may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall constitute for all purposes one agreement. Any signature hereto delivered by a Party by facsimile transmission shall be deemed an original signature hereto.
- 4. If any term or other provision of this Second Amendment is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Second Amendment shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any adverse manner to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Second Amendment so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.
  - 5. The terms and provisions of Article XIV of the MIPA shall be deemed to apply hereto, *mutatis mutandis*.
- 6. This Second Amendment, together with the MIPA (as amended by this Second Amendment), constitutes the full and entire understanding and agreement between the Parties with regard to the subjects hereof and thereof.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have executed this Second Amendment by their duly authorized representatives on and as of the date hereof.

# **SM ENERGY COMPANY**

By: <u>/s/ DAVID W. COPELAND</u> Name: David W. Copeland

Title: Executive Vice President, General Counsel and Corporate

Secretary

# VENADO EF L.P.

By: Venado EF GP LLC, its general partner

By: /s/ R. SCOTT GARRICK
Name: R. Scott Garrick
Title: Chief Executive Officer

[SECOND AMENDMENT TO MEMBERSHIP INTEREST PURCHASE AGREEMENT]

# SM ENERGY COMPANY AND SUBSIDIARIES RATIO OF EARNINGS TO FIXED CHARGES

For the Three Months Ended

For the Years Ended December 31, March 31, 2017 2016 2015 2014 2013 2012 (in thousands, except ratios) \$ 118,940 Income (loss) before income taxes (1,201,916) \$ (722,861) \$ 1,064,699 \$ 278,611 (83,517)Add: Fixed charges 49,615 167,600 155,510 117,147 102,758 77,841 Add: Amortization of capitalized interest 3,209 13,905 11,448 11,784 9,095 9,116 Less: Capitalized interest (17,004)(25,051)(12,135)(2,189)(16,165)(10,952)(8,716) 1,177,129 382,201 (1,037,415) (583,286) Earnings before fixed charges \$ 169,575 Fixed charges: Interest expense (1) \$ 46,953 148,685 128,149 98,554 89,711 63,720 Capitalized interest 2,189 17,004 25.051 16,165 10.952 12,135 Interest expense component of rent (2) 473 1,911 2,310 2,428 2,095 1,986 49,615 167,600 155,510 117,147 102,758 77,841 Total fixed charges \$ \$ 10.0 Ratio of earnings to fixed charges 3.4 3.7 Insufficient coverage \$ 1,205,015 738,796 \$ \$ 86,557

<sup>(1)</sup> Includes amortization of deferred financing costs and discount. Interest expense for the year ended December 31, 2016, excludes the \$10.0 million paid to terminate a second lien facility that was no longer necessary to fund acquisition activity.

<sup>(2)</sup> Represents a reasonable approximation of the portion of rental expense assumed to be attributable to the interest factor.

#### CERTIFICATION

### I, Javan D. Ottoson, certify that:

- I have reviewed this quarterly report on Form 10-Q of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 3, 2017

/s/ JAVAN D. OTTOSON Javan D. Ottoson

President and Chief Executive Officer

#### CERTIFICATION

### I, A. Wade Pursell, certify that:

- I have reviewed this quarterly report on Form 10-Q of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 3, 2017

/s/ A. WADE PURSELL

A. Wade Pursell

Executive Vice President and Chief Financial Officer

### CERTIFICATION

### PURSUANT TO

### 18 U.S.C. SECTION 1350,

### AS ADOPTED PURSUANT TO

## SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of SM Energy Company (the "Company") for the quarterly period endedMarch 31, 2017, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Javan D. Ottoson, as President and Chief Executive Officer of the Company, and A. Wade Pursell, as Executive Vice President and Chief Financial Officer of the Company, each hereby certifies, pursuant to and solely for the purpose of 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, to the best of his knowledge and belief, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

## /s/ JAVAN D. OTTOSON

Javan D. Ottoson President and Chief Executive Officer May 3, 2017

## /s/ A. WADE PURSELL

A. Wade Pursell Executive Vice President and Chief Financial Officer May 3, 2017